UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): February 10, 2022

CEPTON, INC.

(Ex	act name of registrant as specified in its cl	harter)
Delaware	001-39959	27-2447291
(State or other jurisdiction of incorporation)	(Commission File Number)	(IRS Employer Identification No.)
(Addre	399 West Trimble Road San Jose, CA 95131 ss of principal executive offices, including	g zip code)
Registrant'	's telephone number, including area code:	408-459-7579
(Former	Growth Capital Acquisition Corp. 300 Park Avenue, 16th Floor New York, New York 10022 name or former address, if changed since	last report)
Check the appropriate box below if the Form 8-K fill following provisions:	ling is intended to simultaneously satisfy	y the filing obligation of the registrant under any of the
☐ Written communications pursuant to Rule 425 unde	er the Securities Act (17 CFR 230.425)	
☐ Soliciting material pursuant to Rule 14a-12 under t	he Exchange Act (17 CFR 240.14a-12)	
☐ Pre-commencement communications pursuant to R	tule 14d-2(b) under the Exchange Act (17	CFR 240.14d-2(b))
☐ Pre-commencement communications pursuant to R	tule 13e-4(c) under the Exchange Act (17	CFR 240.13e-4(c))
Securities registered pursuant to Section 12(b) of the Ad	ct:	
Title of Each Class:	Trading Symbol(s)	Name of Each Exchange on Which Registered:
Common stock, par value \$0.00001 per share Redeemable warrants, exercisable for common stock at an exercise price of \$11.50 per share, subject to adjustment	CPTN CPTNW	The Nasdaq Capital Market The Nasdaq Capital Market
Indicate by check mark whether the registrant is an exchapter) or Rule 12b-2 of the Securities Exchange Act of		Rule 405 of the Securities Act of 1933 (§230.405 of this
		Emerging growth company ⊠
If an emerging growth company, indicate by check man or revised financial accounting standards provided purs		he extended transition period for complying with any new . \Box

INTRODUCTORY NOTE

On February 10, 2022 (the "<u>Closing Date</u>"), Cepton, Inc., a Delaware corporation (the "<u>Company</u>") (*f*/*k*/a Growth Capital Acquisition Corp. ("<u>GCAC</u>")), consummated the previously announced merger (the "<u>Closing</u>") pursuant to that certain Business Combination Agreement, dated August 4, 2021, and as amended by the Amendment to the Business Combination Agreement, dated January 21, 2022 (as so amended, the "<u>Business Combination Agreement</u>"), by and among GCAC, GCAC Merger Sub Inc., a Delaware corporation and a wholly owned subsidiary of GCAC ("<u>Merger Sub</u>"), and Cepton Technologies, Inc., a Delaware corporation ("<u>Legacy Cepton</u>"). GCAC's stockholders approved the Transactions (as defined below) at a special meeting of stockholders held on February 9, 2022 (the "<u>Special Meeting</u>").

Pursuant to the Business Combination Agreement, Merger Sub merged with and into Legacy Cepton (the "Merger" and, together with the other transactions contemplated by the Business Combination Agreement, the "Transactions"), with Legacy Cepton continuing as the surviving corporation in the Merger and a wholly-owned subsidiary of GCAC. On the Closing Date, the registrant changed its name from Growth Capital Acquisition Corp. to Cepton, Inc.

Conversion of Securities and Merger Consideration

Immediately prior to the effective time of the Merger (the "Effective Time"), (i) each share of Legacy Cepton Class F Stock and each share of Legacy Cepton preferred stock automatically converted into a number of shares of Legacy Cepton common stock, at the then-effective conversion rate for such capital stock as calculated pursuant to Legacy Cepton's Amended and Restated Certificate of Incorporation and (ii) each warrant for Legacy Cepton common stock was deemed exercised on a net exercise basis pursuant to the terms of such warrants. All of the shares of Legacy Cepton Class F Stock and the Legacy Cepton preferred stock that converted into shares of common stock of Legacy Cepton are no longer outstanding and ceased to exist, and each holder of shares of Legacy Class F Stock and/or Legacy Cepton preferred stock ceased to have any rights with respect to such securities.

At the Effective Time, by virtue of the Merger and without any action on the part of GCAC, Merger Sub, Legacy Cepton or the holders of any of the following securities:

- (a) Each share of Legacy Cepton common stock (other than the Dissenting Shares and the Cancelled Shares (as such terms are defined in the Business Combination Agreement)) converted into (i) the contingent right to receive Earnout Shares (as defined below) (which may be zero) following the Closing and (ii) a certain number of shares of GCAC Class A common stock equal to (x) \$1,500,000,000 divided by the total number of Legacy Cepton capital stock (on an "as-converted" to Legacy Cepton common stock basis) on a fully diluted basis as of the date of Closing (but excluding company options that are not vested), divided by (y) 10 (the "Merger Consideration").
- (b) Each option to purchase shares of Legacy Cepton common stock, whether or not exercisable and whether or not vested, outstanding immediately prior to the Effective Time was assumed by GCAC and converted into an option to purchase shares of GCAC Class A common stock (each, a "Converted Option"). Each Converted Option has and is subject to the same terms and conditions (including vesting, expiration and exercisability terms) as were applicable to the Legacy Cepton option from which it was converted immediately before the Effective Time, except that (x) each Converted Option became exercisable for that number of shares of GCAC Class A common stock equal to the product (rounded down to the nearest whole number) of (1) the number of shares of Legacy Cepton common stock subject to the Legacy Cepton option immediately before the Effective Time and (2) the Merger Consideration and (y) the per share exercise price for each share of GCAC Class A common stock issuable upon exercise of the Converted Option will be equal to the quotient (rounded up to the nearest whole cent) obtained by dividing (1) the exercise price per share of Legacy Cepton common stock of such Legacy Cepton option immediately before the Effective Time by (2) the Merger Consideration.

Earnout Merger Consideration

In addition to the Merger Consideration set forth above, additional contingent shares ("<u>Earnout Shares</u>") are payable to each holder of Legacy Cepton common stock and/or Legacy Cepton options receiving consideration in the Merger, in the amounts set forth below:

- (a) If the closing share price of GCAC Class A common stock equals or exceeds \$15.00 per share for any 20 trading days within any consecutive 30-trading day period that occurs after the Closing Date and on or prior to the three-year anniversary of the Closing Date, then, GCAC will issue to each holder of Legacy Cepton common stock that is entitled to Earnout Shares a number of shares of GCAC Class A common Stock equal to such holder's pro rata portion of 7,000,000 shares.
- (b) If the closing share price of GCAC Class A common stock equals or exceeds \$17.50 per share for any 20 trading days within any consecutive 30-trading day period that occurs after the Closing Date and on or prior to the three (3)-year anniversary of the Closing Date, GCAC will issue to each holder of Legacy Cepton common stock that is entitled to Earnout Shares, a number of shares of GCAC Class A common stock equal to such holder's Earnout Pro Rata Portion of 6,000,000 shares.

In the event that after the Closing and prior to the three-year anniversary of the Closing Date, (i) there is a Change of Control (as defined in the Business Combination Agreement) (or a definitive agreement providing for a Change of Control has been entered into prior to the three-year anniversary of the Closing Date and such Change of Control is ultimately consummated, even if such consummation occurs after the three -year anniversary of the Closing Date), (ii) any liquidation, dissolution or winding up of GCAC (whether voluntary of involuntary) is initiated, (iii) any bankruptcy, reorganization, debt arrangement or similar proceeding under any bankruptcy, insolvency or similar law, or any dissolution or liquidation proceeding, is instituted by or against GCAC, or a receiver is appointed for GCAC or a substantial part of its assets or properties or (iv) GCAC makes an assignment for the benefit of creditors, or petitions or applies to any governmental authority for, or consents or acquiesces to, the appointment of a custodian, receiver or trustee for all or substantially all of its assets or properties (any of (i) to (iv), an "Acceleration Event"), then any Earnout Shares that have not been previously issued by GCAC (whether or not previously earned) shall be deemed earned and issued by GCAC to the holders of Legacy Cepton common stock, unless, in the case of an Acceleration Event that is a Change of Control, the value of the consideration to be received by the holders of the GCAC Class A common stock in such Change of Control transaction is less than the stock price threshold applicable to the relevant Earnout Shares.

A description of the Merger and the terms of the Business Combination Agreement are included in the proxy statement/consent solicitation statement/prospectus, dated January 24, 2022 (the "<u>Proxy Statement/Consent Solicitation Statement/Prospectus</u>") as filed with the Securities and Exchange Commission (the "<u>SEC</u>") in the section entitled "<u>The Business Combination Proposal (Proposal 1)—The Business Combination Agreement</u>" beginning on page 133 of the Proxy Statement/Consent Solicitation Statement/Prospectus.

PIPE Investment

On August 4, 2021 and October 19, 2021, GCAC entered into certain subscription agreements (the "PIPE Subscription Agreements"), as amended by the Amendment to the PIPE Subscription Agreement, dated February 3, 2022 (the "PIPE Subscription Agreement Amendments"), with the investors therein (the "PIPE Investors"), pursuant to which the PIPE Investors agreed to purchase, and GCAC agreed to sell to the PIPE Investors, an aggregate of 5,950,000 shares of GCAC Class A common stock for gross proceeds to GCAC of \$59.5 million in a private placement. Pursuant to the PIPE Subscription Agreements, the Company gave certain registration rights to the Subscribers with respect to the PIPE Shares. The sale of PIPE Shares was consummated concurrently with the Closing.

The foregoing description of each of the Business Combination Agreement, the PIPE Subscription Agreements and Amended PIPE Subscription Agreements is a summary only and is qualified in their entirety by the full text of the Business Combination Agreement, Form of PIPE Subscription Agreement and Form of PIPE Subscription Agreement Amendment, copies of which are attached hereto as Exhibit 2.1, Exhibit 10.1 and Exhibit 10.2, respectively, and are incorporated herein by reference.

Related Agreements

Contemporaneously with the execution and delivery of the Business Combination Agreement, (i) GCAC and certain Legacy Cepton stockholders entered into Stockholder Support Agreements; (ii) GCAC, the Sponsor Group and certain other GCAC shareholders parties thereto, Legacy Cepton, and certain Legacy Cepton stockholders entered into an Amended and Restated Registration Rights Agreement, which became effective as of Closing; (iii) certain Legacy Cepton stockholders entered into a Confidentiality and Lock-up Agreement with GCAC, each of which became effective as of Closing; (iv) GCAC, Sponsor, Nautilus, HB Strategies, and Legacy Cepton entered into an Unpaid Expenses and Lock-Up Agreement; and (v) Legacy Cepton and certain GCAC stockholders entered into Stockholder Support Agreements, a copy of which are filed as Exhibits 10.2, 10.3, 10.4, 10.5, 10.6 and 10.7, respectively, to this Report, and incorporated herein by reference.

Capitalized terms used but not defined in this Current Report on Form 8-K (the "Report") have the meanings set forth in the Proxy Statement/Consent Solicitation Statement/Prospectus.

Item 1.01 Entry into a Material Definitive Agreement.

On the Closing Date, in connection with the consummation of the Transactions, the Company entered into indemnification agreements with each of its directors and executive officers following the Business Combination. These agreements, among other things, will require the Company to indemnify the Company's directors and executive officers for certain expenses, including attorneys' fees, judgments and fines incurred by a director or officer in any action or proceeding arising out of their services as one of the Company's directors or officers or any other company or enterprise to which the person provides services at the Company's request.

The foregoing description of the indemnification agreements does not purport to be complete and is qualified in its entirety by reference to the full text of the form of indemnification agreement, a copy of which is attached hereto as Exhibit 10.8 and is incorporated herein by reference.

Item 2.01 Completion of Acquisition or Disposition of Assets.

The disclosure set forth in the "Introductory Note" above is incorporated by reference into this Item 2.01.

As previously reported by the Company on a Current Report on Form 8-K filed with the SEC, on February 9, 2022, at the Special Meeting, GCAC's stockholders approved the Business Combination Agreement, the Transactions and the other related proposals presented in the Proxy Statement/Consent Solicitation Statement/Prospectus. On February 10, 2022, the parties to the Business Combination Agreement consummated the Transactions.

Holders of an aggregate of 15,589,540 shares of GCAC Class A common stock sold in GCAC's initial public offering (the "<u>public shares</u>") properly exercised their right to have such shares redeemed for a full pro rata portion of the trust account holding the proceeds from GCAC's initial public offering, calculated as of two (2) business days prior to the consummation of the Business Combination, which was approximately \$10.00 per share, or \$155.9 million in the aggregate.

In connection with the Closing, Silicon Valley Bank ("<u>SVB</u>") was issued 136,994 shares of the Company's common stock as a result of exercising warrants previously issued to SVB by Legacy Cepton (the "<u>SVB Warrant Exercise</u>") and Trinity Capital, Inc. ("<u>Trinity</u>") was issued 73,741 shares of the Company's common stock as a result of the automatic net exercise of warrants previously issued to Trinity by Legacy Cepton (the "<u>Trinity Warrant Exercise</u>" and, together with the SVB Warrant Exercise, the "<u>Warrant Exercises</u>").

As a result of the Business Combination, each share of Legacy Cepton common stock outstanding immediately prior to the effective time of the Business Combination (including shares of Legacy Cepton common stock issued upon conversion of Legacy Cepton Class F Stock and preferred stock immediately prior to the Closing) was converted into the right to receive approximately 2.45 shares of the Company's common stock.

Immediately following consummation of the Transactions, including the redemption of public shares as described above, the consummation of the PIPE Investment and the Warrant Exercises, there were 154,048,001 shares of the Company's common stock issued and outstanding.

The Company's common stock and warrants is expected to commence trading on the Nasdaq Capital Market ("Nasdaq") under the symbols "CPTN" and "CPTNW," respectively, on February 11, 2022, subject to ongoing review of the Company's satisfaction of all listing criteria following the Business Combination.

As noted above, an aggregate of \$155.9 million was paid from the Company's trust account to holders that properly exercised their right to have public shares redeemed, and the remaining balance immediately prior to the Closing of approximately \$16.6 million remained in the trust account. The remaining amount in the trust account was used to fund expenses incurred by Legacy Cepton and GCAC in connection with the Business Combination and will be used for general corporate purposes of the Company following the Business Combination.

FORM 10 INFORMATION

Item 2.01(f) of Form 8-K states that if the registrant was a "shell company" (as such term is defined in Rule 12b-2 under the Securities Exchange Act of 1934, as amended (the "Exchange Act")), as the Company was immediately before the Business Combination, then the registrant must disclose the information that would be required if the registrant were filing a general form for registration of securities on Form 10. Accordingly, the Company is providing below the information that would be included in a Form 10 if it were to file a Form 10. Please note that the information provided below relates to the combined company after the consummation of the Business Combination, unless otherwise specifically indicated or the context otherwise requires.

Cautionary Note Regarding Forward-Looking Statements

Certain statements contained in this Report may constitute "forward-looking statements" within the meaning of the "safe harbor" provisions of the United States Private Securities Litigation Reform Act of 1995. In addition, any statements that refer to projections, forecasts or other characterizations of future events or circumstances, including any underlying assumptions, are forward-looking statements. Forward-looking statements may be identified by the use of words such as "estimate," "plan," "project," "forecast," "intend," "will," "expect," "anticipate," "believe," "seek," "target," "designed to" or other similar expressions that predict or indicate future events or trends or that are not statements of historical matters. The Company cautions readers of this Report that these forward-looking statements are subject to risks and uncertainties, most of which are difficult to predict and many of which are beyond the Company's control, that could cause the actual results to differ materially from the expected results. These forward-looking statements include, but are not limited to, statements regarding estimates and forecasts of financial and performance metrics, projections of market opportunity and market share, potential benefits and the commercial attractiveness to its customers of the Company's products and services, the potential success of the Company's marketing and expansion strategies, the potential for the Company to achieve design awards, and the potential benefits of the Business Combination (including with respect to shareholder value). These statements are based on various assumptions, whether or not identified in this Report, and on the current expectations of the Company's management and are not predictions of actual performance. These forward-looking statements are provided for illustrative purposes only and are not intended to serve as, and must not be relied on by any investor as, a guarantee, an assurance, a prediction or a definitive statement of fact or probability. Actual events and circ

- the conditions affecting the markets in which the Company operates;
- the success of the Company's strategic relationships, including with the Company's Tier 1 partners, none of which are exclusive;
- fluctuations in sales of the Company's major customers;

- fluctuations in capital spending in the Automotive and Smart Infrastructure markets;
- the impact of the coronavirus pandemic on the global economy and financial markets, including any restrictions on the Company's operations and the operations of the Company's customers and suppliers resulting from public health requirements and government mandates;
- changes in applicable laws or regulations;
- the possibility that the Company's business may be adversely affected by other economic business, and/or competitive factors;
- the risk that current trends in the Automotive and Smart Infrastructure markets decelerate or do not continue;
- estimates for the financial performance of the Company's business may prove to be incorrect or materially different from actual results;
- failure to realize the anticipated benefits of the Business Combination;
- risks relating to the uncertainty of the projected financial and operating information with respect to the Company, including whether the
 Company will be able to achieve its target milestones, its pricing and sales volume targets, and its proposed production timelines and win the
 engagements contemplated in its projected pipeline, and the ability of OEMs and other strategic partners to re-source or cancel vehicle or
 technology programs;
- risks related to future market adoption of the Company's offerings;
- the final terms of the Company's arrangement with its Tier 1 partner and, in turn, its Tier 1 partner's award with OEM-B differing from the Company's expectations, including with respect to volume and timing, or the arrangement can be terminated or may not materialize into a long-term contract partnership arrangement;
- risks related to the Company's marketing and growth strategies;
- the effects of competition on the Company's future business;
- expectations with respect to future operating and financial performance and growth, including when the Company will generate positive cash flow from operations;
- the Company's ability to raise funding on reasonable terms as necessary to develop its product in the timeframe contemplated by its business plan, and to comply with the terms of any restrictive, financial or other covenants included in the agreements governing such funding;
- the Company's ability to execute its business plans and strategy;
- the outcome of any legal proceedings that may be instituted against the Company related to Business Combination; and
- other risks and uncertainties described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the section entitled "Risk Factors" beginning on page 62 thereof and are incorporated herein by reference.

If any of these risks materialize or any of the Company's assumptions prove incorrect, actual results could differ materially from the results implied by these forward-looking statements. There may be additional risks that the Company is not presently aware of or that the Company currently believe are immaterial that could also cause actual results to differ from those contained in the forward-looking statements. In addition, forward-looking statements reflect the Company's expectations, plans or forecasts of future events and views as of the date of this Report. The Company anticipates that subsequent events and developments will cause the Company's assessments to change. However, while the Company may elect to update these forward-looking statements at some point in the future, the Company specifically disclaims any obligation to do so. These forward-looking statements should not be relied upon as representing the Company's assessments as of any date subsequent to the date of this Report. Accordingly, undue reliance should not be placed upon the forward-looking statements. Actual results, performance or achievements may, and are likely to, differ materially, and potentially adversely, from any projections and forward-looking statements and the assumptions on which those forward-looking statements were based. There can be no assurance that the data contained herein is reflective of future performance to any degree. You are cautioned not to place undue reliance on forward-looking statements as a predictor of future performance as projected financial information and other information are based on estimates and assumptions that are inherently subject to various significant risks, uncertainties and other factors, many of which are beyond the Company's control.

Business

The business of the Company is described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the section entitled "Information About Cepton" beginning on page 207 thereof and that information is incorporated herein by reference.

Risk Factors

The risks associated with the Company's business are described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the section entitled "*Risk Factors*" beginning on page 62 thereof and are incorporated herein by reference. A summary of the risks associated with the Company's business are also described on pages 51-53 of the Proxy Statement/Consent Solicitation Statement/Prospectus under the heading "*Risk Factors*" and are incorporated herein by reference.

Financial Information

The financial information of GCAC is described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the sections entitled "Selected Financial and Other Data of GCAC" and "GCAC's Management's Discussion and Analysis of Financial Condition and Results of Operations" beginning on pages 59 and 200 thereof, respectively, and are incorporated herein by reference.

The financial information of Legacy Cepton is described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the sections entitled "Cepton's Management's Discussion and Analysis of Financial Condition and Results of Operations" and "Cepton's Management's Discussion and Analysis of Financial Condition and Results of Operations—Qualitative and Quantitative Disclosures About Market Risk," beginning on pages 225 and 237 thereof, respectively, and are incorporated herein by reference.

Reference is made to the disclosure set forth in Item 9.01 of this Report relating to financial information of GCAC and Legacy Cepton, which is incorporated herein by reference.

Properties

The facilities of the Company are described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the section entitled "*Information About Cepton*" beginning on page 207 thereof and that information is incorporated herein by reference.

Security Ownership of Certain Beneficial Owners and Management

The following table sets forth information known to the Company regarding the beneficial ownership of the Company's common stock immediately following consummation of the Transactions by:

- each person who is known by the Company to be the beneficial owner of more than 5% of the outstanding shares of the Company's common stock;
- each of the Company's named executive officers and directors; and
- all of the Company's executive officers and directors as a group.

Beneficial ownership is determined according to the rules of the SEC, which generally provide that a person has beneficial ownership of a security if he, she or it possesses sole or shared voting or investment power over that security, which includes the power to dispose of or to direct the disposition of the security or has the right to acquire such powers within 60 days. In computing the number of shares of the Company's common stock beneficially owned by a person or entity and the percentage ownership, the Company deemed outstanding shares of its common stock subject to options and warrants held by that person or entity that are currently exercisable or exercisable within 60 days of the Closing Date. The Company did not deem these shares outstanding, however, for the purpose of computing the percentage ownership of any other person or entity.

Unless otherwise noted, the address of each beneficial owner is c/o Cepton, Inc., 399 West Trimble Road, San Jose, CA, 95131.

The beneficial ownership of the Company's common stock is based on 154,048,001 shares of the Company's common stock issued and outstanding immediately following consummation of the Transactions, including the redemption of public shares as described above and the consummation of the PIPE Investment and the Warrant Exercises.

Beneficial Ownership Table

Name and Address of Beneficial Owners	Number of Shares of Common Stock Beneficially Owned	%
5% Holders:		
Koito Manufacturing, Ltd. ⁽¹⁾	19,624,741	12.7%
LDV Partners Fund I, L.P. ⁽²⁾	16,088,422	10.4%
Yupeng Cui ⁽³⁾	9,679,376	6.3%
Executive Officers and Directors:		
Dr. Jun Pei ⁽⁴⁾	30,174,923	19.6%
Dr. Jun Ye ⁽⁵⁾	25,916,966	16.8%
Dr. Winston Fu ⁽⁶⁾	16,088,422	10.4%
Dr. Mark McCord ⁽⁷⁾	10,409,248	6.8%
Dr. Liqun Han ⁽⁸⁾	2,734,762	1.7%
Dr. Dongyi Liao ⁽⁹⁾	2,104,607	1.3%
George Syllantavos ⁽¹⁰⁾	1,367,500	*
Xiaogang (Jason) Zhang	_	_
Takayuki Katsuda ⁽¹¹⁾	_	_
Mei (May) Wang	_	
Directors and executive officers as a group (ten individuals)	88,796,428	56.1%

Less than one percent

- (1) Consists of 19,624,741 shares of common stock, including 5,000,000 PIPE shares. The business address of Koito Manufacturing Co., Ltd. is 4-8-3, Takanawa, Minato-ku, Tokyo, Japan.
- (2) Consists of 16,088,422 shares of common stock, including 200,000 PIPE shares.
- 3) Consists of 9,679,376 shares of common stock.
- (4) Consists of 1,224,617 shares of common stock owned by Dr. Pei and 28,950,306 shares of common stock owned by the Pei 2000 Trust of which Dr. Pei is a trustee.
- (5) Consists of 20,818,496 shares of common stock owned by Dr. Ye, 2,449,235 shares of common stock owned by the Lynnelle Lin Ye Trust dated December 8, 2020 of which Dr. Ye is a trustee, 2,449,235 shares of common stock owned by the Brion Qi Ye Irrevocable Trust dated December 8, 2020 of which Dr. Ye is a trustee, and 200,000 PIPE shares, owned by the Ye-Wang Family Trust, dated March 31, 2007, of which Dr. Ye is a trustee.
- (6) Consists of 16,088,422 shares of common stock, including 200,000 PIPE shares, owned by LDV Partners Fund I, L.P. Dr. Winston Fu is a managing member of LDV Partners I (GP), Ltd., which is the general partnership that manages LDV Partners Fund I, L.P. Dr. Fu disclaims beneficial ownership of the 16,088,422 shares of common stock except to the extent of any pecuniary interest he may have therein, directly or indirectly.
- (7) Consists of 10,409,248 shares of common stock owned by the McCord Trust, dated January 7, 2020, of which Dr. McCord is a trustee.
- (8) Consists of 489,847 shares of common stock owned by The Han-Ouyang Living Trust, U/A, dated March 21, 2021, of which Mr. Han is a trustee, and 2,244,915 shares of common stock issuable pursuant to options exercisable within 60 days of the Closing Date.
- (9) Consists of 2,104,607 shares of common stock issuable pursuant to options exercisable within 60 days of the Closing Date.
- $(10) \ \ Consists \ of \ 1,367,500 \ shares \ of \ common \ stock \ owned \ by \ Nautilus \ Carriers \ LLC \ of \ which \ Mr. \ Syllantavos \ is \ a \ managing \ member.$
- (11) Mr. Katsuda is the Managing Corporate Officer and a member of the board of directors of Koito Manufacturing Co., Ltd. The business address of Mr. Katsuda is c/o Koito Manufacturing Co., Ltd., 4-8-3, Takanawa, Minato-ku, Tokyo, Japan.

Directors and Executive Officers

Directors

In connection with the Closing, Prokopios (Akis) Tsirigakis, Harry Braunstein, Gary Leibler, and Evan Breibart resigned as directors of GCAC, the size of the board of directors of GCAC (the "GCAC Board") was increased from five (5) to seven (7), Jun Pei, Winston Fu, Jun Ye, Xiaogang (Jason) Zhang, Takayuki Katsuda, Mei (May) Wang were appointed to the GCAC Board, and George Syllantavos continued as a director, in each case, as described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the section entitled "Management After the Business Combination—Executive Officers and Directors After the Business Combination," which is incorporated herein by reference.

On February 10, 2022, Jun Ye, Winston Fu, and Mei (May) Wang were designated to serve as Class A Directors of the board of directors of the Company (the "Board"), with their terms expiring at the first annual meeting of the stockholders following the Closing. George Syllantavos and Jason Zhang were designated to serve as Class B Directors, with their terms expiring at the second annual meeting of the stockholders following the Closing. Jun Pei and Takayuki Katsuda were designated to serve as Class C Directors with their terms expiring at the third annual meeting of the stockholders following the Closing. Biographical information for these individuals is set forth in the Proxy Statement/Consent Solicitation Statement/Prospectus in the section titled "Management After the Business Combination" beginning on page 268, which is incorporated herein by reference. Jun Pei was appointed to serve as the Chairman of the Board.

Independence of Directors

Nasdaq listing standards require that a majority of the Board be independent. An "independent director" is defined generally as a person other than an officer or employee of the company or its subsidiaries or any other individual having a relationship which in the opinion of the Board, would interfere with the director's exercise of independent judgment in carrying out the responsibilities of a director. The Board has determined that Jun Ye, Xiaogang (Jason) Zhang, Mei (May) Wang, and George Syllantavos are "independent directors" as defined in the Nasdaq listing standards and applicable SEC rules. Our independent directors will have regularly scheduled meetings at which only independent directors are present.

In connection with the appointment of Jun Pei as chair of the Board, the Board created the position of Lead Independent Director and appointed Jun Ye to serve in that role. As Lead Independent Director, Jun Ye will have the authority and responsibilities described in our Corporate Governance Guidelines and will generally help coordinate the efforts of our non-employee directors.

Director Compensation

We expect to grant each of our non-employee directors (other than Dr. Ye, who is one of our founders, and Mr. Katsuda, who is affiliated with Koito Manufacturing, Ltd., our automotive Tier 1 partner) an option to purchase up to 10,000 shares of the Company's common stock with an exercise price equal to the per-share value of our common stock on the grant date. We expect each option will vest in monthly installments over a two-year period, subject to the director's continued service on the Board. In addition, we expect to pay each of our non-employee directors (other than Dr. Ye and Mr. Katsuda) a cash annual retainer of \$40,000 for their service on the Board, payable in quarterly installments. The Company plans to continue evaluating the compensation to be provided to its non-employee directors.

Committees of the Board of Directors

The standing committees of the Board consist of an audit committee (the "<u>Audit Committee</u>"), a compensation committee (the "<u>Compensation Committee</u>") and a nominating and corporate governance committee (the "<u>Nominating and Corporate Governance Committee</u>"). Each of the committees reports to the Board.

The Board appointed George Syllantavos, Xiaogang (Jason) Zhang and Mei (May) Wang to serve on the Audit Committee, with George Syllantavos as chair. The Board appointed George Syllantavos, Jun Ye and Mei (May) Wang to serve on the Compensation Committee, with Jun Ye as chair. The Board appointed George Syllantavos, Jun Ye and Xiaogang (Jason) Zhang to serve on the Nominating and Corporate Governance Committee, with Jun Ye as chair.

Executive Officers

Effective immediately prior to the Closing, Prokopios (Akis) Tsirigakis resigned as President and Co-Chief Executive Officer and George Syllantavos resigned as Co-Chief Executive Officer and Chief Financial Officer. The Board appointed Jun Pei to serve as President and Chief Executive Officer, Winston Fu to serve as Chief Financial Officer and Secretary, Mark McCord to serve as Chief Technology Officer, Liqun Han to serve as Senior Vice President of Operations and Dongyi Liao to serve as Senior Vice President of Applications. Biographical information for Jun Pei, Winston Fu, Mark McCord, Liqun Han and Dongyi Liao is set forth in the Proxy Statement/Consent Solicitation Statement/Prospectus in the section titled "Management After the Business Combination" beginning on page 268, which is incorporated herein by reference.

Executive Compensation

Executive Compensation

The executive compensation of the Company's named executive officers and directors is described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the section entitled "Executive and Director Compensation of Cepton" beginning on page 275 thereof and is incorporated herein by reference.

The foregoing description of the compensation of the Company's executive officers is qualified in its entirety by the full text of the employment agreements of Jun Pei, Winston Fu, Mark McCord, Liqun Han and Dongyi Liao, copies of which are attached hereto as Exhibit 10.12, Exhibit 10.13, Exhibit 10.14, Exhibit 10.15 and Exhibit 10.16, respectively, and incorporated herein by reference.

Compensation Committee Interlocks and Insider Participation

None of our executive officers serves as a member of the Board or Compensation Committee (or other committee performing equivalent functions) of any entity that has one or more executive officers serving on the Board or Compensation Committee.

Certain Relationships and Related Transactions

Certain Relationships and Related Person Transactions

Certain relationships and related person transactions are described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the sections entitled "Certain Cepton Relationships and Related Person Transactions" and "Certain GCAC Relationships and Related Person Transactions" beginning on pages 279 and 205 thereof, respectively, and are incorporated herein by reference.

Risk Oversight

Our risk management oversight is described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the section entitled "Management After the Business Combination—Role of the New Cepton Board in Risk Oversight/Risk Committee" beginning on page 271 thereof and that information is incorporated herein by reference.

Legal Proceedings

Reference is made to the disclosure regarding legal proceedings in the section of the Proxy Statement/Consent Solicitation Statement/Prospectus titled "Information About Cepton—Legal Proceedings" beginning on page 223, which is incorporated herein by reference.

Market Price of and Dividends on the Registrant's Common Equity and Related Stockholder Matters

Price Range of Securities and Dividends

The market price of and dividends on GCAC's common equity, warrants and units and related stockholder matters is described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the Section entitled "Market Price and Dividend Information" beginning on page 61 thereof and that information is incorporated herein by reference.

Prior to the Closing, GCAC's publicly traded units, Class A common stock and warrants were listed on the Nasdaq Capital Market under the symbols "GCACU," "GCAC," and "GCACW," respectively. Upon the Closing, the Company's common stock and warrants were listed on the Nasdaq Capital Market under the symbols "CPTN" and "CPTNW," respectively. Publicly traded units automatically separated into their component securities upon the Closing, and as a result, no longer trade as a separate security and were delisted from the Nasdaq Capital Market.

The Company has not paid any cash dividends on shares of its common stock to date. The payment of cash dividends in the future will be dependent upon the Company's revenue and earnings, if any, capital requirements and general financial condition. The payment of any cash dividends will be within the discretion of the Board.

Holders of Record

As of the Closing and following the completion of the Transactions, including the redemption of public shares as described above and the consummation of the PIPE Investment and the Warrant Exercises, the Company had 154,048,001 shares of common stock outstanding held of record by 162 holders and no shares of preferred stock outstanding. Such amounts do not include DTC participants or beneficial owners holding shares through nominee names.

Securities Authorized for Issuance Under Equity Compensation Plans

Reference is made to the disclosure described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the sections entitled "The Incentive Plan Proposal (Proposal 4)" and "The ESPP Proposal (Proposal 5)" beginning on pages 175 and 181, respectively, thereof, which is incorporated herein by reference.

At the Special Meeting, GCAC stockholders approved the Cepton, Inc. 2022 Incentive Award Plan (the "2022 Plan"), which is intended to replace the Cepton Technologies, Inc. Stock Incentive Plan (the "2016 Plan"), and the Employee Stock Purchase Plan (the "ESPP") and the material terms thereunder, including the authorization of the initial share reserve thereunder. In accordance with the terms of the 2022 Plan and the ESPP, 15,123,142 shares and 3,080,960 shares, respectively, were initially reserved for issuance thereunder.

Recent Sales of Unregistered Securities

Reference is made to the disclosure set forth under Item 3.02 of this Report relating to the issuance of the Company's common stock in connection with the Transactions, which is incorporated herein by reference.

Description of Registrant's Securities to be Registered

The Company's securities are described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the section entitled "Description of Securities of GCAC" beginning on page 242 thereof and that information is incorporated herein by reference. As described below, the Company's Second Amended and Restated Certificate of Incorporation (the "A&R Charter") was approved by GCAC's stockholders at the Special Meeting and became effective on February 10, 2022 in connection with the consummation of the Transactions.

Indemnification of Directors and Officers

The indemnification of our directors and officers is described in the Proxy Statement/Consent Solicitation Statement/Prospectus in the section entitled "Management After the Business Combination—Limitation on Liability and Indemnification of Directors and Officers" beginning on page 273 thereof and that information is incorporated herein by reference. The disclosure set forth in Item 1.01 of this Report under the section entitled "Indemnification Agreements" is incorporated herein by reference.

Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

The disclosure set forth in Item 4.01 of this Report is incorporated herein by reference.

Financial Statements and Supplementary Data

The information set forth in Item 9.01 of this Report is incorporated herein by reference.

Item 3.02 Unregistered Sales of Equity Securities.

At the Closing, the Company consummated the PIPE Investment. The disclosure under Item 2.01 of this Report is incorporated herein by reference.

The Company issued the foregoing securities under Section 4(a)(2) of the Securities Act of 1933, as amended (the "Securities Act."), and/or Rule 506 of Regulation D promulgated under the Securities Act, as a transaction not requiring registration under Section 5 of the Securities Act. The parties receiving the securities represented their intentions to acquire the securities for investment only and not with a view to or for sale in connection with any distribution, and appropriate restrictive legends were affixed to the certificates representing the securities (or reflected in restricted book entry with the Company's transfer agent). The parties also had adequate access, through business or other relationships, to information about the Company.

Item 3.03 Material Modification to Rights of Security Holders.

The information set forth in Item 5.03 of this Report is incorporated herein by reference.

Item 4.01 Changes in Registrant's Certifying Accountant.

(a) Dismissal of independent registered public accounting firm.

On February 10, 2022, the Audit Committee informed Marcum LLP ("Marcum"), the Company's independent registered public accounting firm prior to the Transactions, that Marcum will be dismissed effective following the completion of the Company's review for the three and nine months period ended December 31, 2021, which consists only of the pre-Transactions accounts of GCAC.

The report of Marcum on GCAC's financial statements as of March 31, 2021, and for the fiscal years ended March 31, 2021 and March 31, 2020, did not contain an adverse opinion or a disclaimer of opinion, and were not qualified or modified as to uncertainties, audit scope or accounting principles.

During the fiscal years ended March 31, 2021 and March 31, 2020, and the subsequent period through February 10, 2022, there were no disagreements with Marcum on any matter of accounting principles or practices, financial statement disclosures or audited scope or procedures, which disagreements if not resolved to Marcum's satisfaction would have caused Marcum to make reference to the subject matter of the disagreement in connection with its report. During the fiscal years ended March 31, 2021 and March 31, 2020, and the subsequent period through February 10, 2022, there were no reportable events as defined in Item 304(a)(1)(v) of Regulation S-K under the Exchange Act, other than the (i) material weakness in internal controls identified by GCAC's management, in consultation with its advisors, related to an error made in certain of its previously issued financial statements, arising from the manner in which, as of the closing of GCAC's initial public offering, GCAC had an error in the accounting of its Class A common stock subject to possible redemption and public and private warrants, which resulted in the revision of GCAC's Previously Issued Financial Statements (as defined herein), as described in Item 9A. Controls and Procedures in GCAC's Annual Report on Form 10-K for the period ended March 31, 2021, filed with the SEC on July 19, 2021, and GCAC's Amendment No. 1 to Form 10-K, filed with the SEC on December 14, 2021 (the "10-K"), Item 4. Controls and Procedures in GCAC's Quarterly Report on Form 10-Q for the quarterly period ended June 30, 2021, filed with the SEC on August 19, 2021 ("Q1 Form 10-Q"), and in Item 4. Controls and Procedures in GCAC's Quarterly Report on Form 10-Q for the quarterly period ended September 30, 2021, filed with the SEC on November 23, 2021 ("Q2 Form 10-Q"), and (ii) the determination of the audit committee of the board of directors of GCAC that GCAC's Previously Issued Financial Statements could no longer be relied upon due to the accounting errors described above. As used herein, "Previously Issued Financial Statements" consists of: (i) audited balance sheet as of February 2, 2021 filed as Exhibit 99.1 to GCAC's Current Report on Form 8-K filed with the SEC on February 4, 2021, (ii) audited financial statements as of March 31, 2021 contained in the 10-K, (iii) unaudited financial statements as of June 30, 2021 contained in Q1 Form 10-Q; and (iv) unaudited financial statements as of September 30, 2021 contained in Q2 Form 10-Q.

The Company provided Marcum with a copy of the foregoing disclosures prior to the filing of this Report and requested that Marcum furnish a letter addressed to the SEC, which is attached hereto as Exhibit 16.1, stating whether it agrees with such disclosures, and, if not, stating the respects in which is does not agree.

$\begin{tabular}{ll} \textbf{(b) Disclosures regarding the new independent auditor.} \end{tabular}$

On the Closing Date, the Audit Committee approved the engagement of KPMG LLP ("KPMG") as the Company's independent registered public accounting firm to audit the Company's consolidated financial statements for the year ending December 31, 2022. KPMG served as the independent registered public accounting firm of Legacy Cepton prior to the Transactions, including the audit of the consolidated financial statements of Legacy Cepton for the fiscal year ending December 31, 2021. During the fiscal years ended March 31, 2020 and March 31, 2021, GCAC did not consult with KPMG with respect to (i) the application of accounting principles to a specified transaction, either completed or proposed, the type of audit opinion that might be rendered on GCAC's financial statements, and neither a written report nor oral advice was provided to GCAC that KPMG concluded was an important factor considered by GCAC in reaching a decision as to any accounting, auditing or financial reporting issue, or (ii) any other matter that was the subject of a disagreement or a reportable event (each as defined above).

Item 5.01 Changes in Control of the Registrant.

The information set forth above under the "Introductory Note" above and in the section entitled "Security Ownership of Certain Beneficial Owners and Management" in Item 2.01 of this Report is incorporated herein by reference.

Holders of uncertificated shares of GCAC Class A common stock immediately prior to the Closing have continued as holders of uncertificated shares of the Company's common stock. Holders of GCAC Class A common stock who have filed reports under the Exchange Act with respect to those shares should indicate in their next filing, or any amendment to a prior filing, filed on or after the Closing Date that the Company is the successor to GCAC.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

The information set forth above in the sections titled "Directors and Officers," "Executive Compensation," "Certain Relationships and Related Transactions" and "Indemnification of Directors and Officers" in Item 2.01 of this Report is incorporated herein by reference.

As previously disclosed, at the Special Meeting, the stockholders of GCAC considered and approved the 2022 Plan, which is intended to replace the 2016 Plan, and ESPP which became effective immediately upon the Closing. A description of the 2022 Plan and ESPP is included in the Proxy Statement/Consent Solicitation Statement/Prospectus in the sections entitled "*The Incentive Plan Proposal (Proposal 4)*" and "*The ESPP Proposal (Proposal 5)*" beginning on pages 175 and 181, respectively, thereof, which is incorporated herein by reference.

The foregoing description of the 2022 Plan, ESPP and 2016 Plan is qualified in its entirety by the full text of the 2022 Plan and ESPP, which are attached hereto as Exhibit 10.9, Exhibit 10.10 and Exhibit 10.11, respectively, and incorporated herein by reference.

On the Closing Date, the employment agreements previously entered into with Jun Pei, Winston Fu, Mark McCord, Liqun Han, and Dongyi Liao became effective. Reference is made to the disclosure in the section titled "Executive and Director Compensation of Cepton—Executive Employment Agreements" beginning on page 276 of the Proxy Statement/Consent Solicitation Statement/Prospectus, respectively, which are incorporated herein by reference. The foregoing description of the employment agreements is qualified in its entirety by the full text of the employment agreements of Jun Pei, Winston Fu, Mark McCord, Liqun Han and Dongyi Liao, copies of which are attached hereto as Exhibit 10.12, Exhibit 10.13, Exhibit 10.14, Exhibit 10.15 and Exhibit 16, respectively, and incorporated herein by reference.

Item 5.03 Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

Amendments to Certificate of Incorporation and Bylaws

On February 10, 2022, in connection with the consummation of the Transactions, the Company's A&R Charter, and Amended and Restated Bylaws (the "A&R Bylaws") were approved by GCAC's stockholders at the Special Meeting and became effective.

Copies of the A&R Charter and the A&R Bylaws are attached as Exhibit 3.1 and Exhibit 3.2, respectively, to this Report, and are incorporated herein by reference.

The material terms of each of the A&R Charter and the A&R Bylaws and the general effect upon the rights of holders of the Company's capital stock are included in the Proxy Statement/Consent Solicitation Statement/Prospectus under the sections titled "The Amended and Restated Charter Proposal (Proposal 2)," "Description of Securities of GCAC" and "Comparison of Stockholder Rights" beginning on pages 170, 242, and 252 of the Proxy Statement/Consent Solicitation Statement/Prospectus, respectively, which are incorporated herein by reference.

Change in Fiscal Year

Prior to the consummation of the Transactions, GCAC had a March 31 fiscal year end and Legacy Cepton had a December 31 fiscal year end. Effective upon the Closing, the Company changed its fiscal year end to December 31.

Item 5.05 Amendments to the Registrant's Code of Ethics, or Waiver of a Provision of the Code of Ethics.

On the Closing Date, the Board adopted a new a Code of Conduct and Ethics (the "<u>Revised Code</u>"). The Revised Code applies to all employees, executive officers and directors of the Company. The Revised Code was adopted to reflect what the Company considers to be current best practices and policies for an operating company and to make certain technical, administrative, non-substantive amendments to the prior Code of Conduct and Ethics. The adoption of the Revised Code did not relate to or result in any waiver, explicit or implicit, of any provision of the prior Code of Conduct and Ethics.

The above description of the Revised Code does not purport to be complete and is qualified in its entirety by reference to the full text of the Revised Code, a copy of which is filed as Exhibit 14.1 hereto and incorporated herein by reference. The Revised Code is also available on the Investor section of the Company's website at www.cepton.com." The contents of the Company's website are not incorporated by reference in this Report or made a part hereof for any purpose.

Item 5.06 Change in Shell Company Status.

The disclosure set forth in the "Introductory Note" above is incorporated by reference into this Item 5.06.

As a result of the Business Combination, the Company ceased to be a shell company. Reference is made to the disclosure in Item 2.01 of this Report and the Proxy Statement/Consent Solicitation Statement/Prospectus in the sections entitled "*The Business Combination Proposal (Proposal 1)*" beginning on page 133 thereof, which is incorporated herein by reference.

Item 8.01 Other Events.

On February 10, 2022, the parties issued a joint press release announcing the completion of the Business Combination, a copy of which is furnished as Exhibit 99.1 hereto.

Item 9.01 Financial Statement and Exhibits.

(a) Financial Statements of Businesses Acquired.

The unaudited interim condensed consolidated financial statements of Legacy Cepton as of and for the nine months ended September 30, 2021 and September 30, 2020, and the related notes are included in the Proxy Statement/Consent Solicitation Statement/Prospectus beginning on page F-44 of the Proxy Statement/Consent Solicitation Statement/Prospectus and are incorporated herein by reference.

The consolidated financial statements of Legacy Cepton as of and for the years ended December 31, 2020 and December 31, 2019 and the related notes are included in the Proxy Statement/Consent Solicitation Statement/Prospectus beginning on page F-62 of the Proxy Statement/Consent Solicitation Statement/Prospectus and are incorporated herein by reference.

(b) Pro forma financial information.

The unaudited pro forma condensed combined financial information of the Company for the year ended December 31, 2020, as of September 30, 2021 and for the nine months ended September 30, 2021 are filed as Exhibit 99.2 hereto and incorporated herein by reference.

(d) Exhibits.

			Incorporated By Reference		
Exhibit No.	Description	Form	Exhibit	Filing Date	
2.1	Business Combination Agreement, dated as of August 4, 2021, by and among GCAC, Merger Sub and	8-K	2.1	8/5/2021	
	Legacy Cepton.			0,0,00	
2.2	Amendment to Business Combination Agreement, dated as of January 21, 2022, by and among GCAC, Merger Sub and Legacy Cepton.	8-K	2.1	1/24/2022	
3.1	Second Amended and Restated Certificate of Incorporation.				
3.2	Amended and Restated Bylaws.				
4.1	Form of Common Stock Certificate of the Company.				
4.2	Form of Warrant Certificate of the Company (included in Exhibit 4.3).	8-K	4.1	2/4/2021	
4.3	Warrant Agreement, dated January 29, 2021, by and between the Company and Continental Stock Transfer	8-K	4.1	2/4/2021	
	<u>& Trust Company, as warrant agent.</u>				
4.4	Warrant to Purchase Stock for Legacy Cepton	8-K	99.4	1/5/2022	
10.1	Form of PIPE Subscription Agreement.	8-K	10.6	8/5/2021	
10.2	Form of PIPE Subscription Agreement Amendment	8-K	10.1	2/9/2022	
10.3	Form of Stockholder Support Agreement by and among GCAC, Legacy Cepton and the stockholders of	8-K	10.1	8/5/2021	
10.4	Legacy Cepton party thereto.	0.17	10.2	0/5/2021	
10.4	Form of Amended Registration Rights Agreement, by and among GCAC, the Initial Holders and the Legacy Cepton stockholders party thereto.	8-K	10.2	8/5/2021	
10.5	Form of Confidentiality and Lock-Up Agreement, by and between GCAC and the stockholder of Legacy Cepton party thereto.	8-K	10.3	8/5/2021	
10.6	Form of Expenses and Lock-Up Agreement, by and between GCAC and the Legacy Cepton stockholders	8-K	10.4	8/5/2021	
10.7	party thereto. Form of GCAC Stockholder Support Agreement by and between Legacy Cepton and GCAC stockholders.	8-K	10.5	8/5/2021	
10.8++	Form of Director and Officer Indemnification Agreement.				
10.9++	Cepton, Inc. 2022 Equity Incentive Plan.				
10.10++	Cepton, Inc. Employee Stock Purchase Plan.				
10.11++	Legacy Cepton Stock Incentive Plan.				
10.12++	Employment Agreement, dated as of December 7, 2021, by and between Legacy Cepton and Jun Pei.	S-4/A	10.26	12/17/2021	
10.13++	Employment Agreement, dated as of December 7, 2021, by and between Legacy Cepton and Winston Fu.	S-4/A	10.27	12/17/2021	
10.14++	Employment Agreement, dated as of December 7, 2021, by and between Legacy Cepton and Mark McCord.	S-4/A	10.28	12/17/2021	
10.15++	Employment Agreement, dated as of December 7, 2021, by and between Legacy Cepton and Liqun Han.	S-4/A	10.29	12/17/2021	
10.16++	Employment Agreement, dated as of December 7, 2021, by and between Legacy Cepton and Dongyi Liao.	S-4/A	10.30	12/17/2021	
10.17	Purchase Agreement, dated as of November 24, 2021, by and among GCAC, Legacy Cepton and Lincoln Park.	8-K	10.1	11/29/2021	
10.18	Registration Rights Agreement, dated as of November 24, 2021, by and among GCAC, Legacy Cepton and Lincoln Park.	8-K	10.2	11/29/2021	
10.19	Supplier On Board Agreement, dated April 14, 2020, by and between Legacy Cepton and Koito	S-4/A	10.23	1/14/2022	
10.15	Manufacturing Co., Ltd.#	5 1/11	10.25	1/11/2022	
10.20	Software Development Work Order, dated September 22, 2021 by Koito Manufacturing Co., Ltd.#	S-4/A	10.31	1/14/2022	
10.21	Loan Agreement, dated January 4, 2022, by and between Legacy Cepton and Trinity Capital Inc.	8-K	99.1	1/5/2022	
10.22	Pledge Agreement, dated January 4, 2022, by and between Legacy Cepton and Trinity Capital Inc.	8-K	99.2	1/5/2022	
10.23	Participation Rights Agreement, dated January 4, 2022, by and between Legacy Cepton and Trinity Capital	8-K	99.3	1/5/2022	
	<u>Inc.</u>				
14.1	Cepton, Inc. Code of Conduct and Ethics				
16.1	Letter from Marcum LLP to the SEC, dated February 10, 2022.				
21.1	<u>List of Subsidiaries</u>				
99.1	Joint Press Release, dated as of February 10, 2022.				
99.2	Unaudited pro forma condensed combined financial information of the Company for the year ended December 31, 2020, as of September 30, 2021 and for the nine months ended September 30, 2021.				
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)				
	0				

⁺⁺ Indicates a management or compensatory plan.
Portions of exhibit have been omitted.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this Report to be signed on its behalf by the undersigned hereunto duly authorized.

CEPTON, INC.

Date: February 10, 2022 By: /s/ Jun Pei

Name: Jun Pei

Title: President and Chief Executive Officer

SECOND AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF GROWTH CAPITAL ACQUISITION CORP.

Pursuant to the provisions of § 242 and § 245 of the General Corporation Law of the State of Delaware

Growth Capital Acquisition Corp. (the "Corporation"), a corporation organized and the General Corporation Law of the State of Delaware (the "DGCL"), does hereby certify as follows:

FIRST: The present name of the Corporation is Growth Capital Acquisition Corp. The original certificate of incorporation of the Corporation was filed with the Secretary of State of the State of Delaware on January 4, 2010 under the name PinstripesNYS, Inc. (the "Original Certificate"). The first certificate of amendment of the Original Certificate was filed with the Secretary of State of the State of Delaware on February 14, 2020. The second certificate of amendment of the Original Certificate was filed with the Secretary of State of the State of Delaware on February 24, 2020. The third certificate of amendment of the Original Certificate was filed with the Secretary of State of the State of Delaware on February 25, 2020. The Original Certificate was amended and restated in its entirety by the Amended and Restated Certificate of Incorporation of the Corporation filed with the Secretary of State of the State of Delaware on January 29, 2021 (the "A&R Certificate").

SECOND: The certificate of incorporation of the Corporation as heretofore in effect is hereby amended and restated in its entirety as set forth in this Second Amended and Restated Certificate of Incorporation hereinafter provided for (the "Certificate of Incorporation").

THIRD: This Certificate of Incorporation herein certified has been has been duly adopted by the Corporation in accordance with Sections 242 and 245 of the DGCL and has been adopted by the requisite vote of the stockholders of the Corporation in accordance with the DGCL.

FOURTH: This Certificate of Incorporation shall become effective upon the filing of this Certificate of Incorporation with the Secretary of State of the State of Delaware.

FIFTH: This Certificate is being amended and restated in connection with the transactions contemplated by that certain Business Combination Agreement, dated August 4, 2021 (the "Business Combination Agreement"), by and among the Corporation, GCAC Merger Sub Inc. and Cepton Technologies, Inc. As part of the transactions contemplated by the Business Combination Agreement, and in accordance with the A&R Certificate, all shares of outstanding Class B Common Stock of the Corporation shall automatically be converted, on a one-to-one basis, into shares of Class A Common Stock of the Corporation such that, at the effectiveness of this Certificate of Incorporation, only Class A Common Stock remains outstanding. All Class A Common Stock issued and outstanding prior to the effectiveness of this Certificate of Incorporation and all Class A Common Stock issued as part of the Business Combination Agreement, including all Class A Common Stock issued pursuant to the PIPE Investment (as defined in the Business Combination Agreement), shall be renamed as Common Stock for all purposes of this Certificate of Incorporation.

SIXTH: This Certificate Incorporation of the Corporation shall, at the effective time, read as follows:

ARTICLE I.

Name

The name of the Corporation is Cepton, Inc.

ARTICLE II.

Registered Office

The address of the Corporation's registered office in the State of Delaware is 251 Little Falls Drive, in the City of Wilmington, County of New Castle, State of Delaware, 19808, and the name of the Corporation's registered agent at such address is Corporation Service Company.

ARTICLE III.

Purpose and Powers

The purpose of the Corporation is to engage in any lawful act or activity for which a corporation may be organized under the DGCL.

ARTICLE IV.

Capital Stock

(A) <u>Authorized Capital Stock.</u>

The total number of shares of all classes of capital stock which the Corporation shall have authority to issue is three hundred fifty five million (355,000,000) shares of capital stock, consisting of (i) three hundred fifty million (350,000,000) shares of common stock, par value \$0.00001 per share (the "Common Stock"), and (ii) five million (5,000,000) shares of preferred stock, par value \$0.00001 per share (the "Preferred Stock").

Upon the effectiveness of the filing of this Certificate of Incorporation with the Secretary of State of the State of Delaware (the "Effective Time"), each share of the Corporation's Class A Common Stock, par value \$0.0001 per share (the "Old Class A Common Stock"), that was issued and outstanding or held in treasury immediately prior to the Effective Time (including for avoidance of doubt each share of Old Class A Common Stock into which shares of the Corporation's Class B Common Stock, par value \$0.0001 per share (the "Old Class B Common Stock"), automatically converted prior to the Effective Time) will automatically be reclassified as one (1) share of Common Stock (the "New Common Stock"). Any stock certificate that, immediately prior to the Effective Time, represented shares of Old Class A Common Stock (or shares of Old Class B Common Stock that converted into Old Class A Common Stock prior to the Effective Time) will, from and after the Effective Time, automatically and without the necessity of presenting the same for exchange, represent the number of shares of New Common Stock as such shares of Old Class A Common Stock have been reclassified and converted into pursuant to the immediately preceding sentence; provided that, upon surrender of any such certificate to the Corporation for exchange, the Corporation shall issue a new certificate for the shares of New Common Stock previously represented thereby.

Notwithstanding anything to the contrary contained herein, the rights and preferences of the Common Stock shall at all times be subject to the rights and preferences of the Preferred Stock as may be set forth in the Certificate of Incorporation or one or more certificates of designations filed with the Secretary of State of the State of Delaware from time to time in accordance with the DGCL and this Certificate of Incorporation. The number of authorized shares of Common Stock or Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) from time to time by the affirmative vote of the holders of at least a majority of the voting power of the Corporation's then outstanding shares of capital stock entitled to vote thereon, voting together as a single class, irrespective of the provisions of Section 242(b)(2) of the DGCL (or any successor provision thereto), and no vote of the holders of the Common Stock or the Preferred Stock voting separately as a class or series shall be required therefor unless a vote of any such holder is required pursuant to this Certificate of Incorporation (including any certificate of designation relating to any series of Preferred Stock).

(B) Common Stock.

The voting powers, designations, preferences and relative, participating, optional or other special rights and such qualifications, limitations or restrictions of the Common Stock, in addition to those set forth elsewhere herein, are as follows:

- (1) Voting Rights. Each holder of shares of Common Stock shall be entitled to vote at all meetings of the stockholders and to cast one vote for each outstanding share of Common Stock held by such holder on all matters on which stockholders are entitled to vote generally. Notwithstanding the foregoing, except as otherwise required by law, holders of Common Stock shall not be entitled to vote on any amendment to this Certificate of Incorporation (including any certificate of designation relating to any series of Preferred Stock) that relates solely to the terms of one or more outstanding series of Preferred Stock if the holders of such affected series are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to this Certificate of Incorporation (including any certificate of designation relating to any series of Preferred Stock) or pursuant to the DGCL.
- (2) Dividends and Distributions. Subject to the prior rights of the holders of all series of Preferred Stock at the time outstanding having prior rights or preferences as to dividends or other distributions, the holders of shares of Common Stock shall be entitled to receive, when and as declared by the Board of Directors, out of the assets of the Corporation legally available therefor, such dividends and other distributions as may be declared from time to time by the Board of Directors and shall share equally on a per share basis in all such dividends and other distributions.
- (3) Liquidation. Subject to the prior rights of creditors of the Corporation, including without limitation the payment of expenses relating to any liquidation, dissolution or winding up of the Corporation, and the holders of all series of Preferred Stock at the time outstanding having prior rights as to distributions upon liquidation, dissolution or winding up of the Corporation, in the event of any liquidation, dissolution or winding up of the Corporation, either voluntary or involuntary, the holders of shares of Common Stock shall be entitled to receive their ratable and proportionate share of the remaining assets of the Corporation. A merger or consolidation of the Corporation with any other corporation or other entity, or a sale or conveyance of all or any part of the assets of the Corporation (which shall not in fact result in the liquidation of the Corporation and the distribution of assets to its stockholders) shall not be deemed to be a voluntary or involuntary liquidation, dissolution or winding up of the Corporation.

(C) Preferred Stock.

The Board of Directors is hereby expressly authorized, without any action or vote by the Corporation's stockholders (except as may otherwise be provided by the terms of any series of Preferred Stock then outstanding), to provide for the issuance of all or any shares of the Preferred Stock in one or more series of Preferred Stock, and to fix for each such series the voting powers, if any, designations, preferences and relative, participating, optional or other rights and qualifications, limitations or restrictions thereof, if any, as shall be stated and expressed in the resolution or resolutions adopted by the Board of Directors providing for the issuance of such series and the number of shares constituting each such series, and to increase or decrease the number of shares of any such series to the extent permitted by the DGCL.

ARTICLE V.

Board of Directors

(A) Powers of the Board of Directors.

Except as otherwise provided by the DGCL, the business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors.

(B) Number of Directors.

Subject to any rights of the holders of Preferred Stock to elect directors, the Board of Directors shall consist of one or more members, the exact number of which shall be fixed by, or in the manner provided in, the Corporation's Amended and Restated Bylaws (as may be further amended, restated, modified or supplemented from time to time, the "<u>Bylaws</u>").

(C) <u>Classification of the Board of Directors.</u>

The directors of the Corporation (other than those directors elected by the holders of any series or class of Preferred Stock provided for or fixed pursuant to the provisions of Article IV hereof (the "Preferred Stock Directors")) shall be and are divided into three (3) classes, designated Class A, Class B and Class C. Each class shall consist, as nearly as may be possible, of one-third (1/3) of the total number of directors constituting the entire Board of Directors. The Board of Directors may assign members of the Board of Directors already in office at the Effective Time to such classes. Subject to the rights of holders of any series or class of Preferred Stock to elect directors, each director shall serve for a term ending on the date of the third annual meeting of stockholders following the annual meeting of stockholders at which such director was elected; provided that each director initially assigned to Class A shall serve for a term expiring at the Corporation's first annual meeting of stockholders held after the Effective Time; each director initially assigned to Class B shall serve for a term expiring at the Corporation's second annual meeting of stockholders held after the Effective Time; and each director initially assigned to Class C shall serve for a term expiring at the Corporation's third annual meeting of stockholders held after the Effective Time; provided further, that the term of each director shall continue until the election and qualification of his or her successor and be subject to his or her earlier death, disqualification, resignation or removal. If the number of such directors is changed, any increase or decrease shall be apportioned among the classes so as to maintain the number of directors in each class as nearly equal as possible, and any such additional director of any class elected to fill a newly created directorship resulting from an increase in such class shall hold office for a term that shall coincide with the remaining term of that class, but in no case shall a decrease in

(D) Removal of Directors.

Except for any Preferred Stock Director, any director or the entire Board of Directors may be removed from office at any time, but only for cause and only by the affirmative vote of the holders of at least sixty-six and two-thirds percent (66 2/3%) of the total voting power of the outstanding shares of capital stock entitled to vote in the election of directors, voting together as a single class.

(E) <u>Vacancies.</u>

Subject to the rights of any series of Preferred Stock then outstanding, any vacancy on the Board of Directors, by reason of death, resignation, retirement, disqualification or removal or otherwise, and any newly created directorship that results from an increase in the number of directors, shall be filled only by a majority of the Board of Directors then in office, even if less than a quorum, or by a sole remaining director. Any director of any class elected to fill a vacancy resulting from an increase in the number of directors of such class shall hold office for a term that shall coincide with the remaining term of that class. Any director elected to fill a vacancy not resulting from an increase in the number of directors shall have the same remaining term as that of his or her predecessor.

(F) <u>Preferred Stock Directors.</u>

During any period when the holders of any series of Preferred Stock have the right to elect Preferred Stock Directors, then upon commencement and for the duration of the period during which such right continues: (i) the then otherwise total authorized number of directors of the Corporation shall automatically be increased by such specified number of directors, and the holders of such Preferred Stock shall be entitled to elect the additional directors so provided for or fixed pursuant to said provisions, and (ii) each such additional director shall serve until such director's successor shall have been duly elected and qualified, or until such director's right to hold such office terminates pursuant to said provisions, whichever occurs earlier, subject to his or her earlier death, disqualification, resignation or removal. Except as otherwise provided by the Board of Directors in the resolution or resolutions establishing such series, whenever the holders of any series of Preferred Stock having such right to elect Preferred Stock Directors are divested of such right pursuant to the provisions of such stock, the terms of office of all such Preferred Stock

Directors elected by the holders of such stock, or elected to fill any vacancies resulting from the death, resignation, disqualification or removal of such additional directors, shall forthwith terminate and the total authorized number of directors of the Corporation shall be reduced accordingly.

(G) Powers and Authority.

In addition to the powers and authority expressly conferred upon them herein or by statute, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation, subject, nevertheless, to the provisions of the DGCL and this Certificate of Incorporation.

ARTICLE VI.

Stockholder Action

(A) Election of Directors.

Elections of directors need not be by written ballot except and to the extent provided in the Bylaws.

(B) Advance Notice.

Advance notice of nominations for the election of directors or proposals or other business to be considered by stockholders, which are made by any stockholder of the Corporation, shall be given in the manner and to the extent provided in the Bylaws.

(C) Stockholder Action by Written Consent.

Any action required or permitted to be taken by the stockholders of the Corporation may be effected only at a duly called annual or special meeting of stockholders of the Corporation and may not be effected by any consent in writing by such stockholders in lieu of a meeting of stockholders.

Notwithstanding the foregoing, any action required or permitted to be taken by the holders of Preferred Stock, voting separately as a series or separately as a class with one or more other such series, may be taken without a meeting, without prior notice and without a vote, to the extent expressly so provided by the applicable certificate of designation relating to such series of Preferred Stock.

ARTICLE VII.

Limitation of Director Liability; Indemnification

- (A) No director of the Corporation shall be personally liable to the Corporation or to any of its stockholders for monetary damages for breach of fiduciary duty as a director, except to the extent such elimination from liability or limitation thereof is not permitted under the DGCL as the same exists or may hereafter be amended. If the DGCL is amended hereafter to authorize the further elimination or limitation of the liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent authorized by the DGCL, as so amended. Any repeal or modification of this Article VII, because of amendments or modifications of the DGCL or otherwise, shall not adversely affect any right or protection of a director of the Corporation existing at the time of such repeal or modification with respect to acts or omissions occurring prior to the effective date of such repeal or modification.
- (B) To the fullest extent permitted by applicable law, as the same exists or may hereafter be amended, the Corporation shall indemnify and hold harmless each person who is or was made a party or is threatened to be made a party to or is otherwise involved in any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (a "proceeding") by reason of the fact that he or she is or was a director or officer of the Corporation or, while a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, other enterprise or nonprofit entity, including service with respect to an employee benefit plan (an "indemnitee"), whether the basis of such proceeding is alleged action in an official capacity as a director, officer, employee or agent, or in any other capacity while serving as a director, officer, employee or agent, against all liability and loss suffered and expenses (including, without limitation, attorneys' fees, judgments, fines, taxes and penalties and amounts paid in settlement) reasonably incurred by such indemnitee in connection with such proceeding. The Corporation shall to the

fullest extent not prohibited by applicable law pay the expenses (including attorneys' fees) incurred by an indemnitee in defending or otherwise participating in any proceeding in advance of its final disposition; provided, however, that, to the extent required by applicable law, such payment of expenses in advance of the final disposition of the proceeding shall be made only upon receipt of an undertaking, by or on behalf of the indemnitee, to repay all amounts so advanced if it shall ultimately be determined that the indemnitee is not entitled to be indemnified under this Section 8(B) or otherwise. The rights to indemnification and advancement of expenses conferred by this Section 8(B) shall be contract rights and such rights shall continue as to an indemnitee who has ceased to be a director, officer, employee or agent and shall inure to the benefit of his or her heirs, executors and administrators.

ARTICLE VIII.

Amendment of Bylaws

In furtherance and not in limitation of the powers conferred upon it by the laws of the State of Delaware, the Board of Directors shall have the power to adopt, amend, alter or repeal the Bylaws by the affirmative vote of a majority of the entire Board of Directors (assuming no vacancies on the Board of Directors). The Bylaws may also be adopted, amended, altered or repealed by the affirmative vote of at least sixty-six and two-thirds percent (66 2/3%) of the total voting power of the Corporation's issued and outstanding shares of capital stock entitled to vote generally in the election of directors, voting together as a single class.

ARTICLE IX.

Amendment of Certificate of Incorporation

The Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation in the manner now or hereafter prescribed in this Certificate of Incorporation or the DGCL, and all rights herein conferred upon stockholders are granted subject to such reservation; provided, however, that, notwithstanding any other provision of this Certificate of Incorporation (and in addition to any other vote that may be required by law), the affirmative vote of the holders of at least sixty-six and two-thirds percent (66 2/3%) of the voting power of the Corporation's issued and outstanding shares of capital stock entitled to vote generally in the election of directors, voting together as a single class, shall be required to amend, alter, change or repeal or to adopt any provision of this Certificate of Incorporation inconsistent with any provision of Article VI, Article VII, Article VIII, this Article IX, or Article X.

ARTICLE X.

Corporation Opportunity

In the event that a member of the Board of Directors who is not an employee of the Corporation or its subsidiaries, or any employee or agent of such member, other than someone who is an employee of the Corporation or its subsidiaries (collectively, the "Covered Persons"), acquires knowledge of any business opportunity matter, potential transaction, interest or other matter, unless such matter, transaction or interest is presented to, or acquired, created or developed by, or otherwise comes into the possession of, a Covered Person expressly and solely in connection with such individual's service as a member of the Board of Directors of the Corporation (a "Corporate Opportunity"), then the Corporation to the maximum extent permitted from time to time under the DGCL (including Section 122(17) thereof): (a) renounces any expectancy that such Covered Person offer an opportunity to participate in such Corporate Opportunity to the Corporation; and (b) waives any claim that such opportunity constituted a Corporate Opportunity that should have been presented by such Covered Person to the Corporation or any of its affiliates. No amendment or repeal of this paragraph shall apply to or have any effect on the liability or alleged liability of any officer, director or stockholder of the Corporation for or with respect to any opportunities of which such officer, director or stockholder becomes aware prior to such amendment or repeal.

ARTICLE XI.

Forum Selection

Unless the Corporation consents in writing to the selection of an alternative forum, (A) (i) any derivative action or proceeding brought on behalf of the Corporation, (ii) any action asserting a claim of breach of a fiduciary duty owed by any current or former director, officer, other employee or stockholder of the Corporation to the Corporation or the Corporation's stockholders, (iii) any action asserting a claim arising pursuant to any provision of the DGCL, this Certificate of Incorporation or the Bylaws (as either may be amended or restated) or as to which the DGCL confers jurisdiction on the Court of Chancery of the State of Delaware or (iv) any action asserting a claim governed by the internal affairs doctrine of the law of the State of Delaware shall, to the fullest extent permitted by law, be exclusively brought in the Court of Chancery of the State of Delaware or, if such court does not have subject matter jurisdiction thereof, the federal district court of the State of Delaware; and (B) the federal district courts of the United States of America shall, to the fullest extent permitted by applicable law, be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act of 1933, as amended.

* * *

IN WITNESS WHEREOF, the Corporation has caused this Certificate of Incorporation to be executed on its behalf this 10th day of February 2022.

GROWTH CAPITAL ACQUISITION CORP.

By: /s/ George Syllantavos

Name: George Syllantavos

Title: Co-Chief Executive Officer

8

AMENDED AND RESTATED BYLAWS

OF

CEPTON, INC.

A Delaware Corporation

Effective February 10, 2022

Table of Contents

		Annex C Pages
ARTICLE I	OFFICES	1
1.1	Principal Executive Office	1
1.2	Registered Office	1
1.3	Other Offices	1
ARTICLE II	STOCKHOLDERS' MEETINGS	1
2.1	Place of Meetings	1
2.2	Annual Meetings	1
2.3	Special Meetings	1
2.4	Notice	1
2.5	Adjournments	2
2.6	Quorum	2
2.7	Voting	2
2.8	Participation at Stockholder Meetings by Remote Communications	2
2.9	Proxies	3
2.10	No Stockholder Action by Written Consent	3
2.11	Record Date	3
2.12	Stockholders' List	3
2.13	Conduct of Meetings	4
2.14	Advance Notice of Stockholder Business and Director Nominations	4
ARTICLE III	DIRECTORS	7
3.1	Powers and Duties	7
3.2	Number and Qualifications	8
3.3	Classified Board of Directors	8
3.4	Resignations and Removals of Directors	8
3.5	Vacancies	8
3.6	Regular Meetings	8
3.7	Special Meetings	8
3.8	Organization	8
3.9	Meetings by Means of Conference Telephone	9
3.10	Quorum	9
3.11	Action of the Board by Written Consent	9
3.12	Expense Reimbursement and Compensation	9
3.13	Chairman and Vice Chairman of the Board	9
3.14	Committees	9
3.15	Telephonic Meetings	10
ARTICLE IV	OFFICERS	10
4.1	General	10
4.2	Appointment and Term	10
4.3	Resignations	10
4.4	Vacancies	10
4.5	Compensation	10
4.6	Authority and Duties of Officers	10
ARTICLE V	STOCK	11
5.1	Certificates	11
5.2	Transfers	11
5.3	Lost, Stolen, or Destroyed Certificates	11
5.4	Record Owners	11

		Annex C Pages
ARTICLE VI	NOTICES	11
6.1	Notices	11
6.2	Waivers of Notice	12
ARTICLE VII	INDEMNIFICATION AND ADVANCEMENT OF EXPENSES	12
7.1	Definitions	12
7.2	Indemnification	12
7.3	Determination	12
7.4	Expenses Payable in Advance	13
7.5	Claim	13
7.6	Other Indemnification of Advancement	13
7.7	Insurance	13
ARTICLE VIII	GENERAL PROVISIONS	13
8.1	Fiscal Year	13
8.2	Corporate Seal	13
8.3	Maintenance and Inspection of Records	13
8.4	Reliance Upon Books, Reports and Records	14
8.5	Dividends	14
8.6	Emergency Bylaws	14
8.7	Certificate of Incorporation Governs	14
8.8	Severability	14
8.9	Actions with Respect to Securities of Other Entities	14
ARTICLE IX	AMENDMENTS	14
9.1	Amendments	14

AMENDED AND RESTATED BYLAWS OF CEPTON, INC.

ARTICLE I

OFFICES

- 1.1 <u>Principal Executive Office</u>. The principal executive office of Cepton, Inc. (the "<u>Corporation</u>") shall be at such place established by the Board of Directors of the Corporation (the "<u>Board</u>") in its discretion. The Board shall have full power and authority to change the location of the principal executive office.
- 1.2 <u>Registered Office</u>. The registered office of the Corporation shall be as set forth in the Corporation's Amended and Restated Certificate of Incorporation (as may be amended, restated, modified or supplemented from time to time, the "<u>Certificate of Incorporation</u>").
- 1.3 Other Offices. The Corporation may also have offices at such other places, both within and outside of the State of Delaware, as the Board may from time to time determine.

ARTICLE II

STOCKHOLDERS' MEETINGS

- 2.1 <u>Place of Meetings.</u> Meetings of stockholders shall be held at such place, if any, either within or outside of the State of Delaware, as shall be designated from time to time by the Board and specified in the notice of the meeting. In the absence of such designation, stockholders' meetings shall be held at the principal executive office of the Corporation.
- 2.2 <u>Annual Meetings</u>. The annual meeting of stockholders for the election of directors and for the transaction of such other business as may properly come before the meeting shall be held at such time and date as shall be designated from time to time by the Board and stated in the Corporation's notice of the meeting. The Board may postpone, reschedule or cancel any previously scheduled annual meeting of stockholders, before or after the notice for such meeting has been sent to the stockholders.
- 2.3 <u>Special Meetings</u>. Special meetings of the stockholders for any purpose or purposes may be called at any time by a resolution adopted by the majority of the Board, and may not be called by any other person or persons. The Board acting pursuant to a resolution may postpone, reschedule or cancel any previously scheduled special meeting of stockholders, before or after the notice for such meeting has been sent to the stockholders. Business transacted at any special meeting of stockholders shall be limited to matters relating to the purpose or purposes stated in the notice of meeting.
- Notice. Whenever stockholders are required or permitted to take any action at a meeting, whether annual or special, a written notice of the meeting shall be given by the Corporation to each stockholder of record entitled to vote at such meeting as of the record date for determining the stockholders entitled to notice of such meeting. Such notice shall state the place, if any, date and hour of the meeting, the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting, the record date for determining the stockholders entitled to vote at the meeting (if such date is different from the record date for stockholders entitled to notice of the meeting), and, in the case of a special meeting, the purpose or purposes for which the meeting was called. Unless otherwise required by law, the Certificate of Incorporation or these Bylaws (as may be further amended, restated, modified or supplemented from time to time, these "Bylaws"), notice of any meeting shall be given not less than ten (10) nor more than sixty (60) days before the date of the meeting to each stockholder of record entitled to notice of and to vote at such meeting.

- 2.5 Adjournments. Any meeting of stockholders, annual or special, whether or not a quorum is present, may be adjourned from time to time for any reason by either the chairman of the meeting, by a resolution adopted by the majority of the Board or in accordance with Section 2.6. Notwithstanding the provisions in Section 2.4 hereof, notice need not be given of any such adjourned meeting if the time, place, if any, and date of the meeting and the means of remote communications, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting and the record date for determining the stockholders entitled to vote at the meeting (if such date is different from the record date for determining stockholders entitled to notice of the meeting) are announced at the meeting at which the adjournment is taken; provided, however, that if the date of any adjourned meeting is more than thirty (30) days after the date for which the meeting was originally called or a new record date is fixed for the adjourned meeting, notice of the adjourned meeting shall be given in conformity with Section 2.4. At such adjourned meeting, any business may be transacted that might have been transacted at the original meeting if such meeting had been held as originally called.
- 2.6 Quorum. Unless otherwise required by applicable law or the Certificate of Incorporation, the holders of a majority in voting power of the outstanding shares of capital stock of the Corporation entitled to vote thereat, present in person or represented by proxy, shall constitute a quorum at all meetings of the stockholders for the transaction of business. If, however, such quorum shall not be present or represented at any meeting of the stockholders, then either the chairman of the meeting or the stockholders entitled to vote thereon, present in person or represented by proxy, shall have power to adjourn the meeting from time to time, in the manner provided in Section 2.5 hereof, until a quorum shall be present or represented. A quorum, once established, shall not be broken by the withdrawal of enough stockholders to leave less than a quorum.

2.7 <u>Voting</u>.

- (a) Unless otherwise required by law or the Certificate of Incorporation, each stockholder entitled to vote at any meeting of stockholders shall be entitled to one (1) vote for each share of stock held by such stockholder which has voting power on all matters submitted to a vote of stockholders of the Corporation.
- (b) Unless otherwise required by law, the Certificate of Incorporation, these Bylaws, the rules or regulations of any stock exchange applicable to the Corporation, or any regulation applicable to the Corporation or its securities, (i) every matter brought before any meeting of the stockholders, other than the election of directors, shall be decided by the affirmative vote of the holders of a majority of the voting power of the shares of stock entitled to vote on such matter that are present in person or represented by proxy at the meeting and are voted for or against the matter, voting as a single class, and (ii) directors shall be elected by vote of the holders of a plurality of the votes cast. Abstentions and broker non-votes shall not be counted as votes for or against the matter. Notwithstanding the foregoing, two or more classes or series of stock shall only vote together as a single class if and to the extent the holders thereof are entitled to vote together as a single class at a meeting. Where a separate vote by class is required, the vote of the holders of a majority in total voting power of each class of Corporation's outstanding capital stock represented at the meeting and entitled to vote on such matter and are voted for or against the matter shall be the act of such class, except as otherwise provided by law, the Certificate of Incorporation or these Bylaws. The Board, in its discretion, or the Chairman of the Board, or the presiding officer of a meeting of the stockholders, in such person's discretion, may require that any votes cast (including election of directors) at such meeting shall be cast by written ballot.
- 2.8 <u>Participation at Stockholder Meetings by Remote Communications.</u> The Board may, in its sole discretion, determine that a meeting of stockholders shall not be held at any place, but may instead be held solely by means of remote communication in accordance with Section 211(a)(2) of the General Corporation Law of the State of Delaware (the "<u>DGCL</u>") or any successor provision. If authorized by the Board in its sole discretion, and subject to such guidelines and procedures as the Board may adopt, stockholders and proxyholders not physically present at a meeting of stockholders may, by means of remote communication, (a) participate in a meeting of stockholders, and (b) be deemed present in person and vote at a meeting of stockholders, whether such meeting is to be held at a designated place or solely by remote communication, <u>provided</u> that (x) the Corporation may implement reasonable measures to verify that each person deemed present and permitted to vote at the meeting by means of remote communication is a stockholder or proxyholder, (y) the Corporation may implement reasonable measures to provide such stockholders and proxyholders a reasonable opportunity to participate in the meeting and to vote on matters submitted to the

stockholders, including an opportunity to read or hear the proceedings of the meeting substantially concurrently with such proceedings, and (z) if any stockholder or proxyholder votes or takes other action at the meeting by means of remote communication, a record of such vote or other action shall be maintained by the Corporation.

- 2.9 <u>Proxies.</u> Each stockholder entitled to vote at a meeting of stockholders has the right to do so either in person or by one or more agents authorized by a proxy, which may be in the form of a telegram, cablegram or other means of electronic transmission, filed with the Secretary of the Corporation, but no such proxy shall be voted or acted upon after three (3) years from its date, unless the proxy provides for a longer period. A duly executed proxy shall be irrevocable if it states that it is irrevocable and if, and only as long as, it is coupled with an interest sufficient in law to support an irrevocable power. A stockholder may revoke any proxy which is not irrevocable by attending the meeting and voting in person or by delivering an instrument in writing stating that the proxy is revoked or by filing another proxy bearing a later date with the Secretary of the Corporation.
- 2.10 No Stockholder Action by Written Consent. Subject to the rights of the holders of any class or series of preferred stock then outstanding, as may be set forth in the certificate of designations for such class or series of preferred stock, any action required or permitted to be taken at any annual or special meeting of stockholders may be taken only upon the vote of stockholders at an annual or special meeting duly noticed and called in accordance with the DGCL and may not be taken by written consent of stockholders without a meeting.

2.11 Record Date.

- (a) In order that the Corporation may determine the stockholders entitled to notice of any meeting of the stockholders or any adjournment thereof, the Board may fix a record date for the determination of the stockholders entitled to notice of any meeting or adjournment thereof. The record date so fixed shall not precede the date upon which the resolution fixing the record date is adopted by the Board, and which record date shall not be more than sixty (60) nor less than ten (10) days before the date of such meeting. If no record date is fixed by the Board, the record date for determining stockholders entitled to notice of or to vote at a meeting of the stockholders shall be at the close of business on the day next preceding the day on which notice is given, or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board may fix a new record date for determination of stockholders entitled to notice of or to vote at the adjourned meeting.
- (b) In order that the Corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment of any rights, or to exercise rights in respect of any change, conversion or exchange of stock or in respect of any other lawful action, the Board may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board, and which record date shall not be more than sixty (60) days prior to such action. If no such record date is fixed, the record date for determining the stockholders for any such purpose shall be at the close of business on the date on which the Board adopts the resolution relating thereto.
- 2.12 Stockholders' List. A complete list of the stockholders entitled to vote at any meeting of stockholders (provided, however, if the record date for determining the stockholders entitled to vote is less than ten (10) days before the date of the meeting, the list shall reflect the stockholders entitled to vote as of the tenth day before the meeting date), arranged in alphabetical order and showing the address of each stockholder, and the number of shares registered in the name of each stockholder, shall be prepared by the officer having charge of the stock ledger. Such list shall be open to examination by any stockholder, for any purpose germane to the meeting, for a period of at least ten (10) days before such meeting (a) on a reasonably accessible electronic network, provided that the information required to gain access to such list is provided with the notice of the meeting, or (b) during ordinary business hours at the principal place of business of the Corporation. If the meeting is to be held at a place, then a list of stockholders entitled to vote at the meeting shall be produced and kept at the time and place of the meeting during the whole time thereof and may be examined by any stockholder who is present. If the meeting is to be held solely by means of remote communication, then the list shall also be open to the examination of any stockholder during the

whole time of the meeting on a reasonably accessible electronic network, and the information required to access such list shall be provided with the notice of the meeting. Except as otherwise provided by law, the stock ledger shall be the only evidence as to who are the stockholders entitled to examine the list required by this Section 2.12 or to vote in person or by proxy at any meeting of stockholders.

- 2.13 <u>Conduct of Meetings</u>. The meetings of the stockholders shall be presided over by the Chairman of the Board, or if he or she is not present, by the Chief Executive Officer, or if neither the Chairman of the Board, nor the Chief Executive Officer is present, by a chairman elected by a resolution adopted by the majority of the Board. The order of business at all meetings of stockholders shall be as determined by the chairman of the meeting.
 - 2.14 Advance Notice of Stockholder Business and Director Nominations.
 - (a) Annual Meetings of Stockholders.
- Nominations of persons for election to the Board and the proposal of other business to be considered by the stockholders may be made at an annual meeting of stockholders only (i) pursuant to the Corporation's notice of meeting (or any supplement thereto) delivered pursuant to Section 2.4 and Article VI hereof, (ii) by or at the direction of the Board or any duly authorized committee thereof, or (iii) by any stockholder of the Corporation who (x) is a stockholder of record at the time of delivery by the stockholder of the notice provided for in Section 2.14(a)(2) to the Secretary of the Corporation and at the time of the annual meeting, (y) who is entitled to vote at the meeting and upon such election, and (z) who complies with the notice procedures set forth in Section 2.14(a)(2); clause (iii) shall be the exclusive means for a stockholder to make nominations or submit other business (other than matters properly brought under Rule 14a-8 under the Securities Exchange Act of 1934, as amended (the "Exchange Act") and included in the Corporation's notice of meeting) before an annual meeting of stockholders. Except as otherwise required by law, any failure to comply with these procedures shall result in the nullification of such nomination or proposal. Notwithstanding the foregoing, if a stockholder is entitled to vote only for a specific class or category of directors at a meeting of the stockholders, such stockholder's right to nominate one or more individuals for the election of a director at the meeting shall be limited to such class or category of directors.
- Without qualification, for any nominations or other business to be properly brought before an annual meeting of stockholders by a stockholder pursuant to clause (iii) of Section 2.14(a)(1), the stockholder must have given timely notice thereof, in proper written form as provided in Section 2.14(c), to the Secretary of the Corporation and any such proposed business (other than nominations of persons for the election to the Board) must constitute a proper matter for stockholder action under the DGCL. To be timely, such a stockholder's notice shall be delivered to the Secretary at the principal executive office of the Corporation not later than the close of business on the ninetieth (90th) day nor earlier than the close of business on the one hundred twentieth (120th) day prior to the first anniversary date of the preceding year's annual meeting (which date shall, for purposes of the Corporation's annual meeting of stockholders in the year of the closing of the merger contemplated by that certain Business Combination Agreement, dated as of August 4, 2021, by and among Growth Capital Acquisition Corp., GCAC Merger Sub Inc., and Cepton Technologies, Inc., be deemed to have occurred on May 15th of such year); provided, however, that in the event that the date of the annual meeting is advanced more than thirty (30) days prior to such anniversary date or delayed more than seventy (70) days after such anniversary date then to be timely such notice must be so delivered, or mailed and received, not later than the ninetieth (90th) day prior to such annual meeting or, if later, the tenth (10th) day following the day on which public announcement of the date of such annual meeting was first made. In no event shall the adjournment or postponement of any meeting, or any announcement thereof, commence a new time period (or extend any time period) for the giving of a stockholder's notice as described above. The number of nominees a stockholder may nominate for election at the annual meeting (or in the case of a stockholder giving the notice on behalf of a beneficial owner, the number of nominees a stockholder may nominate for election at the annual meeting on behalf of such beneficial owner) shall not exceed the number of directors to be elected as such annual meeting.

- Special Meetings of Stockholders. Only such business shall be conducted at a special meeting of stockholders as shall have been brought before the meeting pursuant to the Corporation's notice of meeting delivered pursuant to Section 2.4 and Article VI hereof. Nominations of persons for election to the Board may be made at a special meeting of stockholders at which directors are to be elected pursuant to the Corporation's notice of meeting (1) by or at the direction of the Board or any duly authorized committee thereof or (2) provided that the Board has determined that directors shall be elected at such meeting, by any stockholder of the Corporation who (x) is a stockholder of record at the time of delivery by the stockholder of the notice provided for in this Section 2.14(b) to the Secretary of the Corporation and at the time of the special meeting, (v) who is entitled to vote at the meeting and upon such election, and (z) who complies with the notice procedures set forth in this Section 2.14(b). In the event the Corporation calls a special meeting of stockholders for the purpose of electing one or more directors to the Board, any such stockholder entitled to vote in such election of directors may nominate a person or persons for election to such position(s) as specified in the Corporation's notice of meeting, if the stockholder's notice, in proper written form as set forth in Section 2.14(c), shall be delivered to the Secretary at the principal executive office of the Corporation not earlier than the close of business on the one hundred twentieth (120th) day prior to such special meeting and not later than the close of business on the later of the ninetieth (90th) day prior to such special meeting or the tenth (10th) day following the day on which public announcement is first made of the date of the special meeting and of the nominees proposed by the Board to be elected at such meeting. In no event shall the public announcement of an adjournment or postponement of a special meeting commence a new time period (or extend any time period) for the giving of a stockholder's notice as described above. Notwithstanding the foregoing, if a stockholder is entitled to vote only for a specific class or category of directors at a special meeting of the stockholders, such stockholder's right to nominate one or more individuals for the election of a director at the meeting shall be limited to such class or category of directors.
- (c) Form of Notice. To be in proper written form, such stockholder's notice to the Secretary (whether pursuant to clauses (a)(2) or (b) of this Section 2.14) must set forth:
- (1) as to each person, if any, whom the stockholder proposes to nominate for election or re-election as a director (i) all information relating to such person that is required to be disclosed in solicitations of proxies for election of directors in an election contest, or is otherwise required, in each case pursuant to Regulation 14A under the Exchange Act, (ii) such person's written consent to being named in the proxy statement as a nominee and to serving as a director if elected and (iii) a reasonably detailed description of any compensatory, payment or other financial agreement, arrangement or understanding that such person has with any other person or entity other than the Corporation including the amount of any payment or payments received or receivable thereunder, in each case in connection with candidacy or service as a director of the Corporation;
- as to any other business (other than the nomination of persons for election as directors) that the stockholder desires to bring before the meeting, (i) a brief description of the business proposed to be brought before the meeting, (ii) the text of the proposal or business (including the text of any resolutions proposed for consideration and, in the event that such business includes a proposal to amend these Bylaws, the language of the proposed amendment), (iii) the reasons why the stockholder favors the proposal, (iv) the reasons for conducting such business at the meeting, and (v) any material interest in such business of such stockholder and the beneficial owner, if any, on whose behalf the proposal is made; and
- as to the stockholder giving the notice and the beneficial owner, if any, on whose behalf the nomination or proposal is made (i) the name and address of such stockholder, as they appear on the Corporation's books, and of such beneficial owner, (ii) the class or series and number of shares of the Corporation's capital stock that are, directly or indirectly, owned beneficially and of record by such stockholder and by such beneficial owner, (iii) a description of any agreement, arrangement or understanding with respect to the nomination or proposal between or among such stockholder and/or such beneficial owner, any of their respective affiliates or associates, and any others acting in concert with any of the foregoing, including, in the case of a nomination, the nominee, (iv) a description of any agreement, arrangement or understanding (including any derivative or short positions, profit interests, options, warrants, convertible securities, stock appreciation, forwards, futures, swaps, or similar rights, hedging transactions, and borrowed or loaned shares) that has been entered into as of the date of the stockholder's notice by, or on behalf of, such stockholder and such beneficial owner, whether or not such instrument or right shall be subject to settlement in underlying shares of capital stock of the Corporation, the effect or intent of which is to

mitigate loss to, manage risk or benefit of share price changes for, or increase or decrease the voting power of such stockholder or such beneficial owner with respect to shares of capital stock of the Corporation, (v) a representation that the stockholder is a holder of record of stock of the Corporation entitled to vote at such meeting and intends to appear in person or by proxy at the meeting to propose such business or nomination, (vi) a representation whether the stockholder or the beneficial owner, if any, intends or is part of a group which intends (A) to deliver a proxy statement and/or form of proxy to holders of at least the percentage of the Corporation's outstanding capital stock required to approve or adopt the proposal or elect the nominee and/or (B) otherwise to solicit proxies or votes from stockholders in support of such proposal or nomination, (vii) any other information relating to such stockholder and beneficial owner, if any, required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for, as applicable, the proposal and/or for the election of directors in an election contest pursuant to and in accordance with Section 14(a) of the Exchange Act and the rules and regulations promulgated thereunder and (viii) such other information relating to any proposed item of business as the Corporation may reasonably require to determine whether such proposed item of business is a proper matter for stockholder action.

The foregoing notice requirements of this Section 2.14(c) shall be deemed satisfied by a stockholder with respect to business other than a nomination if the stockholder has notified the Corporation of his, her or its intention to present a proposal at an annual meeting in compliance with applicable rules and regulations promulgated under the Exchange Act and such stockholder's proposal has been included in a proxy statement that has been prepared by the Corporation to solicit proxies for such annual meeting.

If requested by the Corporation, the information required under clauses (c)(3)(ii), (iii) and (iv) of this Section 2.14 shall be supplemented by such stockholder and any such beneficial owner not later than ten (10) days after the record date for the meeting to disclose such information as of the record date.

(d) General.

(1) The Corporation may require any proposed nominee for election or re-election as a director to furnish such other information, in addition to the information set forth in the stockholder's notice delivered pursuant to this Section 2.14, as it may reasonably require to determine the eligibility of such proposed nominee to serve as a director of the Corporation and whether such nominee qualifies as an "independent director" or "audit committee financial expert" under applicable law, securities exchange rules or regulations, or any publicly-disclosed corporate governance guideline or committee charter of the Corporation.

Except as otherwise expressly provide in any applicable rule or regulation promulgated under the Exchange Act, only such persons who are nominated in accordance with the procedures set forth in this Section 2.14 shall be eligible to be elected at an annual or special meeting of stockholders of the Corporation to serve as directors, and only such business as shall have been brought before the meeting in accordance with the procedures set forth in this Section 2.14 shall be conducted at a meeting of stockholders. Except as otherwise provided by law, the chairman of the meeting shall have the power and duty to (i) determine whether a nomination or any business proposed to be brought before the meeting was made or proposed, as the case may be, in accordance with the procedures set forth in this Section 2.14 (including whether the stockholder or beneficial owner, if any, on whose behalf the nomination or proposal is made or solicited (or is part of a group which solicited) or did not so solicit, as the case may be, proxies or votes in support of such stockholder's nominee or proposal in compliance with such stockholder's representation as required by Section 2.14(c)(3)(vi), and, (ii) if any proposed nomination or business was not made or proposed in compliance with this Section 2.14, to declare that such nomination shall be disregarded or that such proposed business shall not be transacted. Notwithstanding the foregoing provisions of this Section 2.14, unless otherwise required by law, if the stockholder who has delivered a notice pursuant to this Section 2.14 (or a qualified representative of such stockholder) does not appear at the annual or special meeting of stockholders of the Corporation to present a nomination or proposed business, such nomination shall be disregarded and such proposed business shall not be transacted, notwithstanding that proxies in respect of such vote may have been received by the Corporation. To be considered a "qualified representative" of the stockholder, a person must be a duly authorized officer, manager or partner of such stockholder or must be authorized by a writing executed by such stockholder or by telegram, cablegram or other means of electronic transmission that is deemed valid in accordance with Section 2.9 hereof delivered by

such stockholder to act for such stockholder as proxy at the meeting of stockholders, and such person must produce such writing or telegram, cablegram or electronic transmission, or a reliable reproduction of the writing or telegram, cablegram or electronic transmission, at the meeting of stockholders.

- (3) For purposes of this Section 2.14, "public announcement" shall mean disclosure in a press release reported by the Dow Jones News Service, Associated Press or comparable national news service, or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Sections 13, 14 or 15(d) of the Exchange Act and the rules and regulations promulgated thereunder.
- (4) Notwithstanding the foregoing provisions of this Section 2.14, stockholders shall also comply with all applicable requirements of the Exchange Act and the rules and regulations thereunder with respect to the matters set forth in this Section 2.14; <u>provided, however</u>, that any references in these Bylaws to the Exchange Act or the rules promulgated thereunder are not intended to and shall not limit the requirements applicable to nominations or proposals as to any other business to be considered pursuant to clause (a)(1)(iii) or (b) of this Section 2.14. Nothing in this Section 2.14 shall be deemed to affect any rights (x) of stockholders to request inclusion of proposals or nominations in the Corporation's proxy statement pursuant to Rule 14a-8 promulgated under the Exchange Act or (y) of the holders of any series of preferred stock to elect directors pursuant to any applicable provisions of the Certificate of Incorporation.
- Submission of Questionnaire, Representation and Agreement. To be eligible to be a nominee for election or re-election as a director of the Corporation nominated by a stockholder pursuant to Section 2.14(a)(1)(iii), the candidate for nomination must deliver (in accordance with the time periods prescribed for delivery of notice under clauses (a)(2) or (b) of this Section 2.14, as applicable) to the Secretary at the principal executive office of the Corporation (1) a completed written questionnaire (in a form provided by the Corporation) with respect to the background, qualifications, stock ownership and independence of such person and the background of any other person or entity on whose behalf the nomination is being made (which questionnaire shall be provided by the Secretary upon written request) and (2) a written representation and agreement (in the form provided by the Secretary upon written request) that such person (1) is not and, if elected as a director during his or her term of office, will not become a party to (x) any agreement, arrangement or understanding with, and has not given any commitment or assurance to, any person or entity as to how such person, if elected as a director of the Corporation, will act or vote on any issue or question in his or her capacity as a director (a "Voting Commitment") that has not been disclosed to the Corporation or (y) any Voting Commitment that could limit or interfere with such person's ability to comply, if elected as a director of the Corporation, with such person's fiduciary duties under applicable law, (2) is not and will not become a party to any agreement, arrangement or understanding with any person or entity other than the Corporation with respect to any direct or indirect compensation, reimbursement or indemnification in connection with service or action as a director of the Corporation that has not been disclosed therein and (3) in such person's individual capacity and on behalf of any person or entity on whose behalf the nomination is being made, would be in compliance, if elected as a director of the Corporation, and will comply with all applicable publicly disclosed corporate governance, conflict of interest, confidentiality and stock ownership and trading policies and guidelines of the Corporation (and, if requested by any candidate for nomination, the Secretary of the Corporation shall provide to such candidate for nomination all such policies and guidelines then in effect).

ARTICLE III

DIRECTORS

3.1 <u>Powers and Duties</u>. Subject to the provisions of the DGCL and to any limitations in the Certificate of Incorporation relating to action required to be approved by the stockholders, the business and affairs of the Corporation shall be managed, and all corporate powers shall be exercised, by or under the direction and control of the Board. The Board may delegate the management of the day-to-day operation of the business of the Corporation, provided that the business and affairs of the Corporation shall remain under the ultimate direction and control of the Board.

- 3.2 <u>Number and Qualifications</u>. The Board shall consist of one or more members, the exact number of which shall be fixed from time to time by resolution of the Board. Unless otherwise required by law or by the Certificate of Incorporation, directors need not be stockholders of the Corporation or residents of the State of Delaware. No reduction of the authorized number of directors shall have the effect of removing any director before that director's term of office expires.
- 3.3 <u>Classified Board of Directors.</u> The Board shall be divided into classes, with each such class serving for a term, as set forth in the Certificate of Incorporation.
- Resignations and Removals of Directors. Any director of the Corporation may resign from the Board or any committee thereof at any time, by giving notice in writing or by electronic transmission to the Chairman of the Board, the President or the Secretary of the Corporation and, in the case of a committee, to the chairman of such committee, if there be one and if there is no such chairman, to the Chairman of the Board. Such resignation shall take effect at the time therein specified (which may be upon the happening of an event specified therein) or, if no time is specified, immediately. Unless otherwise specified in such notice, the acceptance of such resignation shall not be necessary to make it effective. Except as otherwise required by law or the Certificate of Incorporation and except for any director elected by the holders of any series or class of preferred stock provided for or fixed pursuant to the provisions of Article V of the Certificate of Incorporation, any director or the entire Board may be removed from office at any time, but only for cause, and only by the affirmative vote of the holders of at least sixty-six and two-thirds percent (66 2/3%) of the total voting power of the outstanding shares of capital stock of the Corporation entitled to vote in the election of directors, voting together as a single class. Unless otherwise provided by the charter of the committee, any director serving on a committee of the Board may be removed from such committee at any time by the Board.
- 3.5 <u>Vacancies</u>. Except as otherwise required by law or the Certificate of Incorporation, any vacancy on the Board, by reason of death, resignation, retirement, disqualification or removal or otherwise, and any newly created directorship that results from an increase in the number of directors, shall be filled only by a majority of the Board then in office, even if less than a quorum, or by a sole remaining director. Any director of any class elected to fill a vacancy resulting from an increase in the number of directors of such class shall hold office for a term that shall coincide with the remaining term of that class. Any director elected to fill a vacancy not resulting from an increase in the number of directors shall have the same remaining term as that of his or her predecessor.
- 3.6 <u>Regular Meetings</u>. Regular meetings of the Board shall be held at such place or places, within or without the State of Delaware, on such date or dates and at such time or times, as shall have been established by the Board and publicized among all directors. A notice of each regular meeting shall not be required.
- 3.7 <u>Special Meetings.</u> Special meetings of the Board for any purpose or purposes may be called at any time by the Chairman of the Board, the Chief Executive Officer, if any, the President or any two directors then in office. Notice of each such meeting shall be given to each director, if by mail, addressed to such director at his or her residence or usual place of business, at least five (5) days before the day on which such meeting is to be held, or shall be sent to such director at such place by facsimile, electronic mail or other electronic transmissions, or be delivered personally or by telephone, in each case at least twenty-four (24) hours prior to the time set for such meeting. A notice of special meeting need not state the purpose of such meeting, and, unless indicated in the notice thereof, any and all business may be transacted at a special meeting.
- 3.8 <u>Organization</u>. Meetings of the Board shall be presided over by the Chairman of the Board, or in his or her absence by the Vice Chairman of the Board, if any, or in his or her absence by the Chief Executive Officer, if any, if such person is a member of the Board, or in the absence of any such person, by a chairperson chosen at the meeting. The Secretary shall act as secretary of the meeting, but in his or her absence the chairperson of the meeting may appoint any person to act as secretary of the meeting.

- 3.9 <u>Meetings by Means of Conference Telephone</u>. Members of the Board of the Corporation, or any committee thereof, may participate in a meeting of the Board or such committee by means of a conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other, and participation in a meeting through the use of such equipment shall constitute presence in person at such meeting.
- 3.10 Quorum. Except as otherwise required by law, these Bylaws or the Certificate of Incorporation, at all meetings of the Board or any committee thereof, a majority of the entire Board or a majority of the directors constituting such committee, as the case may be, shall constitute a quorum for the transaction of business and the act of a majority of the directors or committee members present at any meeting at which there is a quorum shall be the act of the Board or such committee, as applicable. If a quorum shall not be present at any meeting of the Board or any committee thereof, a majority of the directors present thereat may adjourn the meeting from time to time, without notice other than announcement at the meeting of the time and place of the adjourned meeting, until a quorum shall be present.
- 3.11 Action of the Board by Written Consent. Unless otherwise provided in the Certificate of Incorporation or these Bylaws, any action required or permitted to be taken at any meeting of the Board or any committee thereof may be taken without a meeting if all of the members of the Board or such committee, as the case may be, consent thereto in writing or by electronic transmission and the writing or writings or electronic transmission or transmissions are filed with the minutes of proceedings of the Board or such committee. Such filing shall be in paper form if the minutes are maintained in paper form and shall be in electronic form if the minutes are maintained in electronic form. After an action is taken, the consent or consents relating thereto shall be filed with the minutes of proceedings of the Board or such committee.
- 3.12 <u>Expense Reimbursement and Compensation</u>. Directors and members of committees may receive such compensation, if any, for their services and such reimbursement of expenses as may be fixed or determined by resolution of the Board. This Section 3.12 shall not be construed to preclude any director from serving the Corporation in any other capacity as an officer, agent, employee or otherwise and receiving compensation for those services.
- 3.13 <u>Chairman and Vice Chairman of the Board.</u> The Corporation shall have a Chairman of the Board and, at the Board's discretion, a Vice Chairman of the Board. Any such Chairman of the Board or Vice Chairman of the Board may be an officer of this Corporation as determined by the Board pursuant to Section 4.1. The Chairman of the Board shall preside at all meetings of the stockholders and of the Board and shall exercise and perform such other powers and duties as may be from time to time assigned to him or her by the Board or as may be prescribed by these Bylaws.

3.14 <u>Committees</u>.

(a) The Board may, by resolution, designate from among its members one (1) or more committees, each such committee to consist of one (1) or more of the directors of the Corporation, the exact number of which shall be fixed from time to time by resolution of the Board. The Board may designate one or more directors as alternate members of any committee to replace any absent or disqualified member at any meeting of the committee. In the absence or disqualification of a member of a committee, the member or members present at any meeting and not disqualified from voting, whether or not he, she or they constitute a quorum, may unanimously appoint another member of the Board to act at the meeting in the place of any such absent or disqualified member. Any such committee, to the extent provided in the resolution of the Board establishing such committee, shall have and may exercise all the powers and authority of the Board in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers which may require it; provided, however, no such committee shall have the power or authority in reference to the following matters: (i) approving or adopting, or recommending to the stockholders, any action or matter (other than the election or removal of directors) expressly required by the DGCL to be submitted to stockholders for approval, or (ii) adopting, amending or repealing any bylaw of the Corporation. All committees of the Board shall keep minutes of their meetings and shall report their proceedings to the Board when requested or required by the Board.

- (b) Any committee of the Board may adopt such rules and regulations not inconsistent with the provisions of law, the Certificate of Incorporation or these Bylaws for the conduct of its meetings as such committee may deem proper.
- 3.15 <u>Telephonic Meetings</u>. Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, members of the Board, or any committee designated by the Board, may participate in a meeting of the Board, or such committee, as the case may be, by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other, and such participation in a meeting shall constitute presence in person at the meeting.

ARTICLE IV

OFFICERS

- 4.1 <u>General</u>. The officers of the Corporation shall be chosen by the Board and shall include a President, a Chief Executive Officer, and a Secretary. The Board, in its discretion, may also appoint such additional officers as the Board may deem necessary or desirable, including a Chief Financial Officer, one or more Vice Presidents, one or more Assistant Secretaries, a Treasurer and one or more Assistant Treasurers, each of whom shall hold office for such period, have such authority and perform such duties as the Board may from time to time determine. Subject to the rules or regulations of any stock exchange applicable to the Corporation or other applicable law, the Board may delegate to any officer of this Corporation or any committee of the Board the power to appoint, remove and prescribe the term and duties of any officer provided for in this Section 4.1. Any number of offices may be held by the same person, unless otherwise provided by the Certificate of Incorporation or these Bylaws.
- 4.2 <u>Appointment and Term</u>. Each officer shall serve at the pleasure of the Board and shall hold office until such officer's successor has been appointed, or until such officer's earlier death, resignation or removal. Any officer may be removed, either with or without cause, by the Board or by any officer upon whom such power of removal may be conferred by the Board.
- 4.3 <u>Resignations</u>. An officer may resign from his or her position at any time, by giving notice in writing or electronic transmission to the Corporation. Such resignation shall be without prejudice to any rights, if any, the Corporation may have under any contract to which the officer is a party. Such resignation shall take effect at the time therein specified (which may be upon the happening of an event specified therein), or, if no time is specified, immediately; and unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.
- 4.4 <u>Vacancies</u>. A vacancy in any office because of death, resignation, removal, disqualification or otherwise shall be filled by the Board in the manner prescribed in these Bylaws for election or appointment to such office.
- 4.5 <u>Compensation</u>. The Board shall fix, or may appoint a committee to fix, the compensation of all officers of the Corporation appointed by the Board. Subject to the rules or regulations of any stock exchange applicable to the Corporation or other applicable law, the Board may authorize any officer upon whom the power to appoint officers may have been conferred pursuant to Section 4.1 to fix the compensation of such officers.
- 4.6 <u>Authority and Duties of Officers</u>. All officers of the Corporation shall respectively have such authority and perform such duties in the management of the business of the Corporation as may be provided herein or designated from time to time by the Board and, to the extent not so provided, as generally pertain to their respective offices, subject to the control of the Board.

ARTICLE V

STOCK

- 5.1 Certificates. The shares of the Corporation shall be represented by certificates, provided that the Board may provide by resolution or resolutions that some or all of any or all classes or series of the Corporation's stock shall be uncertificated shares. Any such resolution shall not apply to shares represented by a certificate until such certificate is surrendered to the Corporation. Every holder of stock in the Corporation represented by certificates shall be entitled to have a certificate signed by, or in the name of, the Corporation by any two authorized officers, representing the number of shares registered in certificate form. Any or all of the signatures on the certificate may be a facsimile signature. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if such person were such officer, transfer agent or registrar at the date of issuance.
- 5.2 Transfers. Shares of stock of the Corporation shall be transferable upon the Corporation's books by the holders thereof, in person or by their duly authorized attorneys or legal representatives, upon surrender to the Corporation of the certificate or certificates representing such shares endorsed by the appropriate person or persons (or, with respect to uncertificated shares, by delivery of duly executed instructions or in any other manner permitted by applicable law). Certificates representing such shares, if any, shall be cancelled and new certificates, if the shares are to be certificated, shall thereupon be issued. Shares of capital stock of the Corporation that are not represented by a certificate shall be transferred in accordance with applicable law. A record shall be made of each transfer. Whenever any transfer of shares shall be made for collateral security, and not absolutely, it shall be so expressed in the entry of the transfer if, when the certificates are presented, both the transferor and transferee request the Corporation to do so. The Board shall have power and authority to make such rules and regulations as it may deem necessary or proper concerning the issue, transfer and registration of certificates for shares of stock of the Corporation.
- 5.3 <u>Lost, Stolen, or Destroyed Certificates</u>. The Board may direct a new certificate or uncertificated shares be issued in place of any certificate theretofore issued by the Corporation alleged to have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming the certificate of stock to be lost, stolen or destroyed. When authorizing such issuance of a new certificate or uncertificated shares, the Board may, in its discretion, require the owner of such lost, stolen or destroyed certificate to give the Corporation a bond (or other adequate security) in such sum as it may direct as indemnity against any claim that may be made against the Corporation on account of the alleged loss, theft or destruction of such certificate or the issuance of such new certificate or uncertificated shares. The Board may adopt such other provisions and restrictions with reference to lost certificates, not inconsistent with applicable law, as it shall in its discretion deem appropriate.
- 5.4 <u>Record Owners.</u> The Corporation shall be entitled to recognize the exclusive right of a person registered on its books as the owner of shares to receive dividends, and to vote as such owner, and shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person, whether or not it shall have express or other notice thereof, except as otherwise required by law.

ARTICLE VI

NOTICES

6.1 Notices.

(a) Whenever notice is required by law, the Certificate of Incorporation or these Bylaws, to be given to any director, member of a committee or stockholder, such notice may be given by mail, addressed to such director, member of a committee or stockholder, at such person's address as it appears on the books of the Corporation or given by the stockholder for such purpose, with postage thereon prepaid, and such notice shall be deemed to be given at the time when the same shall be deposited in the United States mail. Notice may also be given personally or by facsimile, electronic mail or other means of electronic transmission in accordance with applicable law. Without limiting the foregoing, any notice to stockholders given by the Corporation pursuant to the DGCL, the Certificate of Incorporation or these Bylaws shall be effective if given by a form of electronic transmission consented to by the stockholder to whom the notice is given.

- (b) Notice to a stockholder given by a form of electronic transmission in accordance with these Bylaws shall be deemed given: (i) if by facsimile telecommunication, when directed to a number at which the stockholder has consented to receive notice; (ii) if by electronic mail, when directed to an electronic mail address at which the stockholder has consented to receive notice; (iii) if by a posting on an electronic network, together with separate notice to the stockholder of such specific posting, upon the later of such posting and the giving of such separate notice; and (iv) if by another form of electronic transmission, when directed to the stockholder. For purposes of these Bylaws, "electronic transmission" means any form of communication, not directly involving the physical transmission of paper, that creates a record that may be retained, retrieved and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process.
- (c) Any notice to stockholders given by the Corporation may be given by a single written notice to stockholders who share an address if consented to by the stockholders at such address to whom such notice is given. Any such consent shall be revocable by the stockholders by written notice to the Corporation. Any stockholder who fails to object in writing to the Corporation, within sixty (60) days of having been given written notice by the Corporation of its intention to send the single notice as set forth in this Section 6.1(c) shall be deemed to have consented to receiving such single written notice.
- Maivers of Notice. Whenever any notice is required by applicable law, the Certificate of Incorporation or these Bylaws, to be given to any director, member of a committee or stockholder, a waiver thereof in writing, signed by the person or persons entitled to notice, or a waiver thereof given by electronic transmission by the person or persons entitled to notice, in each case, whether before or after the time stated therein, shall be deemed equivalent thereto. Attendance of a person at a meeting, present in person or represented by proxy, shall constitute a waiver of notice of such meeting, except where the person attends the meeting for the express purpose of objecting at the beginning of the meeting to the transaction of any business because the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any annual or special meeting of stockholders or any regular or special meeting of the directors or members of a committee of directors need be specified in any written waiver of notice unless so required by law, the Certificate of Incorporation or these Bylaws.

ARTICLE VII

INDEMNIFICATION AND ADVANCEMENT OF EXPENSES

- 7.1 Definitions. For purposes of this Article VII, the following terms shall have the meanings set forth below:
- (a) "Action" means any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative.
- (b) "Indemnified Party" means any person who is or was a party or is threatened to be made a party to any Action by reason of the fact that such person is or was a director or officer of the Corporation (which shall include actions taken in connection with or relating to the incorporation of the Corporation) or, while a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, limited liability company, joint venture, trust or other enterprise, including any employee benefit plan of the Corporation.
- 7.2 <u>Indemnification</u>. The Corporation shall indemnify and hold harmless, to the fullest extent permitted by applicable law as it presently exists or may hereafter be amended, any Indemnified Party against any and all liability and loss suffered and expenses (including attorneys' fees) reasonably incurred by such Indemnified Party. Notwithstanding the preceding sentence, except as provided in Section 7.5 of this Article VII, the Corporation shall be required to indemnify an Indemnified Party in connection with an Action (or part thereof) commenced by such Indemnified Party only if the commencement of such Action (or part thereof) by the Indemnified Party was authorized in the specific case by the Board of the Corporation.
- 7.3 <u>Determination.</u> Any indemnification under this Article VII (unless ordered by a court) shall be made by the Corporation only as authorized in the specific case upon a determination that the indemnification of Indemnified Party is proper in the circumstances because such Indemnified Party has met the applicable standard

of conduct required by applicable law, as the case may be. Such determination shall be made, with respect to an Indemnified Party who is a director or officer at the time of such determination, (a) by a majority vote of the directors who are not parties to such Action, even though less than a quorum, or (b) by a committee of such directors designated by a majority vote of such directors, even though less than a quorum, or (c) if there are no such directors, or if such directors so direct, by independent legal counsel in a written opinion or (d) by the stockholders. Such determination shall be made, with respect to former directors or officers, by any person or persons having the authority to act on the matter on behalf of the Corporation. To the extent, however, that a present or former Indemnified Party of the Corporation has been successful on the merits or otherwise in defense of any Action or in defense of any claim, issue or matter therein, such Indemnified Party shall be indemnified against expenses (including attorneys' fees) actually and reasonably incurred by such Indemnified Party in connection therewith, without the necessity of authorization in the specific case.

- 7.4 <u>Expenses Payable in Advance</u>. Expenses, including without limitation attorneys' fees, incurred by an Indemnified Party in defending any Action shall be paid by the Corporation in advance of the final disposition of such Action upon receipt of an undertaking by or on behalf of such Indemnified Person (if required by law) to repay such amount if it shall ultimately be determined that such Indemnified Party is not entitled to be indemnified by the Corporation as authorized in this Article VII.
- 7.5 <u>Claim.</u> If a claim for indemnification under this Article VII (following the final disposition of such proceeding) is not paid in full within sixty (60) days after the Corporation has received a claim therefor by the Indemnified Party, or if a claim for any advancement of expenses under this Article VII is not paid in full within thirty (30) days after the Corporation has received a statement or statements requesting such amounts to be advanced, the Indemnified Party shall thereupon (but not before) be entitled to file suit to recover the unpaid amount of such claim. If successful in whole or in part, the Indemnified Party shall be entitled to be paid the expense of prosecuting such claim to the fullest extent permitted by law. In any such action, the Corporation shall have the burden of proving that the Indemnified Party is not entitled to the requested indemnification or advancement of expenses under applicable law.
- 7.6 Other Indemnification or Advancement. The rights to indemnification and advancement of expenses provided by this Article VII shall not be construed to be exclusive of or limit any other rights to which any Indemnified Party or other person may be entitled under the Certificate of Incorporation or any bylaw, agreement, vote of the stockholders or disinterested directors or otherwise, both as to action in such Indemnified Party's official capacity and as to action in another capacity while holding office.
- 7.7 <u>Insurance</u>. The Corporation may purchase and maintain insurance in the amounts the Board deems appropriate or advisable on behalf of any Indemnified Party against any liability asserted against such Indemnified Party and incurred by such Indemnified Party in such Indemnified Party's capacity, or arising out of such Indemnified Party's status, as an Indemnified Party, whether or not the Corporation would have the power to indemnify such Indemnified Party against such liability under applicable provisions of law.

ARTICLE VIII

GENERAL PROVISIONS

- 8.1 Fiscal Year. The fiscal year of the Corporation shall be fixed by resolution of the Board.
- 8.2 <u>Corporate Seal</u>. The Corporation may adopt and may subsequently alter the corporate seal and it may use the same by causing it or a facsimile thereof to be impressed or affixed or reproduced or otherwise.
- 8.3 <u>Maintenance and Inspection of Records</u>. The Corporation shall, either at its principal executive office or at such place or places as designated by the Board, keep a record of its stockholders listing their

names and addresses and the number and class of shares held by each stockholder, a copy of these Bylaws as amended to date, accounting books and other records.

- 8.4 <u>Reliance Upon Books, Reports and Records.</u> Each director and each member of any committee designated by the Board shall, in the performance of his or her duties, be fully protected in relying in good faith upon the books of account or other records of the Corporation and upon such information, opinions, reports or statements presented to the Corporation by any of its officers or employees, or committees of the Board so designated, or by any other person as to matters which such director or committee member reasonably believes are within such other person's professional or expert competence and who has been selected with reasonable care by or on behalf of the Corporation
- 8.5 <u>Dividends</u>. Subject to the requirements of the DGCL and the provisions of the Certificate of Incorporation, dividends on the capital stock of the Corporation may be declared by the Board at any regular or special meeting of the Board (or any action by written consent in lieu thereof in accordance with Section 3.11 hereof), and may be paid in cash, in property, or in shares of the Corporation's capital stock. Before payment of any dividend, there may be set aside out of any funds of the Corporation available for dividends such sum or sums as the Board from time to time, in its absolute discretion, deems proper as a reserve or reserves to meet contingencies, or for purchasing any of the shares of capital stock, warrants, rights, options, bonds, debentures, notes, scrip or other securities or evidences of indebtedness of the Corporation, or for equalizing dividends, or for repairing or maintaining any property of the Corporation, or for any proper purpose, and the Board may modify or abolish any such reserve. In the event that the Board declares a dividend on the capital stock of the Corporation pursuant to this Section 8.5, the Board may fix a record date in order that the Corporation may determine the stockholders entitled to receive payment of any dividend, which record date shall be fixed in accordance with Section 2.11(b).
- 8.6 <u>Emergency Bylaws</u>. In the event of any emergency, disaster or catastrophe, as referred to in Section 110 of the DGCL or any successor provision, or other similar emergency condition, as a result of which a quorum of the Board or a standing committee of the Board cannot readily be convened for action, then the director or directors in attendance at the meeting shall constitute a quorum. Such director or directors in attendance may further take action to appoint one or more of themselves or other directors to membership on any standing or temporary committees of the Board as they shall deem necessary and appropriate.
- 8.7 <u>Certificate of Incorporation Governs</u>. In the event of any conflict between the provisions of the Certificate of Incorporation and these Bylaws, the provisions of the Certificate of Incorporation shall govern.
- 8.8 <u>Severability.</u> Any determination that any provision of these Bylaws is for any reason inapplicable, illegal or ineffective shall not affect or invalidate any other provision of these Bylaws.
- 8.9 Actions with Respect to Securities of Other Entities. All stock and other securities of other entities owned or held by the Corporation for itself, or for other parties in any capacity, shall be voted (including by written consent), and all proxies with respect thereto shall be executed, by the person or persons authorized to do so by resolution of the Board or, in the absence of such authorization, by the President, Chief Executive Officer, Secretary or such other officer of the Company designated by the Board.

ARTICLE IX

AMENDMENTS

9.1 <u>Amendments.</u> These Bylaws may be altered, amended or repealed, in whole or in part, or new Bylaws may be adopted by the Board or by the stockholders as expressly provided in the Certificate of Incorporation.

* * *

	EXHIDIT 4.1				
Number C- []					
	SEE REVERSE FOR IMPORTANT NOTICE REGARDING OWNERSHIP AND				
	TRANSFER RESTRICTIONS AND CERTAIN OTHER INFORMATION				
CPTN	CUSIP 15673X 101 SEE REVERSE FOR CERTAIN DEFINITIONS				
CEPTON, INC.					
	A Delaware Corporation				
	INCORPORATED UNDER THE LAWS OF THE STATE OF DELAWARE. COMMON STOCK				

SPECIMEN

SIEGIVIEN					
FULLY PAID AND NON-ASSESSABLE SHARES OF COMMON STOC CEPTON, INC.	CK. PAR VALUE OF \$0.00001 PER SHARE OF				
transferable on the books of the company in person or by duly authorized attorney upon surrender of this certificate properly endorsed.					
This certificate is not valid unless countersigned by the Transfer Agent and registered by the Registrar. Witness the seal of the Company and the facsimile signatures of its duly authorized officers.					
President and Chief Executive Officer	Chief Financial Officer				

CEPTON, INC.

The Company will furnish without charge to each stockholder who so requests the powers, designations, preferences and relative, participating, optional or other special rights of each class of stock or series thereof of the Company and the qualifications, limitations, or restrictions of such preferences and/or rights. This certificate and the shares of common stock represented hereby are issued and shall be held subject to all the provisions of the Company's certificate of incorporation and all amendments thereto and resolutions of the Board of Directors providing for the issue of securities (copies of which may be obtained from the secretary of the Company), to all of which the holder of this certificate by acceptance hereof assents.

The following abbreviations, when used in the inscription on the face of this certificate, shall be construed as though they were written out in full according to applicable laws or regulations:

TEN COM	— as tenants in common		UNIF GIFT MIN ACT -	—Custo	odian
TEN ENT	— as tenants by the entireties			(Cust)	(Minor)
IT TEN	— as joint tenants with right of survivorship and no	ot as tenants in common	Under Uniform Gifts to	Minors Act	
TTEE	— trustee under Agreement dated			(S	tate)
	A 11th 1 11 to 1	1 1 14 1			
	Additional abbreviations	s may also be used though no	ot in the above list.		
	For value received,	hereby sell, assig	gn and transfer unto		
PLEASE INS	SERT SOCIAL SECURITY OR OTHER IDENTIFYI	NG NUMBER OF ASSIGN	EE		
DI EACE DD	INT OR TYPEWRITE NAME AND ADDRESS INC	TITIDING DOCTAL ZID COL	DE OE ACCIONEE		
PLEASE PK	INT OR TYPEWRITE NAME AND ADDRESS INC	LUDING PUSTAL ZIP CUI	JE OF ASSIGNEE		
Shares of the	e common stock represented by this certificate and do	hereby irrevocably constitut	e and appoint	attorne	y, to transfer the
	the books of the within-named corporation with full I			, attorne	y, to transfer the
DATED _					
		NOTICE: The sid	gnature to this assignment mu	ist correspond i	with the name as
			face of the certificate in eve		
		_	any change whatsoever.	ily purificatai w	inout uncrunon
		· ·	, c		
SIGNATURI	E GUARANTEED:				
THE CICNA	ATURES(S) SHOULD BE GUARANTEED BY AN	ELICIDI E CUADANTOD	INICTITUTION (DANIZO	CTOCK DDOK	EDC CAVINCS
	N ASSOCIATIONS AND CREDIT UNIONS WITH		-		
_), PURSUANT TO S.E.C. RULE 17Ad-15 (OR ANY		III I ROVED DIGINII ORE	COMMITTEE	THE DITTELLION
,	,,	- /			

INDEMNIFICATION AGREEMENT

This Indemnification Agreement ("<u>Agreement</u>") is made as of [_____], 2022 by and between Cepton, Inc., a Delaware corporation (the "<u>Company</u>"), and [_____] ("<u>Indemnitee</u>"). This Agreement supersedes and replaces any and all previous Agreements between the Company and Indemnitee covering the subject matter of this Agreement.

RECITALS

WHEREAS, highly competent persons have become more reluctant to serve publicly-held corporations as directors, officers or in other capacities unless they are provided with adequate protection through insurance or adequate indemnification against inordinate risks of claims and actions against them arising out of their service to and activities on behalf of the corporation;

WHEREAS, Indemnitee may be entitled to indemnification pursuant to the General Corporation Law of the State of Delaware, as the same exists or may hereafter be amended (the "<u>DGCL</u>") and the DGCL expressly provides that the indemnification provisions set forth therein are not exclusive, and thereby contemplate that contracts may be entered into between the Company and members of the board of directors (the "<u>Board</u>"), officers and other persons with respect to indemnification;

WHEREAS, the Amended and Restated Bylaws of the Company (the "<u>Bylaws</u>") and the Second Amended and Restated Certificate of Incorporation of the Company (the "<u>Certificate of Incorporation</u>" and, together with the Bylaws, the "<u>Organizational Documents</u>") require indemnification of the officers and directors of the Company to the fullest extent authorized by the DGCL;

WHEREAS, in order to attract and retain qualified individuals, the Company maintains on an ongoing basis, at its sole expense, liability insurance to protect persons serving the Company and its subsidiaries from certain liabilities;

WHEREAS, the furnishing of such insurance has been a customary and widespread practice among United States-based corporations and other business enterprises, the Company believes that, given current market conditions and trends, such insurance may not be available to it on terms that the Company considers to be commercially reasonable or, if available to it on commercially reasonable terms during some period of time, may be available to it in the future only at higher premiums and with more exclusions, while at the same time, directors, officers, and other persons in service to corporations or business enterprises are being increasingly subjected to expensive and time-consuming litigation relating to, among other things, matters that traditionally would have been brought only against the Company or business enterprise itself;

WHEREAS, the uncertainties relating to such insurance and to indemnification may increase the difficulty of attracting and retaining such person;

WHEREAS, the Board has determined that the increased difficulty in attracting and retaining such persons is detrimental to the best interests of the Company and its stockholders and that the Company should act to assure such persons that there will be increased certainty of such protection in the future;

WHEREAS, the Organizational Documents and the DGCL expressly provide that the indemnification provisions set forth therein are not exclusive, and thereby contemplate that contracts may be entered into between the Company and members of the board of directors, officers and other persons with respect to indemnification;

WHEREAS, it is reasonable, prudent and necessary for the Company contractually to obligate itself to indemnify, and to advance expenses on behalf of, such persons to the fullest extent permitted by applicable law so that they will serve or continue to serve the Company free from undue concern that they will not be so indemnified; and

WHEREAS, this Agreement is a supplement to and in furtherance of the Organizational Documents and any resolutions adopted pursuant thereto, as well as any rights of Indemnitee under any directors' and officers' liability insurance policy, and this Agreement shall not be deemed a substitute therefor, nor to diminish or abrogate any rights of Indemnitee thereunder.

NOW, THEREFORE, in consideration of the premises and the covenants contained herein, the Company and Indemnitee do hereby covenant and agree as follows:

Section 1. Services to the Company. Indemnitee agrees to serve as an [officer/director] of the Company, at the request of the Company. Indemnitee may at any time and for any reason resign from such position (subject to any other contractual obligation or any obligation imposed by operation of law), in which event the Company shall have no obligation under this Agreement to continue Indemnitee in such position. This Agreement shall not be deemed an employment contract between the Company (or any of its subsidiaries or any Enterprise) and Indemnitee. Indemnitee specifically acknowledges that Indemnitee's employment with the Company (or any of its subsidiaries or any Enterprise), if any, is at will, and the Indemnitee may be discharged at any time for any reason, with or without cause, except as may be otherwise provided in any existing or future written employment contract between Indemnitee and the Company (or any of its subsidiaries or any Enterprise), other applicable formal severance policies duly adopted by the Board, or, with respect to service as an [officer/director] of the Company, by the Organizational Documents and the DGCL. The foregoing notwithstanding, this Agreement shall continue in force after Indemnitee has ceased to serve as an [officer/director] of the Company, as provided in Section 16 hereof.

Section 2. <u>Definitions.</u> As used in this Agreement:

- (a) References to "agent" shall mean any person who is or was a director, officer, or employee of the Company or a subsidiary of the Company or other person authorized by the Company to act for the Company, to include such person serving in such capacity as a director, officer, employee, fiduciary or other official of another corporation, partnership, limited liability company, joint venture, trust or other enterprise at the request of, for the convenience of, or to represent the interests of the Company or a subsidiary of the Company.
 - (b) A "Change in Control" shall be deemed to occur upon the earliest to occur after the date of this Agreement of any of the following events:
- i. Acquisition of Stock by Third Party. Any Person (as defined below) is or becomes the Beneficial Owner (as defined below), directly or indirectly, of securities of the Company representing thirty percent (30%) or more of the combined voting power of the Company's then outstanding securities unless the change in relative Beneficial Ownership of the Company's securities by any Person results solely from a reduction in the aggregate number of outstanding shares of securities entitled to vote generally in the election of directors; provided, however, that the foregoing shall not include any Person having such status immediately after the closing of the business combination by among Cepton Technologies, Inc. and Growth Capital Acquisition Corp. (the "Business Combination") unless after the Business Combination such Person is or becomes the Beneficial Owner, directly or indirectly, of additional securities of the Company representing in the aggregate an additional five percent (5%) or more of the combined voting power of the Company's then outstanding securities;
- ii. Change in Board of Directors. During any period of two (2) consecutive years (not including any period prior to the closing of the Business Combination), individuals who at the beginning of such period constitute the Board, and any new director (other than a director designated by a person who has entered into an agreement with the Company to effect a transaction described in Sections 2(b)(i), 2(b)(iii) or 2(b)(iv)) whose election by the Board or nomination for election by the Company's stockholders was approved by a vote of at least two-thirds (2/3) of the directors then still in office who either were directors at the beginning of the period or whose election or nomination for election was previously so approved, cease for any reason to constitute at least a majority of the members of the Board;
- iii. *Corporate Transactions*. The effective date of a merger or consolidation of the Company with any other entity, other than a merger or consolidation which would result in the voting securities of the Company outstanding immediately prior to such merger or consolidation continuing to represent (either by remaining outstanding or by being converted into voting securities of the Surviving Entity (as defined below)) more than fifty percent (50%) of the combined voting power of the voting securities of the Surviving Entity outstanding immediately after such merger or consolidation and with the power to elect at least a majority of the board of directors or other governing body of such Surviving Entity;

iv. *Liquidation*. The approval by the stockholders of the Company of a complete liquidation of the Company or an agreement for the sale or disposition by the Company of all or substantially all of the Company's assets; and

v. *Other Events*. There occurs any other event of a nature that would be required to be reported in response to Item 6(e) of Schedule 14A of Regulation 14A (or a response to any similar item on any similar schedule or form) promulgated under the Exchange Act (as defined below), whether or not the Company is then subject to such reporting requirement.

For purposes of this Section 2(b), the following terms shall have the following meanings:

- (A) "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended from time to time.
- (B) "Person" shall have the meaning as set forth in Sections 13(d) and 14(d) of the Exchange Act; <u>provided</u>, <u>however</u>, that Person shall exclude (i) the Company, (ii) any trustee or other fiduciary holding securities under an employee benefit plan of the Company, and (iii) any entity owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company.
- (C) "Beneficial Owner" shall have the meaning given to such term in Rule 13d-3 under the Exchange Act; <u>provided</u>, <u>however</u>, that Beneficial Owner shall exclude any Person otherwise becoming a Beneficial Owner by reason of the stockholders of the Company approving a merger of the Company with another entity.
 - (D) "Surviving Entity" shall mean the surviving entity in a merger or consolidation or any entity that controls directly or indirectly, such surviving entity.
- (c) "Corporate Status" describes the status of a person who is or was a director, trustee, partner, managing member, officer, employee or, agent or fiduciary of the Company or of any other corporation, limited liability company, partnership or joint venture, trust or other enterprise which such person is or was serving at the request of the Company.
- (d) "Disinterested Director" shall mean a director of the Company who is not and was not a party to the Proceeding in respect of which indemnification is sought by Indemnitee.
- (e) "Enterprise" shall mean the Company and any other corporation, limited liability company, partnership, joint venture, trust or other enterprise of which Indemnitee is or was serving at the request of the Company as a director, officer, trustee, partner, managing member, employee, agent or fiduciary.
- (f) "Expenses" shall include all reasonable attorneys' fees, retainers, court costs, transcript costs, fees and other costs of experts and other professionals, witness fees, travel expenses, duplicating costs, printing and binding costs, telephone charges, postage, delivery service fees, any federal, state, local or foreign taxes imposed on Indemnitee as a result of the actual or deemed receipt of any payments under this Agreement, ERISA excise taxes and penalties, and all other disbursements, obligations or expenses of the types customarily incurred in connection with, or as a result of, prosecuting, defending, preparing to prosecute or defend, investigating, being or preparing to be a deponent or witness in, or otherwise participating in, a Proceeding. Expenses also shall include (i) Expenses incurred in connection with any appeal resulting from any Proceeding, including without limitation the premium, security for, and other costs relating to any cost bond, supersedeas bond, or other appeal bond or its equivalent, and (ii) expenses incurred in connection with recovery under any directors' and officers' liability insurance policies maintained by the Company, regardless of whether Indemnitee is ultimately determined to be entitled to such indemnification, advancement or Expenses or insurance recovery, as the case may be, and (iii) for purposes of Section 14(d) only, Expenses incurred by Indemnitee in connection with the interpretation, enforcement or defense of Indemnitee's rights under this Agreement, the Organizational Documents or under any directors' and officers' liability insurance policies maintained by the Company, by litigation or otherwise. The parties agree that for the purposes of any advancement of Expenses for which Indemnitee has made written demand to the Company in accordance with this Agreement, all Expenses included in such demand that are certified by affidavit of Indemnitee's counsel as being reasonable in the good faith judgment of such counsel shall be presumed conclusively to be reasona

- (g) "Independent Counsel" shall mean a law firm, or a member of a law firm, that is experienced in matters of corporation law and neither presently is, nor in the past five years has been, retained to represent: (i) the Company or Indemnitee in any matter material to either such party (other than with respect to matters concerning the Indemnitee under this Agreement, or of other indemnitees under similar indemnification agreements), or (ii) any other party to the Proceeding giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term "Independent Counsel" shall not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Company or Indemnitee in an action to determine Indemnitee's rights under this Agreement. The Company agrees to pay the reasonable fees and expenses of the Independent Counsel referred to above and to fully indemnify such counsel against any and all Expenses, claims, liabilities and damages arising out of or relating to this Agreement or its engagement pursuant hereto.
- (h) The term "Proceeding" shall include any threatened, pending or completed action, suit, claim, counterclaim, cross claim, arbitration, mediation, alternate dispute resolution mechanism, investigation, inquiry, administrative hearing or any other actual, threatened or completed proceeding, whether brought in the right of the Company or otherwise and whether of a civil, criminal, administrative, legislative, regulatory or investigative (formal or informal) nature, including any appeal therefrom, in which Indemnitee was, is or will be involved as a party, potential party, non-party witness or otherwise by reason of Indemnitee's Corporate Status or by reason of any action taken by Indemnitee (or a failure to take action by Indemnitee) or of any action (or failure to act) on Indemnitee's part while acting pursuant to Indemnitee's Corporate Status, in each case whether or not serving in such capacity at the time any liability or Expense is incurred for which indemnification, reimbursement, or advancement of Expenses can be provided under this Agreement. If the Indemnitee believes in good faith that a given situation may lead to or culminate in the institution of a Proceeding, this shall be considered a Proceeding under this paragraph.
- (i) Reference to "other enterprise" shall include employee benefit plans; references to "fines" shall include any excise tax assessed with respect to any employee benefit plan; references to "serving at the request of the Company" shall include any service as a director, officer, employee or agent of the Company which imposes duties on, or involves services by, such director, officer, employee or agent with respect to an employee benefit plan, its participants or beneficiaries; and a person who acted in good faith and in a manner Indemnitee reasonably believed to be in the best interests of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interests of the Company" as referred to in this Agreement.

Section 3. Indemnity in Third-Party Proceedings. The Company shall indemnify Indemnitee in accordance with the provisions of this Section 3 if Indemnitee is, or is threatened to be made, a party to or a participant in any Proceeding, other than a Proceeding by or in the right of the Company to procure a judgment in its favor, by reason of Indemnitee's Corporate Status. Pursuant to this Section 3, Indemnitee shall be indemnified to the fullest extent permitted by applicable law against all Expenses, judgments, fines and, subject to Section 11(c), amounts paid in settlement (including all interest, assessments and other charges paid or payable in connection with or in respect of such Expenses, judgments, fines and amounts paid in settlement) actually and reasonably incurred by Indemnitee or on Indemnitee's behalf in connection with such Proceeding or any claim, issue or matter therein, if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in or not opposed to the best interests of the Company and, in the case of a criminal Proceeding had no reasonable cause to believe that Indemnitee's conduct was unlawful. The parties hereto intend that this Agreement shall provide to the fullest extent permitted by law for indemnification in excess of that expressly permitted by statute, including, without limitation, any indemnification provided by the Organizational Documents, vote of the Company's stockholders or disinterested directors or applicable law.

Section 4. <u>Indemnity in Proceedings by or in the Right of the Company.</u> The Company shall indemnify Indemnitee in accordance with the provisions of this Section 4 if Indemnitee is, or is threatened to be made, a party to or a participant in any Proceeding by or in the right of the Company to procure a judgment in its favor by reason of Indemnitee's Corporate Status. Pursuant to this Section 4, Indemnitee shall be indemnified to the fullest extent permitted by applicable law against all Expenses actually and reasonably incurred by Indemnitee or on Indemnitee's behalf in connection with such Proceeding or any claim, issue or matter therein, if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in or not opposed to the best interests of the Company. No indemnification for Expenses shall be made under this Section 4 in respect of any claim, issue or matter as to which Indemnitee shall have been finally adjudged by a court to be liable to the Company, unless and only to the extent that the Delaware Court (as hereinafter defined) or any court in which the Proceeding was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, Indemnitee is fairly and reasonably entitled to indemnification.

Section 5. <u>Indemnification for Expenses of a Party Who is Wholly or Partly Successful.</u> To the fullest extent permitted by applicable law and to the extent that Indemnitee is a party to (or a participant in) and is successful, on the merits or otherwise, in any Proceeding or in defense of any claim, issue or matter therein, in whole or in part, the Company shall indemnity Indemnitee against all Expenses actually and reasonably incurred by or on behalf of Indemnitee in connection therewith. If Indemnitee is not wholly successful in such Proceeding but is successful, on the merits or otherwise, as to one or more but less than all claims, issues or matters in such Proceeding, the Company shall indemnify Indemnitee against all Expenses actually and reasonably incurred by Indemnitee or on Indemnitee's behalf in connection with or related to each successfully resolved claim, issue or matter to the fullest extent permitted by law. For purposes of this Section 5 and without limitation, the termination of any claim, issue or matter in such a Proceeding by dismissal, with or without prejudice, shall be deemed to be a successful result as to such claim, issue or matter.

Section 6. <u>Indemnification For Expenses of a Witness.</u> Notwithstanding any other provision of this Agreement, to the fullest extent permitted by applicable law and to the extent that Indemnitee is, by reason of Indemnitee's Corporate Status, a witness, is or was made (or asked) to respond to discovery requests or otherwise asked to participate in any Proceeding to which Indemnitee is not a party, Indemnitee shall be indemnified against all Expenses actually and reasonably incurred by Indemnitee or on Indemnitee's behalf in connection therewith.

Section 7. <u>Partial Indemnification</u>. If Indemnitee is entitled under any provision of this Agreement to indemnification by the Company for some or a portion of Expenses, but not, however, for the total amount thereof, the Company shall nevertheless indemnify Indemnitee for the portion thereof to which Indemnitee is entitled.

Section 8. Additional Indemnification.

- (a) Notwithstanding any limitation in Sections 3, 4, or 5, the Company shall indemnify Indemnitee to the fullest extent permitted by applicable law if Indemnitee is a party to, or is threatened to be made a party to, or a participant in any Proceeding (including a Proceeding by or in the right of the Company to procure a judgment in its favor) by reason of Indemnitee's Corporate Status.
- (b) For purposes of Section 8(a), the meaning of the phrase "to the fullest extent permitted by applicable law" shall include, but not be limited to:
- i. to the fullest extent permitted by the provision of the DGCL that authorizes or contemplates additional indemnification by agreement, or the corresponding provision of any amendment to or replacement of the DGCL, and
- ii. to the fullest extent authorized or permitted by any amendments to or replacements of the DGCL adopted after the date of this Agreement that increase the extent to which a corporation may indemnify its officers and directors.
- Section 9. <u>Exclusions.</u> Notwithstanding any provision in this Agreement, the Company shall not be obligated under this Agreement to make any indemnification payment in connection with any claim involving Indemnitee:
- (a) for which payment has actually been made to or on behalf of Indemnitee under any insurance policy or other indemnity provision, except with respect to any excess beyond the amount paid under any insurance policy or other indemnity provision; or
- (b) for (i) an accounting of profits made from the purchase and sale (or sale and purchase) by Indemnitee of securities of the Company within the meaning of Section 16(b) of the Exchange Act (as defined in Section 2(b) hereof) or similar provisions of state statutory law or common law, (ii) any reimbursement of the Company by the Indemnitee of any bonus or other incentive-based or equity-based compensation or of any profits realized by the Indemnitee from the sale of securities of the Company, as required in each case under the Exchange Act (including any such reimbursements that arise from an accounting restatement of the Company pursuant to Section 304 of the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act"), or the payment to the Company of profits arising from the purchase and sale by Indemnitee of securities in violation of Section 306 of the Sarbanes-Oxley Act), or (iii) any reimbursement of the Company by Indemnitee of any compensation pursuant to any compensation recoupment or clawback policy that may be adopted by the Board or a committee of the Board: or

(c) except as provided in Section 14(d) of this Agreement, in connection with any Proceeding (or any part of any Proceeding) initiated by Indemnitee, including any Proceeding (or any part of any Proceeding) initiated by Indemnitee against the Company or its directors, officers, employees or other indemnitees, unless (i) the Board authorized the Proceeding (or any part of any Proceeding) prior to its initiation, or (ii) such payment arises in connection with any mandatory counterclaim or cross-claim brought or raised by Indemnitee in any Proceeding (or any part of any Proceeding) or (iii) the Company provides the indemnification, in its sole discretion, pursuant to the powers vested in the Company under applicable law.

Section 10. Advances of Expenses. Notwithstanding any provision of this Agreement to the contrary (other than Section 14(d)), the Company shall advance, to the extent not prohibited by law, the Expenses incurred by or on behalf of Indemnitee in connection with any Proceeding (or any part of any Proceeding) not initiated by Indemnitee or any Proceeding initiated by Indemnitee with the prior approval of the Board as provided in Section 9(c), and such advancement shall be made as soon as possible but in any event no later than thirty (30) days after the receipt by the Company of a statement or statements requesting such advances from time to time, whether prior to or after final disposition of any Proceeding. Advances shall be unsecured and interest free. The Company shall, in accordance with such request for advancement (but without duplication), either (i) pay such Expenses on behalf of Indemnitee, or (ii) reimburse Indemnitee for such Expenses. Advances shall be made without regard to Indemnitee's ability to repay the Expenses and without regard to Indemnitee's ultimate entitlement to indemnification under the other provisions of this Agreement. In accordance with Section 14(d), advances shall include any and all reasonable Expenses incurred pursuing an action to enforce this right of advancement, including Expenses incurred preparing and forwarding statements to the Company to support the advances claimed. The Indemnitee shall qualify for advances upon the execution and delivery to the Company of this Agreement, which shall constitute an undertaking providing that the Indemnitee undertakes to repay the amounts advanced (without interest) by the Company pursuant to this Section 10, if and only to the extent that it is ultimately determined that Indemnitee is not entitled to be indemnified by the Company. No other form of undertaking shall be required other than the execution of this Agreement. This Section 10 shall not apply to any claim made by Indemnitee for which indemnity is excluded pursuant to Section 9.

Section 11. Procedure for Notification and Defense of Claim.

(a) Indemnitee shall notify the Company in writing of any matter with respect to which Indemnitee intends to seek indemnification or advancement of Expenses hereunder as soon as reasonably practicable following the receipt by Indemnitee of written notice thereof. The written notification to the Company shall include a description of the nature of the Proceeding and the facts underlying the Proceeding. To obtain indemnification under this Agreement, Indemnitee shall submit to the Company a written request, including therein or therewith such documentation and information as is reasonably available to Indemnitee and is reasonably necessary to determine whether and to what extent Indemnitee is entitled to indemnification following the final disposition of such Proceeding. The omission by Indemnitee to notify the Company hereunder will not relieve the Company from any liability which it may have to Indemnitee hereunder or otherwise than under this Agreement, and any delay in so notifying the Company shall not constitute a waiver by Indemnitee of any rights under this Agreement. The Secretary of the Company shall, promptly upon receipt of such a request for indemnification, advise the Board in writing that Indemnitee has requested indemnification. In addition, Indemnitee shall give the Company such information and cooperation as it may reasonably require and as shall be within Indemnitee's power, to the extent that doing so is consistent with the exercise of the Indemnitee's rights under the federal and state Constitutions. The Company shall provide Indemnitee with such information and cooperation as Indemnitee may reasonably require, to the extent that doing so is consistent with the Company's obligation to cooperate with regulatory or law enforcement agencies.

(b) The Company shall be entitled to participate in the defense of any Proceeding entitled to indemnification under this Agreement or to assume the defense thereof, with counsel chosen by the Company and reasonably satisfactory to Indemnitee (not to be unreasonably withheld) upon delivery to Indemnitee of written notice of the Company's election to do so; provided, however, that if Indemnitee believes, after consultation with counsel selected by Indemnitee, that (i) the use of counsel chosen by the Company to represent Indemnitee would present such counsel with an actual or potential conflict of interest, (ii) the named parties in such Proceeding (including any impleaded parties) include both the Company and Indemnitee and the Indemnitee concludes that there may be one or more legal defense available to him that are different from or in addition to those available to the Company, or (iii) any such representation by such counsel would be precluded under the applicable standards of professional conduct then prevailing, then Indemnitee shall be entitled to retain separate counsel that is selected by Indemnitee and approved by the Company (which approval shall not be unreasonably delayed, conditioned or withheld) (but not more than one law firm plus, if applicable, local counsel in respect of any particular Proceeding), and all Expenses related to such separate counsel shall be borne by the Company.

(c) The Company shall not be liable to Indemnitee under this Agreement for any amounts paid in settlement of any threatened or pending Proceeding pursuant to which the Indemnitee is entitled to indemnification and that is effected without the Company's prior written consent, which shall not be unreasonably withheld; provided, however, that if a Change in Control has occurred, the Company shall be liable for indemnification of the Indemnitee for amounts paid in settlement if an Independent Counsel has approved the settlement. The Company shall not settle any Proceeding pursuant to which the Indemnitee is entitled to indemnification in any manner that would impose any Expenses, claims, liabilities and/or damages on the Indemnitee without the Indemnitee's prior written consent.

Section 12. Procedure Upon Application for Indemnification.

(a) Upon written request by Indemnitee for indemnification pursuant to Section 11(a), a determination, if required by applicable law, with respect to Indemnitee's entitlement thereto shall be made in the specific case: (i) if a Change in Control shall have occurred, by Independent Counsel in a written opinion to the Board, a copy of which shall be delivered to Indemnitee; or (ii) if a Change in Control shall not have occurred, (A) by a majority vote of the Disinterested Directors, even though less than a quorum of the Board, (B) by a committee of Disinterested Directors designated by a majority vote of the Disinterested Directors, even though less than a quorum of the Board, (C) if there are no such Disinterested Directors or, if such Disinterested Directors so direct, by Independent Counsel in a written opinion to the Board, a copy of which shall be delivered to Indemnitee or (D) if so directed by the Board, by the stockholders of the Company; and, if it is so determined that Indemnitee is entitled to indemnification, payment to Indemnitee shall be made within ten (10) days after such determination. Indemnitee shall cooperate with the person, persons or entity making such determination with respect to Indemnitee's entitlement to indemnification, including providing to such person, persons or entity upon reasonable advance request any documentation or information which is not privileged or otherwise protected from disclosure and which is reasonably available to Indemnitee and reasonably necessary to such determination. Any costs or Expenses (including attorneys' fees and disbursements) incurred by or on behalf of Indemnitee in so cooperating with the person, persons or entity making such determination shall be borne by the Company (irrespective of the determination as to Indemnitee's entitlement to indemnification) and the Company hereby indemnifies and agrees to hold Indemnitee harmless therefrom. The Company promptly will advise Indemnitee in writing with respect to any determination that Indemnitee is or is

(b) In the event the determination of entitlement to indemnification is to be made by Independent Counsel pursuant to Section 12(a) hereof, the Independent Counsel shall be selected as provided in this Section 12(b). If a Change in Control shall not have occurred, the Independent Counsel shall be selected by the Board, and the Company shall give written notice to Indemnitee advising Indemnitee of the identity of the Independent Counsel so selected. If a Change in Control shall have occurred, the Independent Counsel shall be selected by Indemnitee (unless Indemnitee shall request that such selection be made by the Board, in which event the preceding sentence shall apply), and Indemnitee shall give written notice to the Company advising it of the identity of the Independent Counsel so selected. In either event, Indemnitee or the Company, as the case may be, may, within ten (10) days after such written notice of selection shall have been given, deliver to the Company or to Indemnitee, as the case may be, a written objection to such selection; provided, however, that such objection may be asserted only on the ground that the Independent Counsel so selected does not meet the requirements of "Independent Counsel" as defined in Section 2 of this Agreement, and the objection shall set forth with particularity the factual basis of such assertion. Absent a proper and timely objection, the person so selected shall act as Independent Counsel. If such written objection is so made and substantiated, the Independent Counsel so selected may not serve as Independent Counsel unless and until such objection is withdrawn or the Delaware Court has determined that such objection is without merit. If, within twenty (20) days after the later of submission by Indemnitee of a written request for indemnification pursuant to Section 11(a) hereof and the final disposition of the Proceeding, no Independent Counsel shall have been selected and not objected to, either the Company or Indemnitee may petition the Delaware Court for resolution of any objection which shall have been made by the Company or Indemnitee to the other's selection of Independent Counsel and/or for the appointment as Independent Counsel of a person selected by such court or by such other person as such court shall designate, and the person with respect to whom all objections are so resolved or the person so appointed shall act as Independent Counsel under Section 12(a) hereof. Upon the due commencement of any judicial proceeding or arbitration pursuant to Section 14(a) of this Agreement, Independent Counsel shall be discharged and relieved of any further responsibility in such capacity (subject to the applicable standards of professional conduct then prevailing).

(c) If the Company disputes a portion of the amounts for which indemnification is requested, the undisputed portion shall be paid and only the disputed portion withheld pending resolution of any such dispute.

Section 13. Presumptions and Effect of Certain Proceedings.

- (a) In making a determination with respect to entitlement to indemnification hereunder, the person or persons or entity making such determination shall, to the fullest extent not prohibited by law, presume that Indemnitee is entitled to indemnification under this Agreement if Indemnitee has submitted a request for indemnification in accordance with Section 11(a) of this Agreement, and the Company shall, to the fullest extent not prohibited by law, have the burden of proof to overcome that presumption in connection with the making by any person, persons or entity of any determination contrary to that presumption. Neither the failure of the Company (including by its directors or Independent Counsel) to have made a determination prior to the commencement of any action pursuant to this Agreement that indemnification is proper in the circumstances because Indemnitee has met the applicable standard of conduct, nor an actual determination by the Company (including by its directors or Independent Counsel) that Indemnitee has not met such applicable standard of conduct, shall be a defense to the action or create a presumption that Indemnitee has not met the applicable standard of conduct.
- (b) Subject to Section 14(e), if the person, persons or entity empowered or selected under Section 12 of this Agreement to determine whether Indemnitee is entitled to indemnification shall not have made a determination within sixty (60) days after receipt by the Company of the request therefor, the requisite determination of entitlement to indemnification shall, to the fullest extent not prohibited by law, be deemed to have been made and Indemnitee shall be entitled to such indemnification, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law; provided, however, that such 60-day period may be extended for a reasonable time, not to exceed an additional thirty (30) days, if the person, persons or entity making the determination with respect to entitlement to indemnification in good faith requires such additional time for the obtaining or evaluating of documentation and/or information relating thereto; and provided, further, that the foregoing provisions of this Section 13(b) shall not apply (i) if the determination of entitlement to indemnification is to be made by the stockholders pursuant to Section 12(a) of this Agreement and if (A) within fifteen (15) days after receipt by the Company of the request for such determination the Board has resolved to submit such determination to the stockholders for their consideration at an annual meeting thereof to be held within seventy-five (75) days after such receipt and such determination is made thereat, or (B) a special meeting of stockholders is called within fifteen (15) days after such receipt for the purpose of making such determination, such meeting is held for such purpose within sixty (60) days after having been so called and such determination is made thereat, or (ii) if the determination of entitlement to indemnification is to be made by
- (c) The termination of any Proceeding or of any claim, issue or matter therein, by judgment, order, settlement or conviction, or upon a plea of <u>nolo contendere</u> or its equivalent, shall not (except as otherwise expressly provided in this Agreement) of itself adversely affect the right of Indemnitee to indemnification or create a presumption that Indemnitee did not act in good faith and in a manner which Indemnitee reasonably believed to be in or not opposed to the best interests of the Company or, with respect to any criminal Proceeding, that Indemnitee had reasonable cause to believe that Indemnitee's conduct was unlawful.
- (d) For purposes of any determination of good faith, Indemnitee shall be deemed to have acted in good faith if Indemnitee's action is based on the records or books of account of the Enterprise, including financial statements, or on information supplied to Indemnitee by the directors or officers of the Enterprise in the course of their duties, or on the advice of legal counsel for the Enterprise or on information or records given or reports made to the Enterprise by an independent certified public accountant or by an appraiser, financial advisor or other expert selected with reasonable care by or on behalf of the Enterprise. The provisions of this Section 13(d) shall not be deemed to be exclusive or to limit in any way the other circumstances in which the Indemnitee may be deemed to have met the applicable standard of conduct set forth in this Agreement.
- (e) The knowledge and/or actions, or failure to act, of any director, officer, trustee, partner, managing member, fiduciary, agent or employee of the Enterprise shall not be imputed to Indemnitee for purposes of determining the right to indemnification under this Agreement.

Section 14. Remedies of Indemnitee.

- (a) Subject to Section 14(e), in the event that (i) a determination is made pursuant to Section 12 of this Agreement that Indemnitee is not entitled to indemnification under this Agreement, (ii) advancement of Expenses is not timely made pursuant to Section 10 of this Agreement, (iii) no determination of entitlement to indemnification shall have been made pursuant to Section 12(a) of this Agreement within ninety (90) days after receipt by the Company of the request for indemnification, (iv) payment of indemnification is not made pursuant to Section 5, 6 or 7 or the penultimate sentence of Section 12(a) of this Agreement within ten (10) days after receipt by the Company of a written request therefor, (v) payment of indemnification pursuant to Section 3, 4 or 8 of this Agreement is not made within ten (10) days after a determination has been made that Indemnitee is entitled to indemnification, or (vi) the Company or any other person takes or threatens to take any action to declare this Agreement void or unenforceable, or institutes any litigation or other action or Proceeding designed to deny, or to recover from, the Indemnitee the benefits provided or intended to be provided to the Indemnitee hereunder, Indemnitee shall be entitled to an adjudication by a court of Indemnitee's entitlement to such indemnification or advancement of Expenses. Alternatively, Indemnitee, at Indemnitee's option, may seek an award in arbitration to be conducted by a single arbitrator pursuant to the Commercial Arbitration Rules of the American Arbitration Association. Indemnitee shall commence such proceeding seeking an adjudication or an award in arbitration within 180 days following the date on which Indemnitee first has the right to commence such proceeding pursuant to this Section 14(a). The Company shall not oppose Indemnitee's right to seek any such adjudication or award in arbitration.
- (b) In the event that a determination shall have been made pursuant to Section 12(a) of this Agreement that Indemnitee is not entitled to indemnification, any judicial proceeding or arbitration commenced pursuant to this Section 14 shall be conducted in all respects as a <u>de novo</u> trial, or arbitration, on the merits and Indemnitee shall not be prejudiced by reason of that adverse determination. In any judicial proceeding or arbitration commenced pursuant to this Section 14 the Company shall have the burden of proving Indemnitee is not entitled to indemnification or advancement of Expenses, as the case may be.
- (c) If a determination shall have been made pursuant to Section 12(a) of this Agreement that Indemnitee is entitled to indemnification, the Company shall be bound by such determination in any judicial proceeding or arbitration commenced pursuant to this Section 14, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law.
- (d) The Company shall, to the fullest extent not prohibited by law, be precluded from asserting in any judicial proceeding or arbitration commenced pursuant to this Section 14 that the procedures and presumptions of this Agreement are not valid, binding and enforceable and shall stipulate in any such court or before any such arbitrator that the Company is bound by all the provisions of this Agreement. It is the intent of the Company that, to the fullest extent permitted by law, the Indemnitee not be required to incur legal fees or other Expenses associated with the interpretation, enforcement or defense of Indemnitee's rights under this Agreement by litigation or otherwise because the cost and expense thereof would substantially detract from the benefits intended to be extended to the Indemnitee hereunder. The Company shall, to the fullest extent permitted by law, indemnity Indemnitee against any and all Expenses and, if requested by Indemnitee, shall (within ten (10) days after receipt by the Company of a written request therefor) advance, to the extent not prohibited by law, such Expenses to Indemnitee, which are incurred by or on behalf of Indemnitee in connection with any action brought by Indemnitee for indemnification or advancement of Expenses from the Company under this Agreement or under any directors' and officers' liability insurance policies maintained by the Company if, in the case of indemnification, Indemnitee is wholly successful on the underlying claims; if Indemnitee is not wholly successful on the underlying claims, then such indemnification shall be only to the extent Indemnitee is successful on such underlying claims or otherwise as permitted by law, whichever is greater.
- (e) Notwithstanding anything in this Agreement to the contrary, no determination as to entitlement of Indemnitee to indemnification under this Agreement shall be required to be made prior to the final disposition of the Proceeding.

Section 15. Non-exclusivity; Survival of Rights; Insurance; Subrogation.

(a) The rights of indemnification and to receive advancement of Expenses as provided by this Agreement (i) shall not be deemed exclusive of any other rights to which Indemnitee may at any time be entitled under applicable law, the Organizational Documents, any agreement, a vote of stockholders or a resolution of directors, or otherwise and (ii) shall be interpreted independently of, and without reference to, any other such rights to which Indemnitee may at any time be entitled. No amendment, alteration or repeal of this Agreement or of any provision hereof shall limit or restrict any right of Indemnitee under this Agreement in respect of any action taken or omitted by Indemnitee in Indemnitee's Corporate Status prior to such amendment, alteration or repeal. To the extent that a change in Delaware law (or other applicable law), whether by statute or judicial decision, permits greater indemnification or advancement of Expenses than would be afforded currently under the Organizational Documents and this Agreement, it is the intent of the parties hereto that Indemnitee shall enjoy by this Agreement the greater benefits so afforded by such change. No right or remedy herein conferred is intended to be exclusive of any other right or remedy, and every other right and remedy shall be cumulative and in addition to every other right and remedy given hereunder or now or hereafter existing at law or in equity or otherwise. The assertion or employment of any right or remedy hereunder, or otherwise, shall not prevent the concurrent assertion or employment of any other right or remedy.

- (b) To the extent that the Company maintains an insurance policy or policies providing liability insurance for directors, officers, employees, or agents of the Enterprise, Indemnitee shall be covered by such policy or policies in accordance with its or their terms to the maximum extent of the coverage available for any such director, officer, employee or agent under such policy or policies. If, at the time of the receipt of a notice of a claim pursuant to the terms hereof, the Company has director and officer liability insurance in effect, the Company shall give prompt notice of such claim or of the commencement of a Proceeding, as the case may be, to the insurers in accordance with the procedures set forth in the respective policies. The Company shall thereafter take all necessary or desirable action to cause such insurers to pay, on behalf of the Indemnitee, all amounts payable as a result of such Proceeding in accordance with the terms of such policies.
- (c) In the event of any payment made by the Company under this Agreement, the Company shall be subrogated, to the fullest extent permitted by law, to the extent of such payment to all of the rights of recovery of Indemnitee, who shall execute all papers required and take all action necessary to secure such rights, including execution of such documents as are necessary to enable the Company to bring suit to enforce such rights. No such payment by the Company shall be deemed to relieve any insurer of its obligations.
- (d) Indemnitee hereby agrees that it does not have any right, title, interest or claim of any kind in or to any monies in the trust account established in connection with Growth Capital Acquisition Corp.'s, a Delaware corporation ("GCAC") initial public offering for the benefit of GCAC and holders of shares issued in such offering, and hereby waives any such claim it may have in the future as a result of, or arising out of, any services provided to GCAC and will not seek recourse against such trust account for any reason whatsoever.
- (e) The Company shall not be liable under this Agreement to make any payment of amounts otherwise indemnifiable hereunder (or for which advancement is provided hereunder) if and to the extent that Indemnitee has otherwise actually received such payment under any insurance policy, contract, agreement or otherwise.
- (f) The Company's obligation to indemnify or advance Expenses hereunder to Indemnitee who is or was serving at the request of the Company as a director, officer, trustee, partner, managing member, fiduciary, employee or agent of any other corporation, limited liability company, partnership, joint venture, trust, employee benefit plan or other enterprise shall be reduced by any amount Indemnitee has actually received as indemnification or advancement of Expenses from such other corporation, limited liability company, partnership, joint venture, trust or other enterprise.
- Section 16. <u>Duration of Agreement.</u> This Agreement shall continue until and terminate upon the later of: (a) ten (10) years after the date that Indemnitee shall have ceased to serve as an officer and director of the Company and of any other corporation, limited liability company, partnership or joint venture, trust or other enterprise which such person is or was serving at the request of the Company, or (b) one (1) year after the final termination of any Proceeding then pending in respect of which Indemnitee is granted rights of indemnification or advancement of Expenses hereunder and of any proceeding (including any appeal thereof) commenced by Indemnitee pursuant to Section 14 of this Agreement relating thereto. The indemnification and advancement of expenses rights provided by or granted pursuant to this Agreement shall be binding upon and be enforceable by the parties hereto and their respective successors and assigns (including any direct or indirect successor by purchase, merger, consolidation or otherwise to all or substantially all of the business or assets of the Company), shall continue as to an Indemnitee who has ceased to be a director, officer, employee or agent of the Company or of any other Enterprise, and shall inure to the benefit of Indemnitee and Indemnitee's spouse, assigns, heirs, devisees, executors and administrators and other legal representatives. The Company shall require and shall cause any successor (whether direct or indirect by purchase, merger, consolidation or otherwise) to all or substantially all of the business or assets of the Company to, by written agreement, expressly assume and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform if no such succession had taken place.
- Section 17. Severability. Nothing in this Agreement is intended to require or shall be construed as requiring the Company to do or fail to do any act in violation of applicable law. If any provision or provisions of this Agreement shall be held to be invalid, illegal or unenforceable for any reason whatsoever: (a) the validity, legality and enforceability of the remaining provisions of this Agreement (including without limitation, each portion of any Section of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that is not itself invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby and shall remain enforceable to the fullest extent permitted by law; (b) such provision or provisions shall be deemed reformed to the extent necessary to conform to applicable law and to give the maximum effect to the intent of the parties hereto; and (c) to the fullest extent possible, the provisions of this Agreement (including, without limitation, each portion of any Section of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that is not itself invalid, illegal or unenforceable) shall be construed so as to give effect to the intent manifested thereby.

Section 18. Enforcement.

- (a) The Company expressly confirms and agrees that it has entered into this Agreement and assumed the obligations imposed on it hereby in order to induce Indemnitee to serve as a director or officer of the Company, and the Company acknowledges that Indemnitee is relying upon this Agreement in serving as a director or officer of the Company or of any other corporation, limited liability company, partnership or joint venture, trust or other enterprise which such person is serving at the request of the Company.
- (b) This Agreement constitutes the entire agreement between the parties hereto with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral, written and implied, between the parties hereto with respect to the subject matter hereof; <u>provided</u>, <u>however</u>, that this Agreement is a supplement to and in furtherance of the Organizational Documents, any directors' and officers' insurance maintained by the company and applicable law, and shall not be deemed a substitute therefor, nor to diminish or abrogate any rights of Indemnitee thereunder.

Section 19. Section 409A. If Indemnitee's right to payment or reimbursement of indemnification or expenses pursuant to this Agreement would not be exempt from Section 409A of the Internal Revenue Code of 1986, as amended (the "Code") pursuant to Treasury Regulation Section 1.0409A-1(b)(10), then (a) the payment or reimbursement of indemnification and expenses provided or advanced to or for Indemnitee pursuant to this Agreement in one taxable year shall not affect the amount of indemnification and expenses provided or advanced to or for Indemnitee in any other taxable year, (b) any reimbursement to Indemnitee of expenses under this Agreement shall be paid to Indemnitee on or before the last day of Indemnitee's taxable year following the taxable year in which the expense was incurred and (c) the right to advancement, reimbursement or payment of indemnification and expenses under this Agreement may not be liquidated or exchanged for any other benefit. In addition, to the extent that this Agreement is subject to Section 409A of the Code, this Agreement shall be interpreted and enforced so as to avoid any tax, penalty or interest under Section 409A of the Code. For purposes of this Section 19, expenses shall be deemed to include in addition to those items included in the definition thereof in Section 2, any liability, loss, judgment, fine and amounts paid in settlement.

Section 20. <u>Modification and Waiver.</u> No supplement, modification or amendment of this Agreement shall be binding unless executed in writing by the parties hereto. No waiver of any of the provisions of this Agreement shall be deemed or shall constitute a waiver of any other provisions of this Agreement nor shall any waiver constitute a continuing waiver.

Section 21. Notice by Indemnitee. Indemnitee agrees promptly to notify the Company in writing upon being served with any summons, citation, subpoena, complaint, indictment, information or other document relating to any Proceeding or matter which may be subject to indemnification or advancement of Expenses covered hereunder. Any failure by Indemnitee to notify the Company will relieve the Company of its advancement or indemnification obligations under this Agreement only to the extent the Company can establish that such omission to notify resulted in actual and material prejudice to it which cannot be reversed or otherwise eliminated without any material negative effect on the Company, and the omission to notify the Company will, in any event, not relieve the Company from any liability which it may have to the Indemnitee under this Agreement or otherwise.

Section 22. <u>Notices.</u> All notices, requests, demands and other communications under this Agreement shall be in writing and shall be deemed to have been duly given if (a) delivered by hand and receipted for by the party to whom said notice or other communication shall have been directed, (b) mailed by certified or registered mail with postage prepaid, on the third business day after the date on which it is so mailed or (c) mailed by reputable overnight courier and receipted for by the party to whom said notice or other communication shall have been directed:

(a) If to Indemnitee, at the address indicated on the signature page of this Agreement, or such other address as Indemnitee shall provide to the Company.

(b) If to the Company to

Cepton, Inc. 399 West Trimble Road San Jose, CA 95131

Attention: Corporate Secretary

or to any other address as may have been furnished to Indemnitee by the Company.

Section 23. <u>Contribution.</u> To the fullest extent permissible under applicable law, if the indemnification provided for in this Agreement is unavailable to Indemnitee for any reason whatsoever, the Company, in lieu of indemnifying Indemnitee, shall contribute to the amount incurred by Indemnitee, whether for judgments, fines, penalties, excise taxes, amounts paid or to be paid in settlement and/or for Expenses, in connection with any claim relating to an indemnifiable event under this Agreement, in such proportion as is deemed fair and reasonable in light of all of the circumstances of such Proceeding in order to reflect (i) the relative benefits received by the Company and Indemnitee as a result of the event(s) and/or transaction(s) giving cause to such Proceeding; and/or (ii) the relative fault of the Company (and its directors, officers, employees and agents) and Indemnitee in connection with such event(s) and/or transaction(s).

Section 24. <u>Applicable Law and Consent to Jurisdiction</u>. This Agreement and the legal relations among the parties shall be governed by, and construed and enforced in accordance with, the laws of the State of Delaware, without regard to its conflict of laws rules. Except with respect to any arbitration commenced by Indemnitee pursuant to Section 14(a) of this Agreement, the Company and Indemnitee hereby irrevocably and unconditionally (i) agree that any action or proceeding arising out of or in connection with this Agreement shall be brought only in the Court of Chancery of the State of Delaware (the "Delaware Court"), and not in any other state or federal court in the United States of America or any court in any other country, (ii) consent to submit to the exclusive jurisdiction of the Delaware Court for purposes of any action or proceeding arising out of or in connection with this Agreement, (iii) appoint, to the extent such party is not otherwise subject to service of process in the State of Delaware, irrevocably Corporation Service Company, 2711 Centerville Road, Suite 400, Wilmington, New Castle County, Delaware 19808 as its agent in the State of Delaware as such party's agent for acceptance of legal process in connection with any such action or proceeding against such party with the same legal force and validity as if served upon such party personally within the State of Delaware, (iv) waive any objection to the laying of venue of any such action or proceeding in the Delaware Court, and (v) waive, and agree not to plead or to make, any claim that any such action or proceeding brought in the Delaware Court has been brought in an improper or inconvenient forum.

Section 25. <u>Identical Counterparts</u>. This Agreement may be executed in one or more counterparts, each of which shall for all purposes be deemed to be an original but all of which together shall constitute one and the same Agreement. Only one such counterpart signed by the party against whom enforceability is sought needs to be produced to evidence the existence of this Agreement.

Section 26. <u>Miscellaneous.</u> Use of the masculine pronoun shall be deemed to include usage of the feminine pronoun where appropriate. The headings of this Agreement are inserted for convenience only and shall not be deemed to constitute part of this Agreement or to affect the construction thereof.

IN WITNESS WHEREOF, the parties have caused this Agreement to be signed as of the day and year first above written.

CEPTON, INC.	INDEMN	NITEE
By: Name: Office:	By: Name: Address:	
	[Signature Page to Indemnification Agreement]	

CEPTON, INC. 2022 EQUITY INCENTIVE PLAN

1. PURPOSE OF PLAN

The purpose of this Cepton, Inc. 2022 Equity Incentive Plan (this "**Plan**") of Cepton, Inc., a Delaware corporation (the "**Corporation**"), is to promote the success of the Corporation by providing an additional means through the grant of awards to attract, motivate, retain and reward selected employees and other eligible persons and to enhance the alignment of the interests of the selected participants with the interests of the Corporation's stockholders.

2. ELIGIBILITY

The Administrator (as such term is defined in Section 3.1) may grant awards under this Plan only to those persons that the Administrator determines to be Eligible Persons. An "Eligible Person" is any person who is either: (a) an officer (whether or not a director) or employee of the Corporation or one of its Subsidiaries; (b) a director of the Corporation or one of its Subsidiaries; or (c) an individual consultant or advisor who renders or has rendered bona fide services (other than services in connection with the offering or sale of securities of the Corporation or one of its Subsidiaries in a capital-raising transaction or as a market maker or promoter of securities of the Corporation or one of its Subsidiaries) to the Corporation or one of its Subsidiaries and who is selected to participate in this Plan by the Administrator; provided, however, that a person who is otherwise an Eligible Person under clause (c) above may participate in this Plan only if such participation would not adversely affect either the Corporation's eligibility to use Form S-8 to register under the Securities Act of 1933, as amended (the "Securities Act"), the offering and sale of shares issuable under this Plan by the Corporation or the Corporation's compliance with any other applicable laws. An Eligible Person who has been granted an award (a "participant") may, if otherwise eligible, be granted additional awards if the Administrator shall so determine. As used herein, "Subsidiary" means any corporation or other entity a majority of whose outstanding stock or voting power is beneficially owned directly or indirectly by the Corporation; and "Board" means the Board of Directors of the Corporation.

3. PLAN ADMINISTRATION

- 3.1 The Administrator. This Plan shall be administered by and all awards under this Plan shall be authorized by the Administrator. The "Administrator" means the Board or one or more committees (or subcommittees, as the case may be) appointed by the Board or another committee (within its delegated authority) to administer all or certain aspects of this Plan. Any such committee shall be comprised solely of one or more directors or such number of directors as may be required under applicable law. A committee may delegate some or all of its authority to another committee so constituted. The Board or a committee comprised solely of directors may also delegate, to the extent permitted by applicable law, to one or more officers of the Corporation, its authority under this Plan. The Board or another committee (within its delegated authority) may delegate different levels of authority to different committees or persons with administrative and grant authority under this Plan. Unless otherwise provided in the Bylaws of the Corporation or the applicable charter of any Administrator: (a) a majority of the members of the acting Administrator shall constitute a quorum, and (b) the vote of a majority of the members present assuming the presence of a quorum or the unanimous written consent of the members of the Administrator shall constitute action by the acting Administrator.
- **3.2 Powers of the Administrator.** Subject to the express provisions of this Plan, the Administrator is authorized and empowered to do all things necessary or desirable in connection with the authorization of awards and the administration of this Plan (in the case of a committee or delegation to one or

more officers, within any express limits on the authority delegated to that committee or person(s)), including, without limitation, the authority to:

- (a) determine eligibility and, from among those persons determined to be eligible, determine the particular Eligible Persons who will receive an award under this Plan;
- (b) grant awards to Eligible Persons, determine the price (if any) at which securities will be offered or awarded and the number of securities to be offered or awarded to any of such persons (in the case of securities-based awards), determine the other specific terms and conditions of awards consistent with the express limits of this Plan, establish the installment(s) (if any) in which such awards shall become exercisable or shall vest (which may include, without limitation, performance and/or time-based schedules), or determine that no delayed exercisability or vesting is required, establish any applicable performance-based exercisability or vesting requirements, determine the circumstances in which any performance-based goals (or the applicable measure of performance) will be adjusted and the nature and impact of any such adjustment, determine the extent (if any) to which any applicable exercise and vesting requirements have been satisfied, establish the events (if any) on which exercisability or vesting may accelerate (which may include, without limitation, retirement and other specified terminations of employment or services, or other circumstances), and establish the events (if any) of termination, expiration or reversion of such awards;
- (c) approve the forms of any award agreements (which need not be identical either as to type of award or among participants);
- (d) construe and interpret this Plan and any agreements defining the rights and obligations of the Corporation, its Subsidiaries, and participants under this Plan, make any and all determinations under this Plan and any such agreements, further define the terms used in this Plan, and prescribe, amend and rescind rules and regulations relating to the administration of this Plan or the awards granted under this Plan;
- (e) cancel, modify, or waive the Corporation's rights with respect to, or modify, discontinue, suspend, or terminate any or all outstanding awards, subject to any required consent under Section 8.7.5;
- (f) accelerate, waive or extend the vesting or exercisability, or modify or extend the term of, any or all such outstanding awards (in the case of options or stock appreciation rights, within the maximum term of such awards) in such circumstances as the Administrator may deem appropriate (including, without limitation, in connection with a retirement or other termination of employment or services, or other circumstances) subject to any required consent under Section 8.7.5;
- (g) adjust the number of shares of Common Stock subject to any award, adjust the price of any or all outstanding awards or otherwise waive or change previously imposed terms and conditions, in such circumstances as the Administrator may deem appropriate, in each case subject to Sections 4 and 8.6 (and subject to the no repricing provision below);
- (h) determine the date of grant of an award, which may be a designated date after but not before the date of the Administrator's action to approve the award (unless otherwise designated by the Administrator, the date of grant of an award shall be the date upon which the Administrator took the action approving the award);
- (i) determine whether, and the extent to which, adjustments are required pursuant to Section 7.1 hereof and take any other actions contemplated by Section 7 in connection with the occurrence of an event of the type described in Section 7;

- (j) acquire or settle (subject to Sections 7 and 8.6) rights under awards in cash, stock of equivalent value, or other consideration (subject to the no repricing provision below);
- (k) impose such restrictions, conditions or limitations (by provision in the applicable Award Agreement) as it determines appropriate as to the timing and manner of any resales by a participant of any shares issued pursuant to an award, including restrictions under an insider trading policy and restrictions as to the use of a specified brokerage firm for such resales; and
- (l) determine the fair market value of the Common Stock or awards under this Plan from time to time and/or the manner in which such value will be determined.
- **Prohibition on Repricing.** Notwithstanding anything to the contrary in Section 3.2 and except for an adjustment pursuant to Section 7.1 or a repricing approved by stockholders, in no case may the Administrator (a) amend an outstanding stock option or SAR to reduce the exercise price or base price of the award, (b) cancel, exchange, or surrender an outstanding stock option or SAR in exchange for cash or other awards for the purpose of repricing the award, or (c) cancel, exchange, or surrender an outstanding stock option or SAR in exchange for an option or SAR with an exercise or base price that is less than the exercise or base price of the original award.
- 3.4 Binding Determinations. Any determination or other action taken by, or inaction of, the Corporation, any Subsidiary, or the Administrator relating or pursuant to this Plan (or any award made under this Plan) and within its authority hereunder or under applicable law shall be within the absolute discretion of that entity or body and shall be conclusive and binding upon all persons. Neither the Board nor any other Administrator, nor any member thereof or person acting at the direction thereof, shall be liable for any act, omission, interpretation, construction or determination made in good faith in connection with this Plan (or any award made under this Plan), and all such persons shall be entitled to indemnification and reimbursement by the Corporation in respect of any claim, loss, damage or expense (including, without limitation, attorneys' fees) arising or resulting therefrom to the fullest extent permitted by law and/or under any directors and officers liability insurance coverage that may be in effect from time to time. Neither the Board nor any other Administrator, nor any member thereof or person acting at the direction thereof, nor the Corporation or any of its Subsidiaries, shall be liable for any damages of a participant should an option intended as an ISO (as defined below) fail to meet the requirements of the Internal Revenue Code of 1986, as amended (the "Code"), applicable to ISOs, should any other award(s) fail to qualify for any intended tax treatment, should any award grant or other action with respect thereto not satisfy Rule 16b-3 promulgated under the Securities Exchange Act of 1934, as amended, or otherwise for any tax or other liability imposed on a participant with respect to an award.
- **3.5 Reliance on Experts.** In making any determination or in taking or not taking any action under this Plan, the Administrator may obtain and may rely upon the advice of experts, including employees and professional advisors to the Corporation. No director, officer or agent of the Corporation or any of its Subsidiaries shall be liable for any such action or determination taken or made or omitted in good faith.
- **3.6 Delegation**. The Administrator may delegate ministerial, non-discretionary functions to individuals who are officers or employees of the Corporation or any of its Subsidiaries or to third parties.

4. SHARES OF COMMON STOCK SUBJECT TO THE PLAN; SHARE LIMITS

Shares Available. Subject to the provisions of Section 7.1, the capital stock that may be delivered under this Plan shall be shares of the Corporation's authorized but unissued Common Stock and any shares of its Common Stock held as treasury shares. For purposes of this Plan, "Common Stock" shall mean the common stock of the Corporation and such other securities or property as may become the subject of awards under this Plan, or may become subject to such awards, pursuant to an adjustment made under Section 7.1.

- **4.2 Aggregate Share Limit.** The maximum number of shares of Common Stock that may be delivered pursuant to awards granted to Eligible Persons under this Plan (the "**Share Limit**") is equal to the sum of the following:
 - (1) 115,123,142 shares of Common Stock, plus
 - (2) the number of any shares subject to stock options granted under the Cepton Technologies, Inc. Stock Incentive Plan (the "2016 Plan") and outstanding on the date of stockholder approval of this Plan (the "Stockholder Approval Date") which expire, or for any reason are cancelled or terminated, after the Stockholder Approval Date without being exercised, plus
 - (3) the number of any shares subject to restricted stock awards granted under the 2016 Plan that are outstanding and unvested on the Stockholder Approval Date that are forfeited, terminated, cancelled or otherwise reacquired by the Corporation without having become vested.

In addition, the Share Limit shall automatically increase on the first trading day in January of each calendar year during the term of this Plan, with the first such increase to occur in January 2023, by an amount equal to the lesser of (i) two percent (2%) of the total number of shares of Common Stock issued and outstanding on December 31 of the immediately preceding calendar year or (ii) such number of shares of Common Stock as may be established by the Board.

- **4.3** *Additional Share Limits.* The following limits also apply with respect to awards granted under this Plan. These limits are in addition to, not in lieu of, the aggregate Share Limit in Section 4.2.
 - (a) The maximum number of shares of Common Stock that may be delivered pursuant to options qualified as incentive stock options granted under this Plan is 40,000,000 shares.
 - (b) Awards that are granted under this Plan during any one calendar year to any person who, on the grant date of the award, is a nonemployee director shall not exceed the number of shares that produce a grant date fair value for the award that, when combined with (i) the grant date fair value of any other awards granted under this Plan during that same calendar year to that individual in his or her capacity as a non-employee director and (ii) the dollar amount of all other cash compensation payable by the Corporation to such non-employee director for his or her services in such capacity during that same calendar year (regardless of whether deferred and excluding any interest or earnings on any portion of such amount that may be deferred), is \$400,000; provided that this limit is \$600,000 as to (1) a non-employee director who is serving as the independent Chair of the Board or as a lead independent director at the time the applicable grant is made or (2) any new non-employee director for the calendar year in which the non-employee director is first elected or appointed to the Board. For purposes of this Section 4.3(b), a "nonemployee director" is an individual who, on the grant date of the award, is a member of the Board who is not then an officer or employee of the Corporation or one of its Subsidiaries. For purposes of this Section 4.3(b), "grant date fair value" means the value of the award as of the date of grant of the award and as determined using the equity award valuation principles applied in the Corporation's financial reporting. The limits of this Section 4.3(b) do not apply to, and shall be determined without taking into account, any award granted to an individual who, on the grant date of the award, is an officer or employee of the Corporation or one of its Subsidiaries. The limits of this Section 4.3(b) apply on an individual basis and not on an aggregate basis to all non-employee directors as a group.

- **4.4 Share-Limit Counting Rules.** The Share Limit shall be subject to the following provisions of this Section 4.4:
 - (a) Shares that are subject to or underlie awards granted under this Plan which expire or for any reason are cancelled or terminated, are forfeited, fail to vest, or for any other reason are not paid or delivered under this Plan shall not be counted against the Share Limit and shall be available for subsequent awards under this Plan.
 - (b) Except as provided below, to the extent that shares of Common Stock are delivered pursuant to the exercise of a stock appreciation right granted under this Plan, the number of underlying shares which are actually issued in payment of the award shall be counted against the Share Limit. (For purposes of clarity, if a stock appreciation right relates to 100,000 shares and is exercised in full at a time when the payment due to the participant is 15,000 shares, 15,000 shares shall be counted against the Share Limit with respect to such exercise and the 85,000 shares not issued shall not be counted against the Share Limit and shall be available for subsequent awards under this Plan.)
 - (c) Shares that are exchanged by a participant or withheld by the Corporation as full or partial payment in connection with any award granted under this Plan, as well as any shares exchanged by a participant or withheld by the Corporation or one of its Subsidiaries to satisfy the tax withholding obligations related to any award granted under this Plan, shall not be counted against the Share Limit and shall be available for subsequent awards under this Plan (but such shares shall not again become available for issuance as ISOs).
 - (d) To the extent that an award granted under this Plan is settled in cash or a form other than shares of Common Stock, the shares that would have been delivered had there been no such cash or other settlement shall not be counted against the Share Limit and shall be available for subsequent awards under this Plan.
 - (e) In the event that shares of Common Stock are delivered in respect of a dividend equivalent right granted under this Plan, the number of shares delivered with respect to the award shall be counted against the Share Limit. (For purposes of clarity, if 1,000 dividend equivalent rights are granted and outstanding when the Corporation pays a dividend, and 50 shares are delivered in payment of those rights with respect to that dividend, 50 shares shall be counted against the Share Limit). Except as otherwise provided by the Administrator, shares delivered in respect of dividend equivalent rights shall not count against any individual award limit under this Plan other than the aggregate Share Limit.
 - (f) The Corporation may not increase the Share Limit by repurchasing shares of Common Stock on the market (by using cash received through the exercise of stock options or otherwise).

Refer to Section 8.11 for application of the share limits of this Plan, including the limits in Sections 4.2 and 4.3, with respect to assumed awards. Each of the numerical limits and references in Sections 4.2 and 4.3, and in this Section 4.4, is subject to adjustment as contemplated by Sections 7 and 8.10.

4.5 No Fractional Shares; Minimum Issue. Unless otherwise expressly provided by the Administrator, no fractional shares shall be delivered under this Plan. The Administrator may pay cash in lieu of any fractional shares in settlements of awards under this Plan. The Administrator may from time to time impose a limit (of not greater than 100 shares) on the minimum number of shares that may be purchased or exercised as to awards (or any particular award) granted under this Plan unless (as to any particular award) the total number purchased or exercised is the total number at the time available for purchase or exercise under the award.

5. AWARDS

- **Type and Form of Awards.** The Administrator shall determine the type or types of award(s) to be made to each selected Eligible Person. Awards may be granted singly, in combination or in tandem. Awards also may be made in combination or in tandem with, in replacement of, as alternatives to, or as the payment form for grants or rights under any other employee or compensation plan of the Corporation or one of its Subsidiaries. The types of awards that may be granted under this Plan are:
 - **5.1.1 Stock Options.** A stock option is the grant of a right to purchase a specified number of shares of Common Stock during a specified period as determined by the Administrator. An option may be intended as an incentive stock option within the meaning of Section 422 of the Code (an "**ISO**") or a nonqualified stock option (an option not intended to be an ISO). The agreement evidencing the grant of an option will indicate if the option is intended as an ISO; otherwise it will be deemed to be a nonqualified stock option. The maximum term of each option (ISO or nonqualified) shall be ten (10) years. The per share exercise price for each option shall be not less than 100% of the fair market value of a share of Common Stock on the date of grant of the option. When an option is exercised, the exercise price for the shares to be purchased shall be paid in full in cash or such other method permitted by the Administrator consistent with Section 5.4.
 - Additional Rules Applicable to ISOs. To the extent that the aggregate fair market value (determined at the time of grant of the applicable option) of stock with respect to which ISOs first become exercisable by a participant in any calendar year exceeds \$100,000, taking into account both Common Stock subject to ISOs under this Plan and stock subject to ISOs under all other plans of the Corporation or one of its Subsidiaries (or any parent or predecessor corporation to the extent required by and within the meaning of Section 422 of the Code and the regulations promulgated thereunder), such options shall be treated as nonqualified stock options. In reducing the number of options treated as ISOs to meet the \$100,000 limit, the most recently granted options shall be reduced first. To the extent a reduction of simultaneously granted options is necessary to meet the \$100,000 limit, the Administrator may, in the manner and to the extent permitted by law, designate which shares of Common Stock are to be treated as shares acquired pursuant to the exercise of an ISO. ISOs may only be granted to employees of the Corporation or one of its subsidiaries (for this purpose, the term "subsidiary" is used as defined in Section 424(f) of the Code, which generally requires an unbroken chain of ownership of at least 50% of the total combined voting power of all classes of stock of each subsidiary in the chain beginning with the Corporation and ending with the subsidiary in question). No ISO may be granted to any person who, at the time the option is granted, owns (or is deemed to own under Section 424(d) of the Code) shares of outstanding Common Stock possessing more than 10% of the total combined voting power of all classes of stock of the Corporation, unless the exercise price of such option is at least 110% of the fair market value of a share of Common Stock on the date of grant of the option and such option by its terms is not exercisable after the expiration of five years from the date such option is granted. If an otherwise-intended ISO fails to meet the applicable requirements of Section 422 of the Code, the option shall be a nonqualified stock option.
 - **5.1.3** *Stock Appreciation Rights.* A stock appreciation right or "SAR" is a right to receive a payment, in cash and/or Common Stock, equal to the excess of the fair market value of a specified number of shares of Common Stock on the date the SAR is exercised over the "base price" of the award, which base price shall be set forth in the applicable award agreement and shall be not less than 100% of the fair market value of a share of Common Stock on the date of grant of the SAR. The maximum term of a SAR shall be ten (10) years.
 - **5.1.4** *Other Awards; Dividend Equivalent Rights.* The other types of awards that may be granted under this Plan include: (a) stock bonuses, restricted stock, performance stock, stock units, restricted stock units, deferred shares, phantom stock or similar rights to purchase or acquire shares, whether at a fixed or variable price (or no price) or fixed or variable ratio related to the Common Stock, and any of which may (but need not) be fully vested at grant or vest upon the passage of time, the

occurrence of one or more events, the satisfaction of performance criteria or other conditions, or any combination thereof; or (b) cash awards. The types of cash awards that may be granted under this Plan include the opportunity to receive a payment for the achievement of one or more goals established by the Administrator, on such terms as the Administrator may provide, as well as discretionary cash awards. Dividend equivalent rights may be granted as a separate award or in connection with another award under this Plan; provided, however, that dividend equivalent rights may not be granted as to a stock option or SAR granted under this Plan. In addition, any dividends and/or dividend equivalents as to the portion of an award that is subject to unsatisfied vesting requirements will be subject to termination and forfeiture to the same extent as the corresponding portion of the award to which they relate in the event the applicable vesting requirements are not satisfied.

- **5.2 Award Agreements.** Each award shall be evidenced by a written or electronic award agreement or notice in a form approved by the Administrator (an "award agreement"), and, in each case and if required by the Administrator, executed or otherwise electronically accepted by the recipient of the award in such form and manner as the Administrator may require.
- **Deferrals and Settlements.** Payment of awards may be in the form of cash, Common Stock, other awards or combinations thereof as the Administrator shall determine, and with such restrictions (if any) as it may impose. The Administrator may also require or permit participants to elect to defer the issuance of shares or the settlement of awards in cash under such rules and procedures as it may establish under this Plan. The Administrator may also provide that deferred settlements include the payment or crediting of interest or other earnings on the deferral amounts, or the payment or crediting of dividend equivalents where the deferred amounts are denominated in shares.
- **Consideration for Common Stock or Awards.** The purchase price (if any) for any award granted under this Plan or the Common Stock to be delivered pursuant to an award, as applicable, may be paid by means of any lawful consideration as determined by the Administrator, including, without limitation, one or a combination of the following methods:
 - (a) services rendered by the recipient of such award;
 - (b) cash, check payable to the order of the Corporation, or electronic funds transfer;
 - (c) notice and third party payment in such manner as may be authorized by the Administrator;
 - (d) the delivery of previously-owned shares of Common Stock which meet the conditions established by the Administrator to avoid adverse accounting consequences to the Corporation (as determined by the Administrator);
 - (e) by a reduction in the number of shares otherwise deliverable pursuant to the award; or
 - (f) subject to such procedures as the Administrator may adopt, pursuant to a "cashless exercise" with a third party who provides financing for the purposes of (or who otherwise facilitates) the purchase or exercise of awards.

In no event shall any shares newly issued by the Corporation be issued for less than the minimum lawful consideration for such shares or for consideration other than consideration permitted by applicable state law. Shares of Common Stock used to satisfy the exercise price of an option shall be valued at their fair market value. The Corporation will not be obligated to deliver any shares unless and until it receives full payment of the exercise or purchase price therefor and any related withholding obligations under Section 8.5 and any other conditions to exercise or purchase have been satisfied. Unless otherwise expressly provided in the applicable award agreement, the Administrator may at any time eliminate or limit a participant's ability to pay any exercise or purchase price of any award or shares by any method other than cash payment to the Corporation.

5.5 Definition of Fair Market Value. For purposes of this Plan, "fair market value" shall mean, unless otherwise determined or provided by the Administrator in the circumstances, the closing price (in regular trading) for a share of Common Stock on the Nasdaq Capital Market (or, if the Common Stock is not then traded on the Nasdaq Capital Market, the principal established securities market on which shares of Common Stock are readily tradable) (the "Market") for the date in question or, if no sales of Common Stock were reported on the Market on that date, the closing price (in regular trading) for a share of Common Stock on the Market for the next preceding day on which sales of Common Stock were reported on the Market. The Administrator may, however, provide with respect to one or more awards that the fair market value shall equal the closing price (in regular trading) for a share of Common Stock on the Market on the last trading day preceding the date in question or the average of the high and low trading prices of a share of Common Stock on the Market for the date in question or the most recent trading day. If the Common Stock is no longer listed or is no longer actively traded on the Market as of the applicable date, the fair market value of the Common Stock shall be the value as reasonably determined by the Administrator for purposes of the award in the circumstances. The Administrator also may adopt a different methodology for determining fair market value with respect to one or more awards if a different methodology is necessary or advisable to secure any intended favorable tax, legal or other treatment for the particular award(s) (for example, and without limitation, the Administrator may provide that fair market value for purposes of one or more awards will be based on an average of closing prices (or the average of high and low daily trading prices) for a specified period preceding the relevant date).

5.6 Transfer Restrictions.

- **5.6.1** *Limitations on Exercise and Transfer.* Unless otherwise expressly provided in (or pursuant to) this Section 5.6 or required by applicable law: (a) all awards are non-transferable and shall not be subject in any manner to sale, transfer, anticipation, alienation, assignment, pledge, encumbrance or charge; (b) awards shall be exercised only by the participant; and (c) amounts payable or shares issuable pursuant to any award shall be delivered only to (or for the account of) the participant.
- **5.6.2** *Exceptions.* The Administrator may permit awards to be exercised by and paid to, or otherwise transferred to, other persons or entities pursuant to such conditions and procedures, including limitations on subsequent transfers, as the Administrator may, in its sole discretion, establish in writing. Any permitted transfer shall be subject to compliance with applicable federal and state securities laws and shall not be for value (other than nominal consideration, settlement of marital property rights, or for interests in an entity in which more than 50% of the voting interests are held by the Eligible Person or by the Eligible Person's family members).
- **5.6.3** *Further Exceptions to Limits on Transfer.* The exercise and transfer restrictions in Section 5.6.1 shall not apply to:
- (a) transfers to the Corporation (for example, in connection with the expiration or termination of the award);
- (b) the designation of a beneficiary to receive benefits in the event of the participant's death or, if the participant has died, transfers to or exercise by the participant's beneficiary, or, in the absence of a validly-designated beneficiary, transfers by will or the laws of descent and distribution;
- (c) subject to any applicable limitations on ISOs, transfers to a family member (or former family member) pursuant to a domestic relations order if received by the Administrator;
- (d) if the participant has suffered a disability, permitted transfers or exercises on behalf of the participant by his or her legal representative; or

- (e) the authorization by the Administrator of "cashless exercise" procedures with third parties who provide financing for the purpose of (or who otherwise facilitate) the exercise of awards consistent with applicable laws and any limitations imposed by the Administrator.
- 5.7 *International Awards*. One or more awards may be granted to Eligible Persons who provide services to the Corporation or one of its Subsidiaries outside of the United States. Any awards granted to such persons may be granted pursuant to the terms and conditions of any applicable sub-plans, if any, appended to this Plan and approved by the Administrator from time to time. The awards so granted need not comply with other specific terms of this Plan, provided that stockholder approval of any deviation from the specific terms of this Plan is not required by applicable law or the rules of any principal securities exchange or market on which shares of Common Stock are then traded.

6. EFFECT OF TERMINATION OF EMPLOYMENT OR SERVICE ON AWARDS

- **General.** The Administrator shall establish the effect (if any) of a termination of employment or service on the rights and benefits under each award under this Plan and in so doing may make distinctions based upon, inter alia, the cause of termination and type of award. If the participant is not an employee of the Corporation or one of its Subsidiaries, is not a member of the Board, and provides other services to the Corporation or one of its Subsidiaries, the Administrator shall be the sole judge for purposes of this Plan (unless a contract or the award otherwise provides) of whether the participant continues to render services to the Corporation or one of its Subsidiaries and the date, if any, upon which such services shall be deemed to have terminated.
- **Events Not Deemed Terminations of Employment.** Unless the express policy of the Corporation or one of its Subsidiaries, or the Administrator, otherwise provides, or except as otherwise required by applicable law, the employment relationship shall not be considered terminated in the case of: (a) sick leave, (b) military leave, or (c) any other leave of absence authorized by the Corporation or one of its Subsidiaries, or the Administrator; provided that, unless reemployment upon the expiration of such leave is guaranteed by contract or law or the Administrator otherwise provides, such leave is for a period of not more than three months. In the case of any employee of the Corporation or one of its Subsidiaries on an approved leave of absence, continued vesting of the award while on leave from the employ of the Corporation or one of its Subsidiaries may be suspended until the employee returns to service, unless the Administrator otherwise provides or applicable law otherwise requires. In no event shall an award be exercised after the expiration of any applicable maximum term of the award.
- **Effect of Change of Subsidiary Status.** For purposes of this Plan and any award, if an entity ceases to be a Subsidiary of the Corporation, a termination of employment or service shall be deemed to have occurred with respect to each Eligible Person in respect of such Subsidiary who does not continue as an Eligible Person in respect of the Corporation or another Subsidiary that continues as such after giving effect to the transaction or other event giving rise to the change in status unless the Subsidiary that is sold, spun-off or otherwise divested (or its successor or a direct or indirect parent of such Subsidiary or successor) assumes the Eligible Person's award(s) in connection with such transaction.

7. ADJUSTMENTS; ACCELERATION

7.1 Adjustments.

(a) Subject to Section 7.2, upon (or, as may be necessary to effect the adjustment, immediately prior to): any reclassification, recapitalization, stock split (including a stock split in the form of a stock dividend) or reverse stock split; any merger, combination, consolidation, conversion or other reorganization; any spin-off, split-up, or similar extraordinary dividend distribution in respect of the Common Stock; or any exchange of Common Stock or other securities of the Corporation, or any similar, unusual or extraordinary corporate transaction in respect of the Common Stock; then the Administrator shall equitably and

proportionately adjust: (i) the number and type of shares of Common Stock (or other securities) that thereafter may be made the subject of awards (including the specific share limits, maximums and numbers of shares set forth elsewhere in this Plan); (ii) the number, amount and type of shares of Common Stock (or other securities or property) subject to any outstanding awards; (iii) the grant, purchase, or exercise price (which term includes the base price of any SAR or similar right) of any outstanding awards; and/or (iv) the securities, cash or other property deliverable upon exercise or payment of any outstanding awards, in each case to the extent necessary to preserve (but not increase) the level of incentives intended by this Plan and the thenoutstanding awards.

- (b) Any adjustment in the number of shares subject to an award under this Section 7.1 shall be rounded down to the nearest whole share, although the Administrator in its sole discretion may make a cash payment in lieu of a fractional share.
- (c) Without limiting the generality of Section 3.4, any good faith determination by the Administrator as to whether an adjustment is required in the circumstances pursuant to this Section 7.1, and the extent and nature of any such adjustment, shall be conclusive and binding on all persons.

7.2 Corporate Transactions - Assumption and Termination of Awards.

Upon any event in which the Corporation does not survive, or does not survive as a public company in respect of its Common (a) Stock (including, without limitation, a dissolution, merger, combination, consolidation, conversion, exchange of securities, or other reorganization, or a sale of all or substantially all of the business, stock or assets of the Corporation, in any case in connection with which the Corporation does not survive or does not survive as a public company in respect of its Common Stock), then the Administrator may, in its sole discretion, make provision for a cash payment in settlement of, or for the termination, assumption, substitution or exchange of any or all outstanding awards or the cash, securities or property deliverable to the holder of any or all outstanding awards, based upon, to the extent relevant under the circumstances, the distribution or consideration payable to holders of the Common Stock upon or in respect of such event. Upon the occurrence of any event described in the preceding sentence in connection with which the Administrator has made provision for the award to be terminated (and the Administrator has not made a provision for the substitution, assumption, exchange or other continuation or settlement of the award): (i) unless otherwise provided in the applicable award agreement, each then-outstanding option and SAR shall become fully vested, all shares of restricted stock then outstanding shall fully vest free of restrictions, and each other award granted under this Plan that is then outstanding shall become payable to the holder of such award (with any performance goals applicable to the award in each case being deemed met, unless otherwise provided in the award agreement, at the "target" performance level); and (ii) each award (including any award or portion thereof that, by its terms, does not accelerate and vest in the circumstances) shall terminate upon the related event; provided that the holder of an option or SAR shall be given reasonable advance notice of the impending termination and a reasonable opportunity to exercise his or her outstanding vested options and SARs (after giving effect to any accelerated vesting required in the circumstances) in accordance with their terms before the termination of such awards (except that in no case shall more than ten days' notice of the impending termination be required and any acceleration of vesting and any exercise of any portion of an award that is so accelerated may be made contingent upon the actual occurrence of the event).

- (b) Without limiting the preceding paragraph, in connection with any event referred to in the preceding paragraph or any change in control event defined in any applicable award agreement, the Administrator may, in its discretion, provide for the accelerated vesting of any award or awards as and to the extent determined by the Administrator in the circumstances.
- For purposes of this Section 7.2, an award shall be deemed to have been "assumed" if (without limiting other circumstances in which an award is assumed) the award continues after an event referred to above in this Section 7.2, and/or is assumed and continued by the surviving entity following such event (including, without limitation, an entity that, as a result of such event, owns the Corporation or all or substantially all of the Corporation's assets directly or through one or more subsidiaries (a "Parent")), and confers the right to purchase or receive, as applicable and subject to vesting and the other terms and conditions of the award, for each share of Common Stock subject to the award immediately prior to the event, the consideration (whether cash, shares, or other securities or property) received in the event by the stockholders of the Corporation for each share of Common Stock sold or exchanged in such event (or the consideration received by a majority of the stockholders participating in such event if the stockholders were offered a choice of consideration); provided, however, that if the consideration offered for a share of Common Stock in the event is not solely the ordinary common stock of a successor corporation or a Parent, the Administrator may provide for the consideration to be received upon exercise or payment of the award, for each share subject to the award, to be solely ordinary common stock of the successor corporation or a Parent equal in fair market value to the per share consideration received by the stockholders participating in the event.
- (d) The Administrator may adopt such valuation methodologies for outstanding awards as it deems reasonable in the event of a cash or property settlement and, in the case of options, SARs or similar rights, but without limitation on other methodologies, may base such settlement solely upon the excess if any of the per share amount payable upon or in respect of such event over the exercise or base price of the award. In the case of an option, SAR or similar right as to which the per share amount payable upon or in respect of such event is less than or equal to the exercise or base price of the award, the Administrator may terminate such award in connection with an event referred to in this Section 7.2 without any payment in respect of such award.
- (e) In any of the events referred to in this Section 7.2, the Administrator may take such action contemplated by this Section 7.2 prior to such event (as opposed to on the occurrence of such event) to the extent that the Administrator deems the action necessary to permit the participant to realize the benefits intended to be conveyed with respect to the underlying shares. Without limiting the generality of the foregoing, the Administrator may deem an acceleration and/or termination to occur immediately prior to the applicable event and, in such circumstances, will reinstate the original terms of the award if an event giving rise to an acceleration and/or termination does not occur.
- (f) Without limiting the generality of Section 3.4, any good faith determination by the Administrator pursuant to its authority under this Section 7.2 shall be conclusive and binding on all persons.
- (g) The Administrator may override the provisions of this Section 7.2 by express provision in the award agreement and may accord any participant a right to refuse any acceleration, whether pursuant to the award agreement or otherwise, in such circumstances as the Administrator may approve. The portion of any ISO accelerated in connection with an event referred to in this Section 7.2 (or such other circumstances as may trigger accelerated vesting of the award) shall remain exercisable as an ISO only to the extent the applicable \$100,000 limitation on ISOs is not exceeded. To the extent exceeded, the accelerated portion of the option shall be exercisable as a nonqualified stock option under the Code.

8. OTHER PROVISIONS

- **Compliance with Laws**. This Plan, the granting and vesting of awards under this Plan, the offer, issuance and delivery of shares of Common Stock, and/or the payment of money under this Plan or under awards are subject to compliance with all applicable federal, state, local and foreign laws, rules and regulations (including, but not limited to, state and federal securities law) and to such approvals by any exchange, regulatory or governmental authority as may, in the opinion of counsel for the Corporation, be necessary or advisable in connection therewith. The person acquiring any securities under this Plan will, if requested by the Corporation or one of its Subsidiaries, provide such assurances and representations to the Corporation or one of its Subsidiaries as the Administrator may deem necessary or desirable to assure compliance with all applicable legal and accounting requirements.
- **8.2 No Rights to Award.** No person shall have any claim or rights to be granted an award (or additional awards, as the case may be) under this Plan, subject to any express contractual rights (set forth in a document other than this Plan) to the contrary.
- **8.3 No Employment/Service Contract.** Nothing contained in this Plan (or in any other documents under this Plan or in any award) shall confer upon any Eligible Person or other participant any right to continue in the employ or other service of the Corporation or one of its Subsidiaries, constitute any contract or agreement of employment or other service or affect an employee's status as an employee at will, nor shall interfere in any way with the right of the Corporation or one of its Subsidiaries to change a person's compensation or other benefits, or to terminate his or her employment or other service, with or without cause. Nothing in this Section 8.3, however, is intended to adversely affect any express independent right of such person under a separate employment or service contract other than an award agreement.
- **Plan Not Funded.** Awards payable under this Plan shall be payable in shares or from the general assets of the Corporation, and no special or separate reserve, fund or deposit shall be made to assure payment of such awards. No participant, beneficiary or other person shall have any right, title or interest in any fund or in any specific asset (including shares of Common Stock, except as expressly otherwise provided) of the Corporation or one of its Subsidiaries by reason of any award hereunder. Neither the provisions of this Plan (or of any related documents), nor the creation or adoption of this Plan, nor any action taken pursuant to the provisions of this Plan shall create, or be construed to create, a trust of any kind or a fiduciary relationship between the Corporation or one of its Subsidiaries and any participant, beneficiary or other person. To the extent that a participant, beneficiary or other person acquires a right to receive payment pursuant to any award hereunder, such right shall be no greater than the right of any unsecured general creditor of the Corporation.
- **8.5 Tax Withholding.** Upon any exercise, vesting, or payment of any award, or upon any other tax withholding event with respect to any award, arrangements satisfactory to the Corporation shall be made to provide for any taxes the Corporation or any of its Subsidiaries may be required or permitted to withhold with respect to such award event or payment. Such arrangements may include (but are not limited to) any one of (or a combination of) the following:
 - (a) The Corporation or one of its Subsidiaries shall have the right to require the participant (or the participant's personal representative or beneficiary, as the case may be) to pay or provide for payment of the amount of any taxes which the Corporation or one of its Subsidiaries may be required or permitted to withhold with respect to such award event or payment.
 - (b) The Corporation or one of its Subsidiaries shall have the right to deduct from any amount otherwise payable in cash (whether related to the award or otherwise) to the participant (or the participant's personal representative or beneficiary, as the case may be) the amount of any taxes which the Corporation or one of its Subsidiaries may be required or permitted to withhold with respect to such award event or payment.

- (c) In any case where a tax is required to be withheld in connection with the delivery of shares of Common Stock under this Plan, the Administrator may in its sole discretion (subject to Section 8.1) require or grant (either at the time of the award or thereafter) to the participant the right to elect, pursuant to such rules and subject to such conditions as the Administrator may establish, that the Corporation reduce the number of shares to be delivered by (or otherwise reacquire) the appropriate number of shares, valued in a consistent manner at their fair market value or at the sales price in accordance with authorized procedures for cashless exercises, necessary to satisfy any applicable withholding obligation on exercise, vesting or payment.
- 8.6 Section 409A. Except as otherwise expressly set forth in an award agreement, it is intended that awards granted under the Plan either be exempt from, or comply with, the requirements of Section 409A of the Code ("Section 409A"). To the extent an award is subject to Section 409A (a "409A Award"), the terms of the Plan, the 409A Award and any award agreement shall be interpreted to comply with the requirements of Section 409A so that the 409A Award is not subject to additional tax or interest under Section 409A, unless the Administrator expressly provides otherwise. A 409A Award shall be subject to such additional rules and requirements as specified by the Administrator from time to time in order for it to comply with the requirements of Section 409A. In this regard, if any amount under a 409A Award is payable upon a "separation from service" to an individual who is considered a "specified employee" (as each term is defined under Section 409A), then no such payment shall be made prior to the date that is the earlier of (a) six months and one day after the participant's separation from service or (b) the participant's death, but only to the extent such delay is necessary for such payment to comply with the requirements of Section 409A(a)(2)(B)(i) of the Code. Each participant is solely responsible and liable for the satisfaction of all taxes and penalties that may be imposed on or in respect of such participant in connection with this Plan, including any taxes and penalties under Section 409A, and neither the Corporation, nor any Subsidiary or any affiliate shall have any obligation to indemnify or otherwise hold such participant or any beneficiary harmless from such taxes or penalties. With respect to any 409A Award, references to "termination of employment" (and substantially similar phrases) shall mean "separation from service" within the meaning of Section 409A. For purposes of Section 409A, each of the payments that may be made in respect of any award granted under the Plan is designated as a separate payment.

8.7 Effective Date, Termination and Suspension, Amendments.

- **8.7.1** Effective Date. This Plan is effective as of February 3, 2022, the date of its approval by the Board (the "Effective Date"). This Plan shall be submitted for and subject to stockholder approval no later than twelve months after the Effective Date. Unless earlier terminated by the Board and subject to any extension that may be approved by stockholders, this Plan shall terminate at the close of business on the day before the tenth anniversary of the Effective Date. After the termination of this Plan either upon such stated termination date or its earlier termination by the Board, no additional awards may be granted under this Plan, but previously granted awards (and the authority of the Administrator with respect thereto, including the authority to amend such awards) shall remain outstanding in accordance with their applicable terms and conditions and the terms and conditions of this Plan.
- **8.7.2** *Board Authorization.* The Board may, at any time, terminate or, from time to time, amend, modify or suspend this Plan, in whole or in part. No awards may be granted during any period that the Board suspends this Plan.
- **8.7.3** *Stockholder Approval*. To the extent then required by applicable law or deemed necessary or advisable by the Board, any amendment to this Plan shall be subject to stockholder approval.
- **8.7.4** *Amendments to Awards.* Without limiting any other express authority of the Administrator under (but subject to) the express limits of this Plan, the Administrator by agreement or resolution may waive conditions of or limitations on awards to participants that the Administrator in the prior

exercise of its discretion has imposed, without the consent of a participant, and (subject to the requirements of Sections 3.2 and 8.6.5) may make other changes to the terms and conditions of awards. Any amendment or other action that would constitute a repricing of an award is subject to the no-repricing provision of Section 3.3.

- **8.7.5** *Limitations on Amendments to Plan and Awards*. No amendment, suspension or termination of this Plan or amendment of any outstanding award agreement shall, without written consent of the participant, affect in any manner materially adverse to the participant any rights or benefits of the participant or obligations of the Corporation under any award granted under this Plan prior to the effective date of such change. Changes, settlements and other actions contemplated by Section 7 shall not be deemed to constitute changes or amendments for purposes of this Section 8.7.
- **Privileges of Stock Ownership.** Except as otherwise expressly authorized by the Administrator, a participant shall not be entitled to any privilege of stock ownership as to any shares of Common Stock not actually delivered to and held of record by the participant. Except as expressly required by Section 7.1 or otherwise expressly provided by the Administrator, no adjustment will be made for dividends or other rights as a stockholder for which a record date is prior to such date of delivery.
- 8.9 Governing Law; Severability.
 - **8.9.1** *Choice of Law.* This Plan, the awards, all documents evidencing awards and all other related documents shall be governed by, and construed in accordance with the laws of the State of Delaware, notwithstanding any Delaware or other conflict of law provision to the contrary.
 - **8.9.2** *Severability.* If a court of competent jurisdiction holds any provision invalid and unenforceable, the remaining provisions of this Plan shall continue in effect.
- **8.10** *Captions*. Captions and headings are given to the sections and subsections of this Plan solely as a convenience to facilitate reference. Such headings shall not be deemed in any way material or relevant to the construction or interpretation of this Plan or any provision thereof.
- 8.11 Stock-Based Awards in Substitution for Stock Options or Awards Granted by Other Corporation. Awards may be granted to Eligible Persons in substitution for or in connection with an assumption of employee stock options, SARs, restricted stock or other stock-based awards granted by other entities to persons who are or who will become Eligible Persons in respect of the Corporation or one of its Subsidiaries, in connection with a distribution, merger or other reorganization by or with the granting entity or an affiliated entity, or the acquisition by the Corporation or one of its Subsidiaries, directly or indirectly, of all or a substantial part of the stock or assets of the employing entity. The awards so granted need not comply with other specific terms of this Plan, provided the awards reflect adjustments giving effect to the assumption or substitution consistent with any conversion applicable to the common stock (or the securities otherwise subject to the award) in the transaction and any change in the issuer of the security. Any shares that are delivered and any awards that are granted by, or become obligations of, the Corporation, as a result of the assumption by the Corporation of, or in substitution for, outstanding awards previously granted or assumed by an acquired company (or previously granted or assumed by a predecessor employer (or direct or indirect parent thereof) in the case of persons that become employed by the Corporation or one of its Subsidiaries in connection with a business or asset acquisition or similar transaction) shall not be counted against the Share Limit or other limits on the number of shares available for issuance under this Plan and shares subject to any substitute awards shall not be available for awards under the Plan in the event of any forfeiture, expiration or cash settlement of such substitute awards.
- **8.12 Non-Exclusivity of Plan.** Nothing in this Plan shall limit or be deemed to limit the authority of the Board, the Administrator, the Corporation, or any Subsidiary to grant awards or authorize any other compensation, with or without reference to the Common Stock, under any other plan or authority.

- 8.13 No Corporate Action Restriction. The existence of this Plan, the award agreements and the awards granted hereunder shall not limit, affect, or restrict in any way the right or power of the Corporation or any Subsidiary (or any of their respective shareholders, boards of directors or committees thereof (or any subcommittees), as the case may be) to make or authorize: (a) any adjustment, recapitalization, reorganization or other change in the capital structure or business of the Corporation or any Subsidiary, (b) any merger, amalgamation, consolidation or change in the ownership of the Corporation or any Subsidiary, (c) any issue of bonds, debentures, capital, preferred or prior preference stock ahead of or affecting the capital stock (or the rights thereof) of the Corporation or any Subsidiary, (d) any dissolution or liquidation of the Corporation or any Subsidiary, (e) any sale or transfer of all or any part of the assets or business of the Corporation or any Subsidiary, (f) any other award, grant, or payment of incentives or other compensation under any other plan or authority (or any other action with respect to any benefit, incentive or compensation), or (g) any other corporate act or proceeding by the Corporation or any Subsidiary. No participant, beneficiary or any other person shall have any claim under any award or award agreement against any member of the Board or the Administrator, or the Corporation or any employees, officers or agents of the Corporation or any Subsidiary, as a result of any such action. Awards need not be structured so as to be deductible for tax purposes.
- **8.14 Other Company Benefit and Compensation Programs.** Payments and other benefits received by a participant under an award made pursuant to this Plan are wholly discretionary and, although provided by the Corporation, do not constitute regular or periodic payments. Unless otherwise required by applicable laws, the payments and other benefits provided under the Plan shall not be deemed a part of a participant's compensation for purposes of the determination of benefits under any other employee welfare or benefit plans or arrangements, if any, provided by the Corporation or any Subsidiary, except where the Administrator expressly otherwise provides or authorizes in writing. Awards under this Plan may be made in addition to, in combination with, as alternatives to or in payment of grants, awards or commitments under any other plans, arrangements or authority of the Corporation or its Subsidiaries.
- **8.15 Clawback Policy.** The awards granted under this Plan are subject to the terms of the Corporation's recoupment, clawback or similar policy as it may be in effect from time to time, as well as any similar provisions of applicable law, any of which could in certain circumstances require repayment or forfeiture of awards or any shares of Common Stock or other cash or property received with respect to the awards (including any value received from a disposition of the shares acquired upon payment of the awards).

CEPTON, INC. EMPLOYEE STOCK PURCHASE PLAN

1. **PURPOSE**

The purpose of this Plan is to assist Eligible Employees in acquiring a stock ownership interest in the Corporation, at a favorable price and upon favorable terms, pursuant to a plan which is intended to qualify as an "employee stock purchase plan" under Section 423 of the Code. This Plan is also intended to encourage Eligible Employees to remain in the employ of the Corporation (or a Subsidiary which may be designated by the Committee as a "Participating Subsidiary") and to provide them with an additional incentive to advance the best interests of the Corporation.

2. **DEFINITIONS**

Capitalized terms used herein which are not otherwise defined shall have the following meanings.

- "Account" means the bookkeeping account maintained by the Corporation, or by a recordkeeper on behalf of the Corporation, for a Participant pursuant to Section 7(a).
- "Board" means the Board of Directors of the Corporation.
- "Code" means the U.S. Internal Revenue Code of 1986, as amended from time to time.
- "Commission" means the U.S. Securities and Exchange Commission.
- "Committee" means the committee appointed by the Board to administer this Plan pursuant to Section 12.
- "Common Stock" means the common stock, par value \$0.00001 per share, of the Corporation, and such other securities or property as may become the subject of Options pursuant to an adjustment made under Section 17.
- "Compensation" means an Eligible Employee's regular gross pay. Compensation includes any amounts contributed as salary reduction contributions to a plan qualifying under Section 401(k), 125 or 129 of the Code. Any other form of remuneration is excluded from Compensation, including (but not limited to) the following: severance pay, overtime payments, commissions, prizes, awards, relocation or housing allowances, stock option exercises, stock appreciation right payments, the vesting or grant of restricted stock, the payment of stock units, performance awards, auto allowances, tuition reimbursement, perquisites, non-cash compensation and other forms of imputed income, bonuses, incentive compensation, special payments, fees and allowances. Notwithstanding the foregoing, Compensation shall not include any amounts deferred under or paid from any nonqualified deferred compensation plan maintained by the Corporation or any Subsidiary.
- "Contributions" means all bookkeeping amounts credited to the Account of a Participant pursuant to Section 7(a).
- "Corporation" means Cepton, Inc., a Delaware corporation, and its successors.
- "Effective Date" means February 3, 2022, the date this Plan was adopted by the Board.
- "Eligible Employee" means any employee of the Corporation, or of any Subsidiary which has been designated in writing by the Committee as a "Participating Subsidiary" (including any Subsidiaries which have become such after the date that this Plan is approved by the stockholders of the Corporation). Notwithstanding the foregoing and unless otherwise provided by the Committee in advance of the applicable Offering Period, "Eligible Employee" shall not include any employee:
- (a) whose customary employment is for not more than five (5) months in a calendar year; or
- (b) whose customary employment is for twenty (20) hours or less per week.

- "Exchange Act" means the U.S. Securities Exchange Act of 1934, as amended from time to time.
- "Exercise Date" means, with respect to an Offering Period, the last day of that Offering Period.
- "Fair Market Value" on any date means:
- (a) if the Common Stock is listed or admitted to trade on a national securities exchange, the closing price of a share of Common Stock on such date on the principal national securities exchange on which the Common Stock is so listed or admitted to trade, or, if there is no trading of the Common Stock on such date, then the closing price of a share of Common Stock on such exchange on the next preceding date on which there was trading in the shares of Common Stock;
- (b) in the absence of exchange data required to determine Fair Market Value pursuant to the foregoing, the value as established by the Committee as of the relevant time for purposes of this Plan.
- "Grant Date" means the first day of each Offering Period, as determined by the Committee and announced to potential Eligible Employees.
- "Individual Limit" has the meaning given to such term in Section 4(b).
- "New Exercise Date" has the meaning given to such term in Section 18.
- "Offering Period" means the period of six (6) consecutive months commencing on each Grant Date; provided, however, that the Committee may declare, as it deems appropriate and in advance of the applicable Offering Period, a shorter (not to be less than three months) Offering Period or a longer (not to exceed 27 months) Offering Period; provided, further, that the Committee may provide, as it deems appropriate and in advance of the applicable Offering Period, that such Offering Period will consist of multiple "purchase periods," with an Exercise Date to occur at the end of each such purchase period. In no event will the Grant Date for an Offering Period occur on or before the Exercise Date (or the final Exercise Date, as the case may be) for the immediately preceding Offering Period.
- "Option" means the stock option to acquire shares of Common Stock granted to a Participant pursuant to Section 8.
- "Option Price" means the per share exercise price of an Option as determined in accordance with Section 8(b).
- "Parent" means any corporation (other than the Corporation) in an unbroken chain of corporations ending with the Corporation in which each corporation (other than the Corporation) owns stock possessing 50% or more of the total combined voting power of all classes of stock in one or more of the other corporations in the chain.
- "Participant" means an Eligible Employee who has elected to participate in this Plan and who has filed a valid and effective Subscription Agreement to make Contributions pursuant to Section 6.
- "Participating Subsidiary" means any Subsidiary that has been designated in writing by the Committee as a Participating Subsidiary for purposes of this Plan.
- "Plan" means this Cepton, Inc. Employee Stock Purchase Plan, as amended from time to time.
- "Rule 16b-3" means Rule 16b-3 as promulgated by the Commission under Section 16, as amended from time to time.
- "Share Limit" has the meaning given to such term in Section 4(a).
- "Subscription Agreement" means the written enrollment agreement or applicable electronic form of enrollment agreement filed by an Eligible Employee with the Corporation (or its designee) pursuant to Section 6 to participate in this Plan.

"Subsidiary" means any corporation (other than the Corporation) in an unbroken chain of corporations (beginning with the Corporation) in which each corporation (other than the last corporation) owns stock possessing 50% or more of the total combined voting power of all classes of stock in one or more of the other corporations in the chain.

3. ELIGIBILITY

Any person employed as an Eligible Employee as of a Grant Date shall be eligible to participate in this Plan during the Offering Period in which such Grant Date occurs, subject to the Eligible Employee satisfying the requirements of Section 6.

STOCK SUBJECT TO THIS PLAN; SHARE LIMITATIONS

- (a) Aggregate Share Limit.
 - (i) Subject to the provisions of Section 17, the capital stock that may be delivered under this Plan will be shares of the Corporation's authorized but unissued Common Stock and any of its shares of Common Stock held as treasury shares. The maximum number of shares of Common Stock that may be delivered pursuant to Options granted under this Plan is 3,080,960 shares, subject to adjustments pursuant to Section 17 (the "Share Limit").
 - (ii) In addition, subject to adjustments pursuant to Section 17, the Share Limit shall automatically increase on the first trading day in January of each of the calendar years during the term of this Plan, with the first such increase to occur in January 2023, by an amount equal to the lesser of (A) one percent (1%) of the total number of shares of Common Stock issued and outstanding on December 31 of the immediately preceding calendar year, (B) 3,000,000 shares of Common Stock, or (C) such number of shares of Common Stock as may be established by the Board.
 - (iii) In the event that during a particular Offering Period all of the shares of Common Stock made available under this Plan are subscribed prior to the expiration of this Plan, this Plan and all outstanding Options hereunder shall terminate at the end of that Offering Period and the shares available shall be allocated for purchase by Participants in that Offering Period on a pro-rata basis determined with respect to Participants' Account balances.
- (b) Individual Share Limit. The maximum number of shares of Common Stock that any one individual may acquire upon exercise of his or her Option with respect to any one Offering Period is 5,000 shares, subject to adjustments pursuant to Section 17 (the "Individual Limit"); provided, however, that the Committee may amend such Individual Limit, effective no earlier than the first Offering Period commencing after the adoption of such amendment, without stockholder approval. The Individual Limit shall be proportionately adjusted for any Offering Period of less than six months, and may, at the discretion of the Committee, be proportionately increased for any Offering Period of greater than six months.
- (c) Shares Not Actually Delivered. Shares that are subject to or underlie Options, which for any reason are cancelled or terminated, are forfeited, fail to vest, or for any other reason are not paid or delivered under this Plan shall again, except to the extent prohibited by law, be available for subsequent Options under this Plan.

5. **OFFERING PERIODS**

During the term of this Plan, the Corporation will offer Options to purchase shares in each Offering Period to all Eligible Employees in that Offering Period. Unless otherwise specified by the Committee in advance of the Offering Period, Offering Periods will be of six (6) months duration. The Committee will specify in

advance of each Offering Period when the Offering Period will commence and the Grant Date of the Offering Period. Each Option shall become effective on the Grant Date of that Offering Period. The term of each Option shall be the duration of the related Offering Period and shall end on the Exercise Date of that Offering Period. Offering Periods shall continue until this Plan is terminated in accordance with Section 18 or 19, or, if earlier, until no shares remain available for Options pursuant to Section 4.

6. **PARTICIPATION**

- (a) Enrollment. An Eligible Employee may become a Participant in this Plan by completing a Subscription Agreement on a form approved by and in a manner prescribed by the Committee (or its delegate). To become effective, a Subscription Agreement must be signed by the Eligible Employee and filed with the Corporation (or its designee) at the time specified by the Committee, but in all cases prior to the start of the Offering Period with respect to which it is to become effective, and must set forth a whole percentage (or, if the Committee so provides, a stated amount) of the Eligible Employee's Compensation to be credited to the Participant's Account as Contributions each pay period.
- (b) Contribution Limits. Notwithstanding the foregoing, a Participant's Contribution election shall be subject to the following limitations:
 - (i) the \$25,000 annual limitation set forth in Section 8(c);
 - (ii) a Participant may not elect to contribute more than fifteen percent (15%) of his or her Compensation each pay period as Plan Contributions; and
 - (iii) such other limits, rules, or procedures as the Committee may prescribe.
- (c) Content and Duration of Subscription Agreements. Subscription Agreements shall contain the Eligible Employee's authorization and consent to the Corporation's withholding from his or her Compensation the amount of his or her Contributions. An Eligible Employee's Subscription Agreement, and his or her participation election and withholding consent thereon, shall remain valid for all Offering Periods until (i) the Eligible Employee's participation terminates pursuant to the terms hereof, (ii) the Eligible Employee files a new Subscription Agreement that becomes effective, or (iii) the Committee requires that a new Subscription Agreement be executed and filed with the Corporation.

7. METHOD OF PAYMENT OF CONTRIBUTIONS

- (a) *Participation Account*. The Corporation shall maintain on its books, or cause to be maintained by a recordkeeper, an Account in the name of each Participant. The percentage (or amount, as applicable) of Compensation elected to be applied as Contributions by a Participant shall be deducted from such Participant's Compensation on each payday during the period for payroll deductions set forth below and such payroll deductions shall be credited to that Participant's Account as soon as administratively practicable after such date. A Participant may not make any additional payments to his or her Account. A Participant's Account shall be reduced by any amounts used to pay the Option Price of shares acquired, or by any other amounts distributed pursuant to the terms hereof.
- (b) *Commencement of Payroll Deductions.* Payroll deductions with respect to an Offering Period shall commence as of the first day of the payroll period which coincides with or immediately follows the applicable Grant Date and shall end on the last day of the payroll period which coincides with or immediately precedes the applicable Exercise Date, unless sooner terminated by the Participant as provided in this Section 7 or until his or her Plan participation terminates pursuant to Section 11.
- (c) Withdrawal During an Offering Period. A Participant may terminate his or her Contributions during an Offering Period (and receive a distribution of the balance of his or her Account in accordance with Section 11) by completing and filing with the Corporation (or its designee), in such form and on such terms as the Committee (or its delegate) may prescribe, a written withdrawal form or applicable

electronic withdrawal form which shall be completed by the Participant. Such termination shall be effective as soon as administratively practicable after its receipt by the Corporation. A withdrawal election pursuant to this Section 7(c) with respect to an Offering Period shall only be effective, however, if it is received by the Corporation prior to the Exercise Date of that Offering Period (or such earlier deadline that the Committee may reasonably require to process the withdrawal prior to the Exercise Date). Partial withdrawals of Accounts, and other modifications or suspensions of Subscription Agreements, except as provided in Section 7(d) or 7(e), are not permitted.

- (d) Change in Contribution Elections for the Following Offering Period. A Participant may discontinue, increase, or decrease the level of his or her Contributions (within Plan limits) by completing and filing with the Corporation (or its designee), on such terms as the Committee (or its delegate) may prescribe, a new Subscription Agreement which indicates such election. Subject to any additional timing requirements that the Committee may impose, an election pursuant to this Section 7(d) shall be effective with the first Offering Period that commences after the Corporation's receipt of such election.
- (e) Discontinuing Contributions During an Offering Period. A Participant may discontinue his or her Contributions (but not increase or otherwise decrease the level of his or her Contributions) during an Offering Period, by filing with the Corporation (or its designee), on such terms as the Committee (or its delegate) may prescribe, a new Subscription Agreement which indicates such election. An election pursuant to this Section 7(e) shall be effective no earlier than the first payroll period that starts after the Corporation's receipt of such election. If a Participant elects to discontinue his or her Contributions pursuant to this Section 7(e), the Contributions previously credited to the Participant's Account for that Offering Period shall be used to exercise the Participant's Option as of the applicable Exercise Date in accordance with Section 9 (unless the Participant makes a timely withdrawal election in accordance with Section 7(c), in which case the Participant's Account will be paid to him or her in cash in accordance with Section 11(a)).

8. GRANT OF OPTION

- (a) *Grant Date; Number of Shares.* On each Grant Date, each Eligible Employee who is a Participant during that Offering Period shall be granted an Option to purchase a number of shares of Common Stock. The Option shall be exercised on the Exercise Date. The number of shares subject to the Option shall be determined by dividing the Participant's Account balance as of the applicable Exercise Date by the Option Price.
- (b) Option Price. The Option Price per share of the shares subject to an Option for an Offering Period shall be the <u>lesser</u> of: (i) 85% of the Fair Market Value of a share of Common Stock on the Exercise Date of that Offering Period; provided, however, that the Committee may provide prior to the start of any Offering Period that the Option Price for that Offering Period shall be determined by applying a discount amount (not to exceed 15%) to either (1) the Fair Market Value of a share of Common Stock on the Grant Date of that Offering Period, or (2) the Fair Market Value of a share of Common Stock on the Exercise Date of that Offering Period, or (3) the lesser of the Fair Market Value of a share of Common Stock on the Grant Date of that Offering Period or the Fair Market Value of a share of Common Stock on the Exercise Date of that Offering Period (or, for purposes of the foregoing clauses (2) and (3), the applicable Exercise Date of that Offering Period, as the case may be). Notwithstanding anything to the contrary in the preceding provisions of this Section 8(b), in no event shall the Option Price per share be less than the par value of a share of Common Stock.
- (c) *Limits on Share Purchases*. Notwithstanding anything else contained herein, a person who is otherwise an Eligible Employee shall not be granted any Option (or any Option granted shall be subject to compliance with the following limitations) or other right to purchase shares under this Plan to the extent:
 - (1) it would, if exercised, cause the person to own stock (within the meaning of Section 423(b)(3) of the Code) possessing 5% or more of the total combined voting power or value of all classes of stock of the Corporation, or of any Parent, or of any Subsidiary; or

(2) such Option causes such individual to have rights to purchase stock under this Plan and any other plan of the Corporation, any Parent, or any Subsidiary which is qualified under Section 423 of the Code which accrue at a rate which exceeds \$25,000 of the fair market value of the stock of the Corporation, of any Parent, or of any Subsidiary (determined at the time the right to purchase such stock is granted, before giving effect to any discounted purchase price under any such plan) for each calendar year in which such right is outstanding at any time.

For purposes of the foregoing, a right to purchase stock accrues when it first becomes exercisable during the calendar year. In determining whether the stock ownership of an Eligible Employee equals or exceeds the 5% limit set forth above, the rules of Section 424(d) of the Code (relating to attribution of stock ownership) shall apply, and stock which the Eligible Employee may purchase under outstanding options shall be treated as stock owned by the Eligible Employee.

9. EXERCISE OF OPTION

- (a) Unless a Participant withdraws from an Offering Period pursuant to Section 7(c) or the Participant's Plan participation is terminated as provided in Section 11, his or her Option for the purchase of shares shall be exercised automatically on the Exercise Date for that Offering Period, without any further action on the Participant's part, and the maximum number of whole shares subject to such Option (subject to the Individual Limit set forth in Section 4(b) and the limitations contained in Section 8(c)) shall be purchased at the Option Price with the balance of such Participant's Account.
- (b) If any amount is not sufficient to purchase a whole share of Common Stock remains in a Participant's Account after the exercise of his or her Option on the Exercise Date, such amount shall be refunded to such Participant as soon as administratively practicable after such date; provided that the Committee may provide in advance of an Offering Period for any such amount with respect to that Offering Period to be credited to the Participant's Account for the next Offering Period, if he or she is a Participant in such next Offering Period.
- (c) If the Share Limit of Section 4(a) is reached, any amount that remains in a Participant's Account after the exercise of his or her Option on the Exercise Date to purchase the number of shares that he or she is allocated shall be refunded to the Participant as soon as administratively practicable after such date.
- (d) If any amount which exceeds the Individual Limit set forth in Section 4(b) or one of the limitations set forth in Section 8(c) remains in a Participant's Account after the exercise of his or her Option on the Exercise Date, such amount shall be refunded to the Participant as soon as administratively practicable after such date.

10. **DELIVERY OF SHARES**

As soon as administratively practicable after the Exercise Date, the Corporation shall, in its discretion, either deliver to each Participant a certificate representing the shares of Common Stock purchased upon exercise of his or her Option, provide for the crediting of such shares in book entry form in the name of the Participant, or provide for an alternative arrangement for the delivery of such shares to a broker or recordkeeping service for the benefit of the Participant. In the event the Corporation is required to obtain from any commission or agency authority to issue any such certificate or otherwise deliver such shares, the Corporation will seek to obtain such authority. If the Corporation is unable to obtain from any such commission or agency authority which counsel for the Corporation deems necessary for the lawful issuance of any such certificate or other delivery of such shares, or if for any other reason the Corporation cannot issue or deliver shares of Common Stock and satisfy Section 21, the Corporation shall be relieved from liability to any Participant except that the Corporation shall return to each Participant to whom such shares cannot be issued or delivered the amount of the balance credited to his or her Account that would have otherwise been used for the purchase of such shares.

11. TERMINATION OF EMPLOYMENT; CHANGE IN ELIGIBLE STATUS

- (a) *General*. Except as provided in the next paragraph, if a Participant ceases to be an Eligible Employee for any reason at any time prior to the last day of an Offering Period in which he or she participates, or if the Participant timely elects to terminate Contributions and withdraw from the Plan pursuant to Section 7(c), such Participant's Account shall be paid to him or her (or, in the event of the Participant's death, to the person or persons entitled thereto under Section 13) in cash, and such Participant's Option and participation in the Plan shall be automatically terminated.
- (b) Change in Employee Status or Leave of Absence. If a Participant (i) ceases to be an Eligible Employee during an Offering Period but remains an employee of the Corporation or a Participating Subsidiary through the Exercise Date, or (ii) during an Offering Period commences a sick leave, military leave, or other leave of absence approved by the Corporation or a Participating Subsidiary, and the leave meets the requirements of Treasury Regulation Section 1.421-1(h)(2) and the Participant is an employee of the Corporation or a Participating Subsidiary or on such leave as of the applicable Exercise Date, such Participant's Contributions shall cease, and the Contributions previously credited to the Participant's Account for that Offering Period shall be used to exercise the Participant's Option as of the applicable Exercise Date in accordance with Section 9 (unless the Participant makes a timely election to terminate Contributions and withdraw from the Plan in accordance with Section 7(c), in which case such Participant's Account shall be paid to him or her in cash in accordance with the foregoing paragraph).
- (c) Re-Enrollment. A Participant's termination from Plan participation precludes the Participant from again participating in this Plan during that Offering Period. However, such termination shall not have any effect upon his or her ability to participate in any succeeding Offering Period, provided that the applicable eligibility and participation requirements are again then met. A Participant's termination from Plan participation shall be deemed to be a revocation of that Participant's Subscription Agreement and such Participant must file a new Subscription Agreement to resume Plan participation in any succeeding Offering Period.
- (d) *Change in Subsidiary Status.* For purposes of this Plan, if a Participating Subsidiary ceases to be a Subsidiary, each person employed by that Subsidiary will be deemed to have terminated employment for purposes of this Plan and will no longer be an Eligible Employee, unless the person continues as an Eligible Employee in respect of the Corporation or another Participating Subsidiary.

12. **ADMINISTRATION**

- (a) The Committee. The Board shall appoint the Committee, which shall be composed of not less than two members of the Board. Subject to the preceding sentence, the Board may, at any time, increase or decrease the number of members of the Committee, may remove from membership on the Committee all or any portion of its members, and may appoint such person or persons as it desires to fill any vacancy existing on the Committee, whether caused by removal, resignation, or otherwise. The Board may also, at any time, assume or change the administration of this Plan.
- (b) Powers and Duties of the Committee. The Committee shall administer this Plan and shall have full power and discretion to adopt, amend and rescind any rules it considers desirable and appropriate for the administration of this Plan and not inconsistent with the terms of this Plan (including, without limitation, rules and deadlines for making elections under the Plan, which deadlines may be more restrictive than the deadlines otherwise set forth in this Plan), to further define the terms used in this Plan, and to make all other determinations necessary or advisable for the administration of this Plan or the effectuation of its purposes. The Committee shall act by majority vote or by unanimous written consent. No member of the Committee shall be entitled to act on or decide any matter relating solely to himself or herself or solely to any of his or her rights or benefits under this Plan. The Committee shall have full power and discretionary authority to construe and interpret the terms and conditions of this Plan and any agreements defining the rights and obligations of the Corporation, any Subsidiary, and any Participant or other person under this Plan, which construction

or interpretation shall be final and binding on all parties including the Corporation, Subsidiaries, Participants and beneficiaries. Notwithstanding anything else contained in this Plan to the contrary, the Committee may also adopt rules, procedures, separate offerings, or sub-plans applicable to particular Subsidiaries or locations, which separate offerings or sub-plans may be designed to be outside the scope of Section 423 of the Code and need not comply with the otherwise applicable provisions of this Plan. The Committee may delegate ministerial non-discretionary functions to third parties, including individuals who are officers or employees of the Corporation or Participating Subsidiaries.

- (c) Decisions of the Committee are Binding; Reliance on Experts. Subject only to compliance with the express provisions hereof, the Board and Committee may act in their absolute discretion in matters within their authority related to this Plan. Any action taken by, or inaction of, the Corporation, any Participating Subsidiary, the Board or the Committee relating or pursuant to this Plan and within its authority hereunder or under applicable law shall be within the absolute discretion of that entity or body and shall be conclusive and binding upon all persons. In making any determination or in taking or not taking any action under this Plan, the Board or Committee, as the case may be, may obtain and may rely on the advice of experts, including professional advisors to the Corporation. No member of the Board or Committee, or officer or agent of the Corporation, will be liable for any action, omission or decision under the Plan taken, made or omitted in good faith.
- (d) *Indemnification.* Neither the Board nor any Committee, nor any member thereof or person acting at the direction thereof, shall be liable for any act, omission, interpretation, construction or determination made in good faith in connection with this Plan, and all such persons shall be entitled to indemnification and reimbursement by the Corporation in respect of any claim, loss, damage or expense (including, without limitation, attorneys' fees) arising or resulting therefrom to the fullest extent permitted by law and/or under any directors and officers liability insurance coverage that may be in effect from time to time.

13. **DEATH BENEFITS**

In the event of the death of a Participant, the Corporation shall deliver such shares and/or cash payable pursuant to the terms hereof to the executor or administrator of the estate of the Participant, or if no such executor or administrator has been appointed (to the knowledge of the Corporation), the Corporation, in its sole discretion, may deliver such shares and/or cash to the spouse or to any one or more dependents or relatives of the Participant, or if no spouse, dependent or relative is known to the Corporation, then to such other person as the Corporation may designate.

14. TRANSFERABILITY

- (a) Neither Contributions credited to a Participant's Account nor any Options or rights with respect to the exercise of Options or right to receive shares under this Plan may be anticipated, alienated, encumbered, assigned, transferred, pledged or otherwise disposed of in any way (other than by will, the laws of descent and distribution, or as provided in Section 13) by the Participant. Any such attempt at anticipation, alienation, encumbrance, assignment, transfer, pledge or other disposition shall be without effect and all amounts shall be paid and all shares shall be delivered in accordance with the provisions of this Plan. Amounts payable or shares deliverable pursuant to this Plan shall be paid or delivered only to (or credited in the name of, as the case may be) the Participant or, in the event of the Participant's death, as provided in Section 13.
- (b) The Corporation may require a Participant to hold any shares the Participant acquires under this Plan in a brokerage account identified by the Corporation until the date the shares are transferred, sold or otherwise disposed of in any way by the Participant, or such earlier time as the Corporation may determine.

15. USE OF FUNDS; INTEREST

All Contributions received or held by the Corporation under this Plan will be included in the general assets of the Corporation and may be used for any corporate purpose. Notwithstanding anything else contained herein to the contrary, no interest will be paid to any Participant or credited to his or her Account under this Plan (in respect of Account balances, refunds of Account balances, or otherwise).

16. **REPORTS**

Statements shall be provided or made available (in writing or electronically) to Participants as soon as administratively practicable following each Exercise Date. Each Participant's statement shall set forth, as of such Exercise Date, that Participant's Account balance immediately prior to the exercise of his or her Option, the Option Price, the number of whole shares purchased and his or her remaining Account balance, if any.

17. ADJUSTMENTS OF AND CHANGES IN THE STOCK

- (a) Upon or in contemplation of any reclassification, recapitalization, stock split (including a stock split in the form of a stock dividend), or reverse stock split; any merger, combination, consolidation, or other reorganization; split-up, spin-off, or any similar extraordinary dividend distribution in respect of the Common Stock (whether in the form of securities or property); any exchange of Common Stock or other securities of the Corporation, or any similar, unusual or extraordinary corporate transaction in respect of the Common Stock; or a sale of substantially all the assets of the Corporation as an entirety occurs; then the Committee shall, in such manner, to such extent (if any) and at such time as it deems appropriate and equitable in the circumstances:
 - (i) proportionately adjust any or all of (A) the number and type of shares of Common Stock or the number and type of other securities that thereafter may be made the subject of Options (including the specific maxima and numbers of shares set forth elsewhere in this Plan), (B) the number, amount and type of shares (or other securities or property) subject to any or all outstanding Options, (C) the Option Price of any or all outstanding Options, or (D) the securities, cash or other property deliverable upon exercise of any outstanding Options, in each case to the extent necessary to preserve (but not increase) the level of incentives intended by this Plan and the then-outstanding Options; or
 - (ii) make provision for a cash payment in settlement of, or for the substitution or exchange of, any or all outstanding Options or the cash, securities or property deliverable to the holder of any or all outstanding Options based upon the distribution or consideration payable to holders of the Common Stock upon or in respect of such event.
- (b) The Committee may adopt such valuation methodologies for outstanding Options as it deems reasonable in the event of a cash or property settlement and, without limitation on other methodologies, may base such settlement solely upon the excess (if any) of the amount payable upon or in respect of such event over the Option Price of the Option.
- (c) In any of such events, the Committee may take such action sufficiently prior to such event to the extent that the Committee deems the action necessary to permit the Participant to realize the benefits intended to be conveyed with respect to the underlying shares in the same manner as is or will be available to stockholders generally.
- (d) Without limiting the generality of Section 12, any good faith determination by the Committee as to whether an adjustment is required in the circumstances pursuant to this Section 17, and the extent and nature of any such adjustment, shall be conclusive and binding on all persons.

18. POSSIBLE EARLY TERMINATION OF PLAN AND OPTIONS

Upon a dissolution or liquidation of the Corporation, or any other event described in Section 17 that the Corporation does not survive, or does not survive as a publicly-traded company in respect of its shares, subject to any provision that has been expressly made by the Board for the survival, substitution, assumption, exchange or other settlement of the Options that are then outstanding under the Plan, each Offering Period then in progress shall be shortened and a new Exercise Date shall be established by the Board or the Committee (the "New Exercise Date"), as of which date the Plan and any Offering Period then in progress shall terminate and all then-outstanding Options under this Plan shall be automatically exercised in accordance with the terms hereof; provided, however, that the New Exercise Date shall not be more than ten (10) days before the date of the consummation of such dissolution, liquidation or other event. The Option Price on the New Exercise Date shall be determined as provided in Section 8(b), and the New Exercise Date shall be treated as the "Exercise Date" for purposes of determining such Option Price.

19. TERM OF PLAN; AMENDMENT OR TERMINATION

- (a) *Effective Date; Termination.* This Plan shall become effective as of the Effective Date. No new Offering Periods shall commence on or after the tenth (10th) anniversary of the Effective Date, and this Plan shall terminate as of the Exercise Date on or immediately following such date unless sooner terminated pursuant to Section 4, Section 18 or this Section 19.
- (b) Board Amendment Authority. The Board may, at any time, terminate or, from time to time, amend, modify or suspend this Plan, in whole or in part, without notice. Stockholder approval for any amendment or modification shall not be required, except to the extent required by law or applicable stock exchange rules, or required under Section 423 of the Code in order to preserve the intended tax consequences of this Plan. No Options may be granted during any suspension of this Plan or after the termination of this Plan, but the Committee will retain jurisdiction as to Options then outstanding in accordance with the terms of this Plan. No amendment, modification, or termination pursuant to this Section 19(b) shall, without written consent of the Participant, affect in any manner materially adverse to the Participant any rights or benefits of such Participant or obligations of the Corporation under any Option granted under this Plan prior to the effective date of such change. Changes contemplated by Section 17 or Section 18 shall not be deemed to constitute changes or amendments requiring Participant consent. Without limiting the generality of the Committee's amendment authority, the Committee shall have the right to designate from time to time the Subsidiaries whose employees may be eligible to participate in this Plan (including, without limitation, any Subsidiary that may become such after the Effective Date), to change the service and other qualification requirements set forth under the definition of Eligible Employee in Section 2, and to change the definition of Compensation set forth in Section 2 (in each case, subject to the requirements of Section 423(b) of the Code and applicable rules and regulations thereunder). Any such change shall not take effect earlier than the first Offering Period that starts on or after the effective date of such change. Any such change shall not constitute an amendment to this Plan requiring stockholder approval.

20. NOTICES

All notices or other communications by a Participant to the Corporation contemplated by this Plan shall be deemed to have been duly given when received in the form and manner specified by the Committee (or its delegate) at the location, or by the person, designated by the Committee (or its delegate) for that purpose.

21. CONDITIONS UPON ISSUANCE OF SHARES

This Plan, the granting of Options under this Plan and the offer, issuance and delivery of shares of Common Stock are subject to compliance with all applicable federal and state laws, rules and regulations (including but not limited to state and federal securities laws) and to such approvals by any listing, regulatory or governmental authority as may, in the opinion of counsel for the Corporation, be necessary or advisable in connection therewith. The person acquiring any securities under this Plan will, if requested by the Corporation and as a condition precedent to the exercise of his or her Option, provide such assurances and representations to the Corporation as the Committee may deem necessary or desirable to assure compliance with all applicable legal and accounting requirements.

22. PLAN CONSTRUCTION

- (a) Section 16. It is the intent of the Corporation that transactions involving Options under this Plan (other than "Discretionary Transactions" as that term is defined in Rule 16b-3(b)(1) promulgated by the Commission under Section 16 of the Exchange Act, to the extent there are any Discretionary Transactions under this Plan), in the case of Participants who are or may be subject to the prohibitions of Section 16 of the Exchange Act, satisfy the requirements for exemption under Rule 16b-3(c) promulgated by the Commission under Section 16 of the Exchange Act to the maximum extent possible. Notwithstanding the foregoing, the Corporation shall have no liability to any Participant for Section 16 consequences of Options or other events with respect to this Plan.
- (b) *Section 423.* Except as the Committee may expressly provide in the case of one or more separate offerings or sub-plans adopted pursuant to Section 12(b), this Plan and Options are intended to qualify under Section 423 of the Code.
- (c) *Interpretation*. If any provision of this Plan or of any Option would otherwise frustrate or conflict with the intents expressed above, that provision to the extent possible shall be interpreted so as to avoid such conflict. If the conflict remains irreconcilable, the Committee may disregard the provision if it concludes that to do so furthers the interest of the Corporation and is consistent with the purposes of this Plan as to such persons in the circumstances.

23. EMPLOYEES' RIGHTS

- (a) No Employment Rights. Nothing in this Plan (or in any Subscription Agreement or other document related to this Plan) will confer upon any Eligible Employee or Participant any right to continue in the employ or other service of the Corporation or any Subsidiary, constitute any contract or agreement of employment or other service or effect an employee's status as an employee at will, nor shall interfere in any way with the right of the Corporation or any Subsidiary to change such person's compensation or other benefits or to terminate his or her employment or other service, with or without cause. Nothing contained in this Section 23(a), however, is intended to adversely affect any express independent right of any such person under a separate employment or service contract other than a Subscription Agreement.
- (b) No Rights to Assets of the Corporation. No Participant or other person will have any right, title or interest in any fund or in any specific asset (including shares of Common Stock) of the Corporation or any Subsidiary by reason of any Option hereunder. Neither the provisions of this Plan (or of any Subscription Agreement or other document related to this Plan), nor the creation or adoption of this Plan, nor any action taken pursuant to the provisions of this Plan will create, or be construed to create, a trust of any kind or a fiduciary relationship between the Corporation or any Subsidiary, on the one hand, and any Participant or other person, on the other hand. To the extent that a Participant or other person acquires a right to receive payment pursuant to this Plan, such right will be no greater than the right of any unsecured general creditor of the Corporation. No special or separate reserve, fund or deposit will be made to assure any such payment.
- (c) *No Stockholder Rights.* A Participant will not be entitled to any privilege of stock ownership as to any shares not actually delivered to and held of record by the Participant. Except as expressly required by Section 17, no adjustment will be made for dividends or other rights as a stockholder for which a record date is prior to such date of delivery.

24. MISCELLANEOUS

(a) Governing Law; Severability. This Plan, the Options, Subscription Agreements, and other documents related to this Plan shall be governed by, and construed in accordance with, the laws of the State of Delaware without reference to any conflict of law principles. If any provision shall be held by a court of competent jurisdiction to be invalid and unenforceable, the remaining provisions of this Plan shall continue in effect.

- (b) *Captions and Headings*. Captions and headings are given to the sections of this Plan solely as a convenience to facilitate reference. Such captions and headings shall not be deemed in any way material or relevant to the construction of interpretation of this Plan or any provision hereof.
- (c) *No Effect on Other Plans or Corporate Authority.* The adoption of this Plan shall not affect any other Corporation or Subsidiary compensation or incentive plans in effect. Nothing in this Plan will limit or be deemed to limit the authority of the Board or Committee (i) to establish any other forms of incentives or compensation for employees of the Corporation or any Subsidiary (with or without reference to the Common Stock), or (ii) to grant or assume options (outside the scope of and in addition to those contemplated by this Plan) in connection with any proper corporate purpose; to the extent consistent with any other plan or authority.
- (d) *No Effect on Other Compensation.* Benefits received by a Participant under an Option granted pursuant to this Plan shall not be deemed a part of the Participant's compensation for purposes of the determination of benefits under any other employee welfare or benefit plans or arrangements, if any, provided by the Corporation or any Subsidiary, except where the Committee or the Board (or the board of directors of the Subsidiary that sponsors such plan or arrangement, as applicable) expressly otherwise provides in writing.

25. STOCKHOLDER APPROVAL

Notwithstanding anything else contained herein to the contrary, the effectiveness of this Plan is subject to the approval of this Plan by the stockholders of the Corporation within twelve months after the Effective Date. No shares shall be issued or delivered under this Plan until such stockholder approval is obtained and, if such stockholder approval is not obtained within such twelve-month period of time, all Contributions credited to a Participant's Account hereunder shall be refunded to such Participant (without interest) as soon as practicable after the end of such twelve-month period.

26. TAX WITHHOLDING

- (a) Notwithstanding anything else contained herein to the contrary, the Corporation may deduct from a Participant's Account balance as of an Exercise Date, before the exercise of the Participant's Option is given effect on such date, the amount of any taxes which the Corporation reasonably determines it or any Subsidiary may be required to withhold with respect to such exercise. In such event, the maximum number of whole shares of Common Stock subject to such Option (subject to the other limits set forth in this Plan) shall be purchased at the Option Price with the balance of the Participant's Account (after reduction for the tax withholding amount).
- (b) Should the Corporation for any reason be unable, or elect not to, satisfy its or any Subsidiary's tax withholding obligations in the manner described in the preceding paragraph with respect to a Participant's exercise of an Option, or should the Corporation or any Subsidiary reasonably determine that it or an affiliated entity has a tax withholding obligation with respect to a disposition of shares acquired pursuant to the exercise of an Option prior to satisfaction of the holding period requirements of Section 423 of the Code or at any other time in respect of a Participant's participation in this Plan, the Corporation or Subsidiary, as the case may be, shall have the right at its option to (i) require the Participant to pay or provide for payment of the amount of any taxes which the Corporation or Subsidiary reasonably determines that it or any affiliate is required to withhold with respect to such event or (ii) deduct from the Participant's Account or from any amount otherwise payable to or for the account of the Participant the amount of any taxes which the Corporation or Subsidiary reasonably determines that it or an affiliate is required to withhold with respect to such event.

27. **NOTICE OF SALE**

Any person who has acquired shares under this Plan shall give prompt written notice to the Corporation of any sale or other transfer of the shares if such sale or transfer occurs (a) within the two-year period after the Grant Date of the Offering Period with respect to which such shares were acquired, or (b) within the twelve-month period after the Exercise Date of the Offering Period with respect to which such shares were acquired.

CEPTON TECHNOLOGIES, INC. STOCK INCENTIVE PLAN

PREFACE

This Plan is divided into two separate equity programs: (1) the option and stock appreciation rights grant program set forth in Section 5 under which Eligible Persons (as defined in Section 3) may, at the discretion of the Administrator, be granted Options and/or SARs, and (2) the stock award program set forth in Section 6 under which Eligible Persons may, at the discretion of the Administrator, be awarded restricted or unrestricted shares of Common Stock. Section 2 of this Plan contains the general rules regarding the administration of this Plan. Section 3 sets forth the requirements for eligibility to receive an Award grant under this Plan. Section 4 describes the capital stock of the Corporation that may be subject to Awards granted under this Plan. Section 7 contains other provisions applicable to all Awards granted under this Plan. Section 8 provides definitions for certain capitalized terms used in this Plan and not otherwise defined herein.

1. PURPOSE OF THE PLAN.

The purpose of this Plan is to promote the success of the Corporation and the interests of its stockholders by providing a means through which the Corporation may grant equity-based incentives to attract, motivate, retain and reward certain officers, employees, directors and other eligible persons and to further link the interests of Award recipients with those of the Corporation's stockholders generally.

2. ADMINISTRATION.

Administrator. This Plan shall be administered by and all Awards under this Plan shall be authorized by the Administrator. The "Administrator" means the Board or one or more committees appointed by the Board or another committee (within its delegated authority) to administer all or certain aspects of this Plan. Any such committee shall be comprised solely of one or more directors or such number of directors as may be required under applicable law. A committee may delegate some or all of its authority to another committee so constituted. The Board or a committee comprised solely of directors may also delegate, to the extent permitted by Section 157(c) of the Delaware General Corporation Law and any other applicable law, to one or more officers of the Corporation, its powers under this Plan (a) to designate the officers and employees of the Corporation and its Affiliates who will receive grants of Awards under this Plan, and (b) to determine the number of shares subject to, and the other terms and conditions of, such Awards. The Board may delegate different levels of authority to different committees with administrative and grant authority under this Plan. Unless otherwise provided in the Bylaws of the Corporation or the applicable charter of any Administrator: (a) a majority of the members of the acting Administrator shall constitute a quorum, and (b) the vote of a majority of the members present assuming the presence of a quorum or the unanimous written consent of the members of the Administrator shall constitute action by the acting Administrator.

- **2.2 Plan Awards; Interpretation; Powers of Administrator.** Subject to the express provisions of this Plan, the Administrator is authorized and empowered to do all things necessary or desirable in connection with the authorization of Awards and the administration of this Plan (in the case of a committee or delegation to one or more officers, within the authority delegated to that committee or person(s)), including, without limitation, the authority to:
 - (a) determine eligibility and, from among those persons determined to be eligible, the particular Eligible Persons who will receive Awards;
 - (b) grant Awards to Eligible Persons, determine the price and number of securities to be offered or awarded to any of such persons, determine the other specific terms and conditions of Awards consistent with the express limits of this Plan, establish the installments (if any) in which such Awards will become exercisable or will vest (which may include, without limitation, performance and/or time-based schedules) or determine that no delayed exercisability or vesting is required, establish any applicable performance targets, and establish the events of termination or reversion of such Awards;
 - (c) approve the forms of Award Agreements, which need not be identical either as to type of Award or among Participants;
 - (d) construe and interpret this Plan and any Award Agreement or other agreements defining the rights and obligations of the Corporation, its Affiliates, and Participants under this Plan, make factual determinations with respect to the administration of this Plan, further define the terms used in this Plan, and prescribe, amend and rescind rules and regulations relating to the administration of this Plan or the Awards;
 - (e) cancel, modify, or waive the Corporation's rights with respect to, or modify, discontinue, suspend, or terminate any or all outstanding Awards, subject to any required consent under Section 7.7.4;
 - (f) accelerate or extend the vesting or exercisability or extend the term of any or all outstanding Awards (within the maximum tenyear term of Awards under Sections 5.4.2 and 6.5) in such circumstances as the Administrator may deem appropriate (including, without limitation, in connection with a termination of employment or services or other events of a personal nature);
 - (g) determine Fair Market Value for purposes of this Plan and Awards;
 - (h) determine the duration and purposes of leaves of absence that may be granted to Participants without constituting a termination of their employment for purposes of this Plan; and

- (i) determine whether, and the extent to which, adjustments are required pursuant to Section 7.3 hereof and authorize the termination, conversion, substitution or succession of awards upon the occurrence of an event of the type described in Section 7.3
- **Binding Determinations.** Any action taken by, or inaction of, the Corporation, any Affiliate, the Board or the Administrator relating or pursuant to this Plan and within its authority hereunder or under applicable law shall be within the absolute discretion of that entity or body and shall be conclusive and binding upon all persons. Neither the Board nor the Administrator, nor any member thereof or person acting at the direction thereof, shall be liable for any act, omission, interpretation, construction or determination made in good faith in connection with this Plan (or any Award), and all such persons shall be entitled to indemnification and reimbursement by the Corporation in respect of any claim, loss, damage or expense (including, without limitation, attorneys' fees) arising or resulting therefrom to the fullest extent permitted by law and/or under any directors and officers liability insurance coverage that may be in effect from time to time.
- **2.4 Reliance on Experts.** In making any determination or in taking or not taking any action under this Plan, the Administrator may obtain and may rely upon the advice of experts, including employees of and professional advisors to the Corporation. No director, officer or agent of the Corporation or any of its Affiliates shall be liable for any such action or determination taken or made or omitted in good faith.
- **2.5 Delegation**. The Administrator may delegate ministerial, non-discretionary functions to individuals who are officers or employees of the Corporation or any of its Affiliates or to third parties.

3. ELIGIBILITY.

Awards may be granted under this Plan only to those persons that the Administrator determines to be Eligible Persons. An "Eligible Person" means any person who qualifies as one of the following at the time of grant of the respective Award:

- (a) an officer (whether or not a director) or employee of the Corporation or any of its Affiliates;
- (b) any member of the Board; or
- (c) any director of one of the Corporation's Affiliates, or any individual consultant or advisor who renders bona fide services (other than services in connection with the offering or sale of securities of the Corporation or one of its Affiliates, as applicable, in a capital raising transaction or as a market maker or promoter of that entity's securities) to the Corporation or one of its Affiliates.

An advisor or consultant may be selected as an Eligible Person pursuant to clause (c) above only if such person's participation in this Plan would not adversely affect (1) the Corporation's eligibility to rely on the Rule 701 exemption from registration under the Securities Act for the offering of shares issuable under this Plan by the Corporation, or (2) the Corporation's compliance with any other applicable laws.

An Eligible Person may, but need not, be granted one or more Awards pursuant to Section 5 and/or one or more Awards pursuant to Section 6. An Eligible Person who has been granted an Award under this Plan may, if otherwise eligible, be granted additional Awards under this Plan if the Administrator so determines. However, a person's status as an Eligible Person is not a commitment that any Award will be granted to that person under this Plan. Furthermore, an Eligible Person who has been granted an Award under Section 5 is not necessarily entitled to an Award under Section 6, or vice versa, unless otherwise expressly determined by the Administrator.

Each Award granted under this Plan must be approved by the Administrator at or prior to the grant of the Award.

4. STOCK SUBJECT TO THE PLAN.

- **Shares** *Available*. Subject to the provisions of Section 7.3.1, the capital stock that may be delivered under this Plan will be shares of the Corporation's authorized but unissued Common Stock and any of its shares of Common Stock held as treasury shares. The shares of Common Stock issued and delivered may be issued and delivered for any lawful consideration.
- **Share Limit.** Subject to the provisions of Section 7.3.1 and further subject to the share counting rules of Section 4.3, the maximum number of shares of Common Stock that may be delivered pursuant to Awards granted under this Plan will not exceed 9,187,533 shares (the "**Share Limit**") in the aggregate. As required under Treasury Regulation Section 1.422-2(b)(3)(i), in no event will the number of shares of Common Stock that may be delivered pursuant to Incentive Stock Options granted under this Plan exceed the Share Limit.
- 4.3 Replenishment and Reissue of Unvested Awards. To the extent that an Award is settled in cash or a form other than shares of Common Stock, the shares that would have been delivered had there been no such cash or other settlement shall not be counted against the shares available for issuance under this Plan. No Award may be granted under this Plan unless, on the date of grant, the sum of (a) the maximum number of shares of Common Stock issuable at any time pursuant to such Award, plus (b) the number of shares of Common Stock that have previously been issued pursuant to Awards granted under this Plan, plus (c) the maximum number of shares of Common Stock that may be issued at any time after such date of grant pursuant to Awards that are outstanding on such date, does not exceed the Share Limit. Shares of Common Stock that are subject to or underlie Options or SARs granted under this Plan that expire or for any reason are canceled or terminated without having been exercised (or shares of Common Stock subject to or underlying the unexercised portion of such Options or SARs in the case of Options or SARs that were partially exercised), as well as shares of Common Stock that are subject to Stock Awards made under this Plan that are forfeited to the Corporation or otherwise repurchased by the Corporation prior to the vesting of such shares for a price not greater than the original purchase or issue price of such shares (as adjusted pursuant to Section 7.3.1) will again, except to the extent prohibited by law or applicable listing or regulatory requirements, be available for subsequent Award grants under this Plan. Shares that are exchanged by a Participant or withheld by the Corporation as full or partial payment in connection with any Award under this Plan, as well as any shares exchanged by a Participant or withheld by the Corporation or one of its Affiliates to satisfy the tax withholding obligations related to any Award, shall be available for subsequent Awards under this Plan. In the case of an exercise of a SAR, only the number of shares actually issued in respect of such exercise shall be charged against this Plan's Share Limit. Adjustments to the Share Limit pursuant to this Section 4.3 are subject to any applicable limitations of the Code in the case of Awards intended to be Incentive Stock Options.

4.4 Reservation of Shares. The Corporation shall at all times reserve a number of shares of Common Stock sufficient to cover the Corporation's obligations and contingent obligations to deliver shares with respect to Awards then outstanding under this Plan.

5. OPTION AND SAR GRANT PROGRAM.

- **Option and SAR Grants in General**. Each Option or SAR shall be evidenced by an Award Agreement in the form approved by the Administrator. The Award Agreement evidencing an Option or SAR shall contain the terms established by the Administrator for that Award, as well as any other terms, provisions, or restrictions that the Administrator may impose on the Option or SAR or any shares of Common Stock subject to the Option or SAR; in each case subject to the applicable provisions and limitations of this Section 5 and the other applicable provisions and limitations of this Plan. The Administrator may require that the recipient of an Option or SAR promptly execute and return to the Corporation his or her Award Agreement evidencing the Award. In addition, the Administrator may require that the spouse of any married recipient of an Option or SAR also promptly execute and return to the Corporation the Award Agreement evidencing the Award granted to the recipient or such other spousal consent form that the Administrator may require in connection with the grant of the Award.
- **Incentive Stock Option Status.** The Administrator will designate each Option granted under this Plan as either an Incentive Stock Option or a Nonqualified Stock Option, and such designation shall be set forth in the applicable Award Agreement. Any Option granted under this Plan that is not expressly designated in the applicable Award Agreement as an Incentive Stock Option will be deemed to be designated a Nonqualified Stock Option under this Plan and not an "incentive stock option" within the meaning of Section 422 of the Code. Incentive Stock Options shall be subject to the provisions of Section 5.5 in addition to the provisions of this Plan applicable to Options generally.

5.3 Option or SAR Price.

- **5.3.1** Option Pricing Limits. Subject to the following provisions of this Section 5.3.1, the Administrator will determine the purchase price per share of the Common Stock covered by each Option (the "exercise price" of the Option) at the time of the grant of the Option, which exercise price will be set forth in the applicable Award Agreement. In no case will the exercise price of an Option be less than the greater of:
 - (a) the par value of the Common Stock;
 - (b) subject to clause (c) below, 100% of the Fair Market Value of a share of Common Stock on the date of grant; or
 - (c) in the case of an Incentive Stock Option granted to a Participant described in Section 5.5.4, 110% of the Fair Market Value of a share of Common Stock on the date of grant.
- 5.3.2 Payment Provisions. The Corporation will not be obligated to deliver certificates for the shares of Common Stock to be purchased on exercise of an Option unless and until it receives full payment of the exercise price therefor, all related withholding obligations under Section 7.6 have been satisfied, and all other conditions to the exercise of the Option set forth herein or in the Award Agreement have been satisfied. The purchase price of any shares of Common Stock purchased on exercise of an Option must be paid in full at the time of each purchase in such lawful consideration as may be permitted or required by the Administrator, which may include, without limitation, one or a combination of the following methods:
 - (a) cash, check payable to the order of the Corporation, or electronic funds transfer;
 - (b) notice and third party payment in such manner as may be authorized by the Administrator;
 - (c) the delivery of previously owned shares of Common Stock;
 - (d) by a reduction in the number of shares of Common Stock otherwise deliverable pursuant to the Award;
 - (e) subject to such procedures as the Administrator may adopt, pursuant to a "cashless exercise"; or

(f) if authorized by the Administrator or specified in the applicable Award Agreement, by a promissory note of the Participant consistent with the requirements of Section 5.3.3.

In no event shall any shares newly-issued by the Corporation be issued for less than the minimum lawful consideration for such shares or for consideration other than consideration permitted by applicable state law. Shares of Common Stock used to satisfy the exercise price of an Option (whether previously-owned shares or shares otherwise deliverable pursuant to the terms of the Option) shall be valued at their Fair Market Value on the date of exercise. Unless otherwise expressly provided in the applicable Award Agreement, the Administrator may eliminate or limit a Participant's ability to pay the purchase or exercise price of any Award by any method other than cash payment to the Corporation.

- **5.3.3** Acceptance of Notes to Finance Exercise. The Corporation may, with the Administrator's approval in each specific case, accept one or more promissory notes from any Eligible Person in connection with the exercise of any Option; provided that any such note shall be subject to the following terms and conditions:
 - (a) The principal of the note shall not exceed the amount required to be paid to the Corporation upon the exercise, purchase or acquisition of one or more Awards under this Plan and the note shall be delivered directly to the Corporation in consideration of such exercise, purchase or acquisition.
 - (b) The initial term of the note shall be determined by the Administrator; provided that the term of the note, including extensions, shall not exceed a period of five years.
 - (c) The note shall provide for full recourse to the Participant and shall bear interest at a rate determined by the Administrator, but not less than the interest rate necessary to avoid the imputation of interest under the Code and to avoid any adverse accounting consequences in connection with the exercise, purchase or acquisition.
 - (d) If the employment or services of the Participant by or to the Corporation and its Affiliates terminates, the unpaid principal balance of the note shall become due and payable on the 30th business day after such termination; provided, however, that if a sale of the shares acquired on exercise of the Option would cause such Participant to incur liability under Section 16(b) of the Exchange Act, the unpaid balance shall become due and payable on the 10th business day after the first day on which a sale of such shares could have been made without incurring such liability assuming for these purposes that there are no other transactions (or deemed transactions) in securities of the Corporation by the Participant subsequent to such termination.

(e) If required by the Administrator or by applicable law, the note shall be secured by a pledge of any shares or rights financed thereby or other collateral, in compliance with applicable law.

The terms, repayment provisions, and collateral release provisions of the note and the pledge securing the note shall conform with all applicable rules and regulations, including those of the Federal Reserve Board and any applicable state law, as then in effect.

Base Price of SARs. The Administrator will determine the base price per share of the Common Stock covered by each SAR at the time of grant of the SAR, which base price will be set forth in the applicable Award Agreement and will not be less than 100% of the Fair Market Value of a share of Common Stock on the date of grant of the SAR.

5.4 Vesting; Term; Exercise Procedure.

- 5.4.1 <u>Vesting.</u> Except as provided in Section 5.8, an Option or SAR may be exercised only to the extent that it is vested and exercisable. The Administrator will determine the vesting and/or exercisability provisions of each Option or SAR (which may be based on performance criteria, passage of time or other factors or any combination thereof), which provisions will be set forth in the applicable Award Agreement. Unless the Administrator otherwise expressly provides, once exercisable an Option or SAR will remain exercisable until the expiration or earlier termination of the Option or SAR.
- **5.4.2** <u>Term.</u> Each Option or SAR shall expire not more than 10 years after its date of grant. Each Option or SAR will be subject to earlier termination as provided in or pursuant to Sections 5.6 and 7.3 or the terms of the applicable Award Agreement.
- 5.4.3 Exercise Procedure. Any exercisable Option or SAR will be deemed to be exercised when (a) the applicable exercise procedures in the related Award Agreement have been satisfied (or, in the absence of any such procedures in the related Award Agreement, the Corporation has received written notice of such exercise from the Participant), (b) in the case of an Option, the Corporation has received any required payment made in accordance with Section 5.3, (c) in the case of an Option or SAR, all withholding obligations arising in connection with the exercise have been satisfied in accordance with Section 7.6, and (d) in the case of an Option or SAR, the Corporation has received any written statement required pursuant to Section 7.5.1.

5.4.4 <u>Fractional Shares/Minimum Issue</u>. Fractional share interests will be disregarded, but may be accumulated. The Administrator, however, may determine that cash, other securities, or other property will be paid or transferred in lieu of any fractional share interests. No Option or SAR may be exercised as to fewer than 100 shares (subject to adjustment pursuant to Section 7.3.1) at one time unless the number as to which the Award is exercised is the total number at the time then subject to the vested and exercisable portion of the Award.

5.5 Limitations on Grant and Terms of Incentive Stock Options.

- 5.5.1 \$100,000 Limit. To the extent that the aggregate Fair Market Value of stock with respect to which incentive stock options (within the meaning of Section 422 of the Code) first become exercisable by a Participant in any calendar year exceeds \$100,000, taking into account both Common Stock subject to Incentive Stock Options under this Plan and stock subject to incentive stock options under all other plans of the Corporation or any of its Affiliates, such options will be treated as nonqualified stock options. For this purpose, the Fair Market Value of the stock subject to options will be determined as of the date the options were awarded. In reducing the number of options treated as incentive stock options to meet the \$100,000 limit, the most recently granted options will be reduced (recharacterized as nonqualified stock options) first. To the extent a reduction of simultaneously granted options is necessary to meet the \$100,000 limit, the Administrator may, in the manner and to the extent permitted by law, designate which shares of Common Stock are to be treated as shares acquired pursuant to the exercise of an incentive stock option.
- **5.5.2** Other Code Limits. Incentive Stock Options may only be granted to individuals that are employees of the Corporation or one of its Affiliates and satisfy the other eligibility requirements of the Code. Any Award Agreement relating to Incentive Stock Options will contain or shall be deemed to contain such other terms and conditions as from time to time are required in order that the Option be an "incentive stock option" as that term is defined in Section 422 of the Code.
- **5.5.3** <u>ISO Notice of Sale Requirement.</u> Any Participant who exercises an Incentive Stock Option shall give prompt written notice to the Corporation of any sale or other transfer of the shares of Common Stock acquired on such exercise if the sale or other transfer occurs within (a) one year after the exercise date of the Option, or (b) two years after the grant date of the Option.
- 5.5.4 <u>Limits on 10% Holders</u>. No Incentive Stock Option may be granted to any person who, at the time the Incentive Stock Option is granted, owns (or is deemed to own under Section 424(d) of the Code) shares of outstanding stock of the Corporation (or any of its Affiliates) possessing more than 10% of the total combined voting power of all classes of stock of the Corporation (or any of its Affiliates), unless the exercise price of such Incentive Stock Option is at least 110% of the Fair Market Value of the stock subject to the Incentive Stock Option and the Incentive Stock Option by its terms is not exercisable more than five years after the date the Incentive Stock Option is granted.

5.6 Effects of Termination of Employment on Options and SARs.

- **5.6.1** <u>Dismissal for Cause</u>. Unless otherwise provided in the applicable Award Agreement and subject to earlier termination pursuant to or as contemplated by Section 5.4.2 or 7.3, if a Participant's employment by or service to the Corporation or any of its Affiliates is terminated by such entity for Cause, the Participant's Option or SAR will terminate on the Participant's Severance Date, whether or not the Option or SAR is then vested and/or exercisable.
- 5.6.2 <u>Death or Disability</u>. Unless otherwise provided in the applicable Award Agreement (consistent with applicable securities laws) and subject to earlier termination pursuant to or as contemplated by Section 5.4.2 or 7.3, if a Participant's employment by or service to the Corporation or any of its Affiliates terminates as a result of the Participant's death or Total Disability:
 - (a) the Participant (or his or her Personal Representative or Beneficiary, in the case of the Participant's Total Disability or death, respectively), will have until the date that is 12 months after the Participant's Severance Date to exercise the Participant's Option or SAR (or portion thereof) to the extent that it was vested and exercisable on the Severance Date;
 - (b) the Option or SAR, to the extent not vested and exercisable on the Participant's Severance Date, shall terminate on the Severance Date; and
 - (c) the Option or SAR, to the extent exercisable for the 12-month period following the Participant's Severance Date and not exercised during such period, shall terminate at the close of business on the last day of the 12-month period.

- 5.6.3 Other Terminations of Employment. Unless otherwise provided in the applicable Award Agreement (consistent with applicable securities laws) and subject to earlier termination pursuant to or as contemplated by Section 5.4.2 or 7.3, if a Participant's employment by or service to the Corporation or any of its Affiliates terminates for any reason other than a termination by such entity for Cause or because of the Participant's death or Total Disability:
 - (a) the Participant will have until the date that is 3 months after the Participant's Severance Date to exercise his or her Option or SAR (or portion thereof) to the extent that it was vested and exercisable on the Severance Date;
 - (b) the Option or SAR, to the extent not vested and exercisable on the Participant's Severance Date, shall terminate on the Severance Date; and
 - (c) the Option or SAR, to the extent exercisable for the 3-month period following the Participant's Severance Date and not exercised during such period, shall terminate at the close of business on the last day of the 3-month period.
- 5.7 Option and SAR Repricing/Cancellation and Regrant/Waiver of Restrictions. Subject to Section 4 and Section 7.7 and the specific limitations on Options and SARs contained in this Plan, the Administrator from time to time may authorize, generally or in specific cases only, for the benefit of any Eligible Person, any adjustment in the exercise or base price, the vesting schedule, the number of shares subject to, or the term of, an Option or SAR granted under this Plan by cancellation of an outstanding Option or SAR and a subsequent regranting of the Option or SAR, by amendment, by substitution of an outstanding Option or SAR, by waiver or by other legally valid means. Such amendment or other action may result in, among other changes, an exercise or base price that is higher or lower than the exercise or base price of the original or prior Option or SAR, provide for a greater or lesser number of shares of Common Stock subject to the Option or SAR, or provide for a longer or shorter vesting or exercise period. In no event, however, may any such amendment or other action reduce the exercise or base price of the Option or SAR to less than the Fair Market Value of a share of Common Stock at the time of such change, or extend the maximum term of the Option or SAR at a time when the exercise or base price of such Award is less than the Fair Market Value of a share of Common Stock.
- 5.8 Early Exercise Options and SARs. The Administrator may, in its discretion, designate any Option or SAR as an "early exercise Option" or "early exercise SAR" which, by express provision in the applicable Award Agreement, may be exercised prior to the date such Option or SAR has vested. If the Participant elects to exercise all or a portion of any early exercise Option or SAR before it is vested, the shares of Common Stock acquired under the Option or SAR which are attributable to the unvested portion of the Option or SAR shall be Restricted Shares. The applicable Award Agreement will specify the extent (if any) to which and the time (if ever) at which the Participant will be entitled to dividends, voting and other rights in respect of such Restricted Shares prior to vesting, and the restrictions imposed on such shares and the conditions of release or lapse of such restrictions. Unless otherwise expressly provided in the applicable Award Agreement, such Restricted Shares shall be subject to the provisions of Sections 6.6 through 6.9, below.

6. STOCK AWARD PROGRAM.

- **Stock Awards in General.** Each Stock Award shall be evidenced by an Award Agreement in the form approved by the Administrator. The Award Agreement evidencing a Stock Award shall contain the terms established by the Administrator for that Stock Award, as well as any other terms, provisions, or restrictions that the Administrator may impose on the Stock Award; in each case subject to the applicable provisions and limitations of this Section 6 and the other applicable provisions and limitations of this Plan. The Administrator may require that the recipient of a Stock Award promptly execute and return to the Corporation his or her Award Agreement evidencing the Stock Award. In addition, the Administrator may require that the spouse of any married recipient of a Stock Award also promptly execute and return to the Corporation the Award Agreement evidencing the Stock Award granted to the recipient or such other spousal consent form that the Administrator may require in connection with the grant of the Stock Award.
- **6.2 Types of Stock Awards**. The Administrator shall designate whether a Stock Award shall be a Restricted Stock Award, and such designation shall be set forth in the applicable Award Agreement.

6.3 Purchase Price.

- **6.3.1** Pricing Limits. Subject to the following provisions of this Section 6.3, the Administrator will determine the purchase price per share of the Common Stock covered by each Stock Award at the time of grant of the Award. In no case will such purchase price be less than the par value of the Common Stock.
- **Payment Provisions.** The Corporation will not be obligated to issue certificates evidencing shares of Common Stock awarded under this Section 6 unless and until it receives full payment of the purchase price therefor and all other conditions to the purchase, as determined by the Administrator, have been satisfied. The purchase price of any shares subject to a Stock Award must be paid in full at the time of the purchase in such lawful consideration as may be permitted or required by the Administrator, which may include, without limitation, one or a combination of the methods set forth in clauses (a) through (f) in Section 5.3.2 and/or past services rendered to the Corporation or any of its Affiliates.
- **6.4 Vesting.** The restrictions imposed on the shares of Common Stock subject to a Restricted Stock Award (which may be based on performance criteria, passage of time or other factors or any combination thereof) will be set forth in the applicable Award Agreement.

- **Term; Settlement of Awards.** A Stock Award shall either vest or be forfeited not more than 10 years after the date of grant. Each Stock Award will be subject to earlier termination as provided in or pursuant to Sections 6.8 and 7.3. Payment of Awards may be in the form of cash, shares of Common Stock, other Awards or combinations thereof as the Administrator shall determine, and with such restrictions as it may impose. The Administrator may also require or permit Participants to elect to defer the issuance of shares or the settlement of Awards in cash under such rules and procedures as it may establish under this Plan. The Administrator may also provide that deferred settlements include the payment or crediting of interest or other earnings on the deferral amounts, or the payment or crediting of dividend equivalents where the deferred amounts are denominated in shares.
- **Stock Certificates; Fractional Shares.** Stock certificates evidencing Restricted Shares will bear a legend making appropriate reference to the restrictions imposed hereunder and will be held by the Corporation or by a third party designated by the Administrator until the restrictions on such shares have lapsed, the shares have vested in accordance with the provisions of the Award Agreement and Section 6.4, and any related loan has been repaid. Fractional share interests will be disregarded, but may be accumulated. The Administrator, however, may determine that cash, other securities, or other property will be paid or transferred in lieu of any fractional share interests.
- **6.7 Dividend and Voting Rights.** Unless otherwise provided in the applicable Award Agreement, a Participant receiving Restricted Shares will be entitled to cash dividend and voting rights for all Restricted Shares issued even though they are not vested, but such rights will terminate immediately as to any Restricted Shares which cease to be eligible for vesting.
- **Termination of Employment; Return to the Corporation**. Unless the Administrator otherwise expressly provides, Restricted Shares subject to an Award that remain subject to vesting conditions that have not been satisfied by the time specified in the applicable Award Agreement (which may include, without limitation, the Participant's Severance Date), will not vest and will be reacquired by the Corporation in such manner and on such terms as the Administrator provides, which terms shall include, to the extent not prohibited by law, return or repayment of the lower of (a) the Fair Market Value of the Restricted Shares at the time of the termination, or (b) the original purchase price of the Restricted Shares, without interest, to the Participant. The Award Agreement shall specify any other terms or conditions of the repurchase if the Award fails to vest. Any other Stock Award that has not been exercised or paid as of a Participant's Severance Date shall terminate on that date unless otherwise expressly provided by the Administrator in the applicable Award Agreement.
- **6.9** *Waiver of Restrictions*. Subject to Sections 4 and 7.7 and the specific limitations on Stock Awards contained in this Plan, the Administrator from time to time may authorize, generally or in specific cases only, for the benefit of any Eligible Person, any adjustment in the vesting schedule, or the restrictions upon or the term of, a Stock Award granted under this Plan by amendment, by substitution of an outstanding Stock Award, by waiver or by other legally valid means.

7. PROVISIONS APPLICABLE TO ALL AWARDS.

- 7.1 Rights of Eligible Persons, Participants and Beneficiaries.
 - **7.1.1** Employment Status. No person shall have any claim or rights to be granted an Award (or additional Awards, as the case may be) under this Plan, subject to any express contractual rights (set forth in a document other than this Plan) to the contrary.
 - No Employment/Service Contract. Nothing contained in this Plan (or in any other documents under this Plan or related to any Award) shall confer upon any Eligible Person or Participant any right to continue in the employ or other service of the Corporation or any of its Affiliates, constitute any contract or agreement of employment or other service or affect an employee's status as an employee at will, nor shall interfere in any way with the right of the Corporation or any Affiliate to change such person's compensation or other benefits, or to terminate his or her employment or other service, with or without cause at any time. Nothing in this Section 7.1.2, or in Section 7.3 or 7.15, however, is intended to adversely affect any express independent right of such person under a separate employment or service contract. An Award Agreement shall not constitute a contract of employment or service.
 - **7.1.3** Plan Not Funded. Awards payable under this Plan will be payable in shares of Common Stock or from the general assets of the Corporation, and (except as to the share reservation provided in Section 4.4) no special or separate reserve, fund or deposit will be made to assure payment of such Awards. No Participant, Beneficiary or other person will have any right, title or interest in any fund or in any specific asset (including shares of Common Stock, except as expressly provided) of the Corporation or any of its Affiliates by reason of any Award hereunder. Neither the provisions of this Plan (or of any related documents), nor the creation or adoption of this Plan, nor any action taken pursuant to the provisions of this Plan will create, or be construed to create, a trust of any kind or a fiduciary relationship between the Corporation or any of its Affiliates and any Participant, Beneficiary or other person. To the extent that a Participant, Beneficiary or other person acquires a right to receive payment pursuant to any Award hereunder, such right will be no greater than the right of any unsecured general creditor of the Corporation.
 - 7.1.4 Charter Documents. The Certificate of Incorporation and Bylaws of the Corporation, as either of them may lawfully be amended from time to time, may provide for additional restrictions and limitations with respect to the Common Stock (including additional restrictions and limitations on the voting or transfer of Common Stock) or priorities, rights and preferences as to securities and interests prior in rights to the Common Stock. These restrictions and limitations are in addition to (and not in lieu of) those set forth in this Plan or any Award Agreement, and are incorporated herein by this reference.

7.2 No Transferability; Limited Exception to Transfer Restrictions.

- **7.2.1** <u>Limit on Exercise and Transfer</u>. Unless otherwise expressly provided in (or pursuant to) this Section 7.2, by applicable law and by the Award Agreement, as the same may be amended:
 - (a) all Awards are non-transferable and will not be subject in any manner to sale, transfer, anticipation, alienation, assignment, pledge, encumbrance or charge;
 - (b) Awards will be exercised only by the Participant; and
 - (c) amounts payable or shares issuable pursuant to an Award will be delivered only to (or for the account of) the Participant.

In addition, the shares shall be subject to the restrictions set forth in the applicable Award Agreement.

- **7.2.2** Further Exceptions to Limits on Transfer. The exercise and transfer restrictions in Section 7.2.1 will not apply to:
 - (a) transfers to the Corporation;
 - (b) transfers by gift or domestic relations order to one or more "family members" (as that term is defined in SEC Rule 701 promulgated under the Securities Act) of the Participant;
 - (c) the designation of a Beneficiary to receive benefits if the Participant dies or, if the Participant has died, transfers to or exercises by the Participant's Beneficiary, or, in the absence of a validly designated Beneficiary, transfers by will or the laws of descent and distribution; or
 - (d) if the Participant has suffered a disability, permitted transfers or exercises on behalf of the Participant by the Participant's duly authorized legal representative.

Notwithstanding anything else in this Section 7.2.2 to the contrary, but subject to compliance with all applicable laws, Incentive Stock Options and Restricted Stock Awards will be subject to any and all transfer restrictions under the Code applicable to such awards or necessary to maintain the intended tax consequences of such Awards. Notwithstanding clause (b) above but subject to compliance with all applicable laws, any contemplated transfer by gift or domestic relations order to one or more "family members" of a Participant as referenced in clause (b) above is subject to the condition precedent that the transfer be approved by the Administrator in order for it to be effective. The Administrator may, in its sole discretion, withhold its approval of any such proposed transfer.

7.3 Adjustments; Changes in Control.

Adjustments. Subject to Section 7.3.2 below, upon (or, as may be necessary to effect the adjustment, immediately prior to): any reclassification, recapitalization, stock split (including a stock split in the form of a stock dividend) or reverse stock split; any merger, combination, consolidation, conversion or other reorganization; any split-up, spin-off, or similar extraordinary dividend distribution in respect of the Common Stock; or any exchange of Common Stock or other securities of the Corporation, or any similar, unusual or extraordinary corporate transaction in respect of the Common Stock; then the Administrator shall equitably and proportionately adjust (1) the number and type of shares of Common Stock (or other securities) that thereafter may be made the subject of Awards (including the specific share limits, maximums and numbers of shares set forth elsewhere in this Plan), (2) the number, amount and type of shares of Common Stock (or other securities or property) subject to any outstanding Awards, (3) the grant, purchase, or exercise or base price of any outstanding Awards, and/or (4) the securities, cash or other property deliverable upon exercise or vesting of any outstanding Awards, in each case to the extent necessary to preserve (but not increase) the level of incentives intended by this Plan and the then-outstanding Awards.

Unless otherwise expressly provided in the applicable Award Agreement, upon (or, as may be necessary to effect the adjustment, immediately prior to) any event or transaction described in the preceding paragraph or a sale of all or substantially all of the business or assets of the Corporation as an entirety, the Administrator shall equitably and proportionately adjust the performance standards applicable to any then-outstanding performance-based Awards to the extent necessary to preserve (but not increase) the level of incentives intended by this Plan and the then-outstanding performance-based Awards.

It is intended that, if possible, any adjustments contemplated by the preceding two paragraphs be made in a manner that satisfies applicable U.S. legal, tax (including, without limitation and as applicable in the circumstances, Section 424 of the Code and Section 409A of the Code) and accounting (so as to not trigger any charge to earnings with respect to such adjustment) requirements.

Without limiting the generality of Section 2.3, any good faith determination by the Administrator as to whether an adjustment is required in the circumstances pursuant to this Section 7.3.1, and the extent and nature of any such adjustment, shall be conclusive and binding on all persons.

Unless otherwise expressly provided by the Administrator, in no event shall a conversion of one or more outstanding shares of the Corporation's preferred stock (if any) or any new issuance of securities by the Corporation for consideration be deemed, in and of itself, to require an adjustment pursuant to this Section 7.3.1.

7.3.2 <u>Consequences of a Change in Control Event.</u> Upon the occurrence of a Change in Control Event, the Administrator may make provision for a cash payment in settlement of, or for the assumption, substitution or exchange of any or all outstanding Awards (or the cash, securities or other property deliverable to the holder(s) of any or all outstanding Awards) based upon, to the extent relevant in the circumstances, the distribution or consideration payable to holders of the Common Stock upon or in respect of such event.

In addition, subject to Section 7.3.4, upon (or, as may be necessary to effectuate the purposes of this acceleration, immediately prior to) the occurrence of a Change in Control Event:

- (a) each Option and SAR will become immediately vested and exercisable, and
- (b) Restricted Stock will immediately vest free of forfeiture restrictions and/or restrictions giving the Corporation the right to repurchase the stock at its original purchase price, and each other Stock Award shall fully vest and be payable;

<u>provided</u>, <u>however</u>, that such acceleration provision shall not apply, unless otherwise expressly provided by the Administrator, with respect to any Award to the extent that the Administrator has made a provision for the substitution, assumption, exchange or other continuation or settlement of the Award, or the Award would otherwise continue in accordance with its terms, in the circumstances.

The foregoing Change in Control Event provisions shall not in any way limit the authority of the Administrator to accelerate the vesting of one or more Awards (as to all or only a portion of any Award) in such circumstances (including, but not limited to, a Change in Control Event) as the Administrator may determine to be appropriate, regardless of whether accelerated vesting of a portion of the Award(s) is otherwise required or contemplated by the foregoing in the circumstances.

The Administrator may adopt such valuation methodologies for outstanding Awards as it deems reasonable in the event of a cash, securities or other property settlement. In the case of Options and SARs, but without limitation on other methodologies, the Administrator may base such settlement solely upon the excess (if any) of the amount payable upon or in respect of such event over the exercise or base price of the Option or SAR, as applicable, to the extent of the then vested and exercisable shares subject to the Option or SAR.

In any of the events referred to in this Section 7.3.2, the Administrator may take such action contemplated by this Section 7.3.2 prior to such event (as opposed to on the occurrence of such event) to the extent that the Administrator deems the action necessary to permit the Participant to realize the benefits intended to be conveyed with respect to the underlying shares. Without limiting the generality of the foregoing, the Administrator may deem an acceleration to occur immediately prior to the applicable event and/or reinstate the original terms of the Award if an event giving rise to an acceleration does not occur.

7.3.3 Early Termination of Awards. Upon the occurrence of a Change in Control Event, each then-outstanding Award (whether or not vested and/or exercisable, but after giving effect to any accelerated vesting required in the circumstances pursuant to Sections 7.3.2 and 7.3.4) shall terminate, subject to any provision that has been expressly made by the Administrator, through a plan of reorganization or otherwise, for the survival, substitution, assumption, exchange or other continuation or settlement of such Award and provided that, in the case of Options and SARs that will not survive or be substituted for, assumed, exchanged, or otherwise continued or settled in the Change in Control Event, the holder of such Award shall be given reasonable advance notice of the impending termination and a reasonable opportunity to exercise his or her outstanding and vested Options and SARs (the vested portion of such Options and SARs determined after giving effect to any accelerated vesting required in the circumstances pursuant to Sections 7.3.2 and 7.3.4) in accordance with their terms before the termination of the Awards (except that in no case shall more than ten days' notice of accelerated vesting and the impending termination be required and any acceleration may be made contingent upon the actual occurrence of the event). For purposes of this Section 7.3, an Award shall be deemed to have been "assumed" if (without limiting other circumstances in which an Award is assumed) the Award continues after the Change in Control Event, and/or is assumed and continued by a Parent (as such term is defined in the definition of Change in Control Event) following a Change in Control Event, and confers the right to purchase or receive, as applicable and subject to vesting and the other terms and conditions of the Award, for each share of Common Stock subject to the Award immediately prior to the Change in Control Event, the consideration (whether cash, shares, or other securities or property) received in the Change in Control Event by the stockholders of the Corporation for each share of Common Stock sold or exchanged in such transaction (or the consideration received by a majority of the stockholders participating in such transaction if the stockholders were offered a choice of consideration); provided, however, that if the consideration offered for a share of Common Stock in the transaction is not solely the ordinary common stock of a successor corporation or a Parent, the Board may provide for the consideration to be received upon exercise or payment of the Award, for each share subject to the Award, to be solely ordinary common stock of the successor corporation or a Parent equal in Fair Market Value to the per share consideration received by the stockholders participating in the Change in Control Event.

7.3.4 Other Acceleration Rules. The Administrator may override the provisions of this Section 7.3 as to any Award by express provision in the applicable Award Agreement and may accord any Participant a right to refuse any acceleration, whether pursuant to the Award Agreement or otherwise, in such circumstances as the Administrator may approve. The portion of any Incentive Stock Option accelerated in connection with a Change in Control Event (or such other circumstances as may trigger accelerated vesting of the Incentive Stock Option) shall remain exercisable as an Incentive Stock Option only to the extent the applicable \$100,000 limitation on Incentive Stock Options is not exceeded. To the extent exceeded, the accelerated portion of the Option shall be exercisable as a Nonqualified Stock Option.

7.4 Termination of Employment or Services.

7.4.1 Events Not Deemed a Termination of Employment. Unless the Administrator otherwise expressly provides with respect to a particular Award, if a Participant's employment by or service to the Corporation or an Affiliate terminates but immediately thereafter the Participant continues in the employ of or service to another Affiliate or the Corporation, as applicable, the Participant shall be deemed to have not had a termination of employment or service for purposes of this Plan and the Participant's Awards. Unless the express policy of the Corporation or the Administrator otherwise provides, a Participant's employment relationship with the Corporation or any of its Affiliates shall not be considered terminated solely due to any sick leave, military leave, or any other leave of absence authorized by the Corporation or any Affiliate or the Administrator; provided that, unless reemployment upon the expiration of such leave is guaranteed by contract or law, such leave is for a period of not more than three months. In the case of any Participant on an approved leave of absence, continued vesting of the Award while on leave from the employ of or service with the Corporation or any of its Affiliates will be suspended until the Participant returns to service, unless the Administrator otherwise provides or applicable law otherwise requires. In no event shall an Award be exercised after the expiration of the term of the Award set forth in the Award Agreement.

- **Effect of Change of Affiliate Status**. For purposes of this Plan and any Award, if an entity ceases to be an Affiliate, a termination of employment or service will be deemed to have occurred with respect to each Eligible Person in respect of such Affiliate who does not continue as an Eligible Person in respect of another Affiliate that continues as such after giving effect to the transaction or other event giving rise to the change in status unless the Affiliate that is sold, spun-off or otherwise divested (or its successor or a direct or indirect parent of such Affiliate or successor) assumes the Eligible Person's award(s) in connection with such transaction.
- **7.4.3** Administrator Discretion. Notwithstanding the provisions of Section 5.6 or 6.8, in the event of, or in anticipation of, a termination of employment or service with the Corporation or any of its Affiliates for any reason, the Administrator may accelerate the vesting and exercisability of all or a portion of the Participant's Award, and/or, subject to the provisions of Sections 5.4.2 and 7.3, extend the exercisability period of the Participant's Option or SAR upon such terms as the Administrator determines and expressly sets forth in or by amendment to the Award Agreement.
- **7.4.4** Termination of Consulting or Affiliate Services. If the Participant is an Eligible Person solely by reason of clause (c) of Section 3, the Administrator shall be the sole judge of whether the Participant continues to render services to the Corporation or any of its Affiliates, unless a written contract or the Award Agreement otherwise provides.

7.5 Compliance with Laws.

7.5.1 General. This Plan, the granting and vesting of Awards under this Plan, and the offer, issuance and delivery of shares of Common Stock, the acceptance of promissory notes and/or the payment of money under this Plan or under Awards are subject to compliance with all applicable federal and state laws, rules and regulations (including but not limited to state and federal securities laws, and federal margin requirements) and to such approvals by any listing, regulatory or governmental authority as may, in the opinion of counsel for the Corporation, be necessary or advisable in connection therewith. The person acquiring any securities under this Plan will, if requested by the Corporation, provide such assurances and representations to the Corporation as the Administrator may deem necessary or desirable to assure compliance with all applicable legal and accounting requirements.

- 7.5.2 <u>Compliance with Securities Laws</u>. No Participant shall sell, pledge or otherwise transfer shares of Common Stock acquired pursuant to an Award or any interest in such shares except in accordance with the express terms of this Plan and the applicable Award Agreement. Any attempted transfer in violation of this Section 7.5 shall be void and of no effect. Without in any way limiting the provisions set forth above, no Participant shall make any disposition of all or any portion of shares of Common Stock acquired or to be acquired pursuant to an Award, except in compliance with all applicable federal and state securities laws and unless and until:
 - (a) there is then in effect a registration statement under the Securities Act covering such proposed disposition and such disposition is made in accordance with such registration statement;
 - (b) such disposition is made in accordance with Rule 144 under the Securities Act; or
 - (c) such Participant notifies the Corporation of the proposed disposition and furnishes the Corporation with a statement of the circumstances surrounding the proposed disposition, and, if requested by the Corporation, furnishes to the Corporation an opinion of counsel acceptable to the Corporation's counsel, that such disposition will not require registration under the Securities Act and will be in compliance with all applicable state securities laws.

Notwithstanding anything else herein to the contrary, neither the Corporation or any Affiliate has any obligation to register the Common Stock or file any registration statement under either federal or state securities laws, nor does the Corporation or any Affiliate make any representation concerning the likelihood of a public offering of the Common Stock or any other securities of the Corporation or any Affiliate.

7.5.3 Share Legends. All certificates evidencing shares of Common Stock issued or delivered under this Plan shall bear the following legends and/or any other appropriate or required legends under applicable laws:

"OWNERSHIP OF THIS CERTIFICATE, THE SHARES EVIDENCED BY THIS CERTIFICATE AND ANY INTEREST THEREIN ARE SUBJECT TO SUBSTANTIAL RESTRICTIONS ON TRANSFER UNDER APPLICABLE LAW AND UNDER AGREEMENTS WITH THE CORPORATION, INCLUDING RESTRICTIONS ON SALE, ASSIGNMENT, TRANSFER, PLEDGE OR OTHER DISPOSITION."

"THE SHARES ARE SUBJECT TO THE CORPORATION'S RIGHT OF FIRST REFUSAL TO REPURCHASE THE SHARES UNDER THE CORPORATION'S STOCK INCENTIVE PLAN AND AGREEMENTS WITH THE CORPORATION THEREUNDER, COPIES OF WHICH ARE AVAILABLE FOR REVIEW AT THE OFFICE OF THE SECRETARY OF THE CORPORATION."

"THE SHARES REPRESENTED BY THIS CERTIFICATE ARE SUBJECT TO A LOCK-UP PERIOD FOLLOWING THE EFFECTIVE DATE OF A REGISTRATION STATEMENT OF THE CORPORATION FILED UNDER THE SECURITIES ACT OF 1933, AS AMENDED ("ACT"), AS SET FORTH IN AN AGREEMENT BETWEEN THE CORPORATION AND THE ORIGINAL HOLDER OF THESE SHARES, A COPY OF WHICH MAY BE OBTAINED AT THE PRINCIPAL OFFICE OF THE CORPORATION. SUCH LOCK-UP PERIOD IS BINDING ON TRANSFEREES OF THESE SHARES."

"THE SECURITIES REPRESENTED HEREBY HAVE NOT BEEN REGISTERED OR QUALIFIED UNDER THE ACT, NOR HAVE THEY BEEN REGISTERED OR QUALIFIED UNDER THE SECURITIES LAWS OF ANY STATE. NO TRANSFER OF SUCH SECURITIES WILL BE PERMITTED UNLESS A REGISTRATION STATEMENT UNDER THE ACT IS IN EFFECT AS TO SUCH TRANSFER, THE TRANSFER IS MADE IN ACCORDANCE WITH RULE 144 UNDER THE ACT, OR IN THE OPINION OF COUNSEL TO THE CORPORATION, REGISTRATION UNDER THE ACT IS UNNECESSARY IN ORDER FOR SUCH TRANSFER TO COMPLY WITH THE ACT AND WITH APPLICABLE STATE SECURITIES LAWS."

- **7.5.4** <u>Confidential Information</u>. Any financial or other information relating to the Corporation obtained by Participants in connection with or as a result of this Plan or their Awards shall be treated as confidential.
- **Tax Withholding.** Upon any exercise, vesting, or payment of any Award or upon the disposition of shares of Common Stock acquired pursuant to the exercise of an Incentive Stock Option prior to satisfaction of the holding period requirements of Section 422 of the Code, the Corporation or any of its Affiliates shall have the right at its option to:
 - (a) require the Participant (or the Participant's Personal Representative or Beneficiary, as the case may be) to pay or provide for payment of at least the minimum amount of any taxes which the Corporation or Affiliate may be required to withhold with respect to such Award event or payment;
 - (b) deduct from any amount otherwise payable (in respect of an Award or otherwise) in cash to the Participant (or the Participant's Personal Representative or Beneficiary, as the case may be) the minimum amount of any taxes which the Corporation or Affiliate may be required to withhold with respect to such Award event or payment; or

(c) reduce the number of shares of Common Stock to be delivered by (or otherwise reacquire shares held by the Participant) the appropriate number of shares of Common Stock, valued at their then Fair Market Value, to satisfy the minimum withholding obligation.

In any case where a tax is required to be withheld in connection with the delivery of shares of Common Stock under this Plan, the Administrator may in its sole discretion (subject to Section 7.5) grant (either at the time of the Award or thereafter) to the Participant the right to elect, pursuant to such rules and subject to such conditions as the Administrator may establish, to have the Corporation reduce the number of shares to be delivered by (or otherwise reacquire) the appropriate number of shares, valued in a consistent manner at their Fair Market Value or at the sales price in accordance with authorized procedures for cashless exercises, necessary to satisfy the minimum applicable withholding obligation on exercise, vesting or payment. In no event shall the shares withheld exceed the minimum whole number of shares required for tax withholding under applicable law. The Corporation may, with the Administrator's approval, accept one or more promissory notes from any Eligible Person in connection with taxes required to be withheld upon the exercise, vesting or payment of any Award under this Plan; provided that any such note shall be subject to terms and conditions established by the Administrator and the requirements of applicable law. Any such note need not otherwise comply with the provisions of Section 5.3.3.

7.7 Plan and Award Amendments, Termination and Suspension.

- **7.7.1** Board Authorization. The Board may, at any time, terminate or, from time to time, amend, modify or suspend this Plan, in whole or in part. No Awards may be granted during any period that the Board suspends this Plan.
- 7.7.2 Stockholder Approval. To the extent then required by applicable law or any applicable listing agency or required under Sections 162, 422 or 424 of the Code to preserve the intended tax consequences of this Plan, or deemed necessary or advisable by the Board, any amendment to this Plan shall be subject to stockholder approval.

- **7.7.3** Amendments to Awards. Without limiting any other express authority of the Administrator under (but subject to) the express limits of this Plan, the Administrator by agreement or resolution may waive conditions of or limitations on Awards to Participants that the Administrator in the prior exercise of its discretion has imposed, without the consent of a Participant, and (subject to the requirements of Sections 2.2 and 7.7.4) may make other changes to the terms and conditions of Awards.
- **7.7.4** <u>Limitations on Amendments to Plan and Awards.</u> No amendment, suspension or termination of this Plan or amendment of any outstanding Award Agreement shall, without written consent of the Participant, affect in any manner materially adverse to the Participant any rights or benefits of the Participant or obligations of the Corporation under any Award granted under this Plan prior to the effective date of such change. Changes, settlements and other actions contemplated by Section 7.3 shall not be deemed to constitute changes or amendments for purposes of this Section 7.7.
- **7.8 Privileges of Stock Ownership.** Except as otherwise expressly authorized by the Administrator, a Participant will not be entitled to any privilege of stock ownership as to any shares of Common Stock not actually delivered to and held of record by the Participant. Except as expressly required by Section 7.3.1, no adjustment will be made for dividends or other rights as a stockholder for which a record date is prior to such date of delivery.
- **Stock-Based Awards in Substitution for Awards Granted by Other Corporation**. Awards may be granted to Eligible Persons in substitution for or in connection with an assumption of employee stock options, stock appreciation rights, restricted stock or other stock-based awards granted by other entities to persons who are or who will become Eligible Persons in respect of the Corporation or one of its Affiliates, in connection with a distribution, merger or other reorganization by or with the granting entity or an affiliated entity, or the acquisition by the Corporation or one of its Affiliates, directly or indirectly, of all or a substantial part of the stock or assets of the employing entity. The Awards so granted need not comply with other specific terms of this Plan, provided the Awards reflect only adjustments giving effect to the assumption or substitution consistent with the conversion applicable to the Common Stock in the transaction and any change in the issuer of the security. Any shares that are delivered and any Awards that are granted by, or become obligations of, the Corporation, as a result of the assumption by the Corporation of, or in substitution for, outstanding awards previously granted by an acquired company (or previously granted by a predecessor employer (or direct or indirect parent thereof) in the case of persons that become employed by the Corporation or one of its Affiliates in connection with a business or asset acquisition or similar transaction) shall not be counted against the Share Limit or other limits on the number of shares available for issuance under this Plan.
- **7.10** *Effective Date of the Plan*. This Plan is effective upon the Effective Date, subject to approval by the stockholders of the Corporation within twelve months after the date the Board approves this Plan.
- 7.11 *Term of the Plan*. Unless earlier terminated by the Board, this Plan will terminate at the close of business on the day before the 10th anniversary of the Effective Date. After the termination of this Plan either upon such stated expiration date or its earlier termination by the Board, no additional Awards may be granted under this Plan, but previously granted Awards (and the authority of the Administrator with respect thereto, including the authority to amend such Awards) shall remain outstanding in accordance with their applicable terms and conditions and the terms and conditions of this Plan.

7.12 Governing Law/Severability/Construction.

- **7.12.1** Choice of Law. This Plan, the Awards, all documents evidencing Awards and all other related documents will be governed by, and construed in accordance with, the laws of the state of Delaware.
- **7.12.2** Severability. If it is determined that any provision of this Plan or an Award Agreement is invalid and unenforceable, the remaining provisions of this Plan and/or the Award Agreement, as applicable, will continue in effect provided that the essential economic terms of this Plan and the Award can still be enforced.
- **7.12.3** Construction. It is intended that this Plan, and any Award under this Plan, will be exempt from, or comply with, Section 409A of the Code so as to not result in any tax, penalty or interest thereunder, and this Plan and each Award shall be construed and interpreted consistent with that intent.
- **7.13** *Captions*. Captions and headings are given to the sections and subsections of this Plan solely as a convenience to facilitate reference. Such headings will not be deemed in any way material or relevant to the construction or interpretation of this Plan or any provision thereof.
- **7.14 Non-Exclusivity of Plan.** Nothing in this Plan will limit or be deemed to limit the authority of the Board or the Administrator to grant awards or authorize any other compensation, with or without reference to the Common Stock, under any other plan or authority.
- No Restriction on Corporate Powers. The existence of this Plan, the Award Agreements, and the Awards granted hereunder, shall not limit, affect or restrict in any way the right or power of the Board or the stockholders of the Corporation to make or authorize: (a) any adjustment, recapitalization, reorganization or other change in the Corporation's or any Affiliate's capital structure or its business; (b) any merger, amalgamation, consolidation or change in the ownership of the Corporation or any Affiliate; (c) any issue of bonds, debentures, capital, preferred or prior preference stocks ahead of or affecting the Corporation's capital stock or the rights thereof; (d) any dissolution or liquidation of the Corporation or any Affiliate; (e) any sale or transfer of all or any part of the Corporation or any Affiliate's assets or business; or (f) any other corporate act or proceeding by the Corporation or any Affiliate. No Participant, Beneficiary or any other person shall have any claim under any Award or Award Agreement against any member of the Board or the Administrator, or the Corporation or any employees, officers or agents of the Corporation or any Affiliate, as a result of any such action.

7.16 Other Company Compensation or Benefit Programs. Payments and other benefits received by a Participant under an Award made pursuant to this Plan shall not be deemed a part of a Participant's compensation for purposes of the determination of benefits under any other employee welfare or benefit plans or arrangements, if any, provided by the Corporation or any Affiliate, except where the Administrator or the Board expressly otherwise provides or authorizes in writing. Awards under this Plan may be made in addition to, in combination with, as alternatives to or in payment of grants, awards or commitments under any other plans or arrangements of the Corporation or any Affiliate.

8. **DEFINITIONS**.

- "Administrator" has the meaning given to such term in Section 2.1.
- "Affiliate" means (a) any corporation (other than the Corporation) in an unbroken chain of corporations ending with the Corporation if, at the time of the determination, each of the corporations other than the Corporation owns stock possessing fifty percent (50%) or more of the total combined voting power of all classes of stock in one of the other corporations in such chain, or (b) any corporation (other than the Corporation) in an unbroken chain of corporations beginning with the Corporation if, at the time of the determination, each of the corporations other than the last corporation in the unbroken chain owns stock possessing fifty percent (50%) or more of the total combined voting power of all classes of stock in one of the other corporations in such chain.
- "Award" means an award of any Option, SAR or Stock Award, or any combination thereof, whether alternative or cumulative, authorized by and granted under this Plan.
- "Award Agreement" means any writing, approved by the Administrator, setting forth the terms of an Award that has been duly authorized and approved.
- "Award Date" means the date upon which the Administrator took the action granting an Award or such later date as the Administrator designates as the Award Date at the time of the grant of the Award.
- "Beneficiary" means the person, persons, trust or trusts designated by a Participant, or, in the absence of a designation, entitled by will or the laws of descent and distribution, to receive the benefits specified in the Award Agreement and under this Plan if the Participant dies, and means the Participant's executor or administrator if no other Beneficiary is designated and able to act under the circumstances.
- "Board" means the Board of Directors of the Corporation.
- "Cause" with respect to a Participant means (unless otherwise expressly provided in the applicable Award Agreement, or another applicable contract with the Participant that defines such term for purposes of determining the effect that a "for cause" termination has on the Participant's Awards) a termination of employment or service based upon a finding by the Corporation or any of its Affiliates, acting in good faith and based on its reasonable belief at the time, that the Participant:
- (a) has been negligent in the discharge of his or her duties to the Corporation or any Affiliate, has refused to perform stated or assigned duties or is incompetent in or (other than by reason of a disability or analogous condition) incapable of performing those duties;

- (b) has been dishonest or committed or engaged in an act of theft, embezzlement or fraud, a breach of confidentiality, an unauthorized disclosure or use of inside information, customer lists, trade secrets or other confidential information;
- (c) has breached a fiduciary duty, or willfully and materially violated any other duty, law, rule, regulation or policy of the Corporation or any of its Affiliates; or has been convicted of, or pled guilty or nolo contendere to, a felony or misdemeanor (other than minor traffic violations or similar offenses);
- (d) has materially breached any of the provisions of any agreement with the Corporation or any of its Affiliates;
- (e) has engaged in unfair competition with, or otherwise acted intentionally in a manner injurious to the reputation, business or assets of, the Corporation or any of its Affiliates; or
- (f) has improperly induced a vendor or customer to break or terminate any contract with the Corporation or any of its Affiliates or induced a principal for whom the Corporation or any Affiliate acts as agent to terminate such agency relationship.

A termination for Cause shall be deemed to occur (subject to reinstatement upon a contrary final determination by the Administrator) on the date on which the Corporation or any Affiliate first delivers written notice to the Participant of a finding of termination for Cause.

"Change in Control Event" means any of the following:

- (a) Approval by stockholders of the Corporation (or, if no stockholder approval is required, by the Board alone) of the complete dissolution or liquidation of the Corporation, other than in the context of a Business Combination that does not constitute a Change in Control Event under paragraph (c) below;
- (b) The acquisition by any individual, entity or group (within the meaning of Section 13(d)(3) or 14(d)(2) of the Exchange Act (a "Person")) of beneficial ownership (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of 50% or more of either (1) the thenoutstanding shares of common stock of the Corporation (the "Outstanding Company Common Stock") or (2) the combined voting power of the then-outstanding voting securities of the Corporation entitled to vote generally in the election of directors (the "Outstanding Company Voting Securities"); provided, however, that, for purposes of this paragraph (b), the following acquisitions shall not constitute a Change in Control Event; (A) any acquisition directly from the Corporation, (B) any acquisition by the Corporation, (C) any acquisition by any employee benefit plan (or related trust) sponsored or maintained by the Corporation or any Affiliate or a successor, (D) any acquisition by any entity pursuant to a Business Combination, (E) any acquisition by a Person described in and satisfying the conditions of Rule 13d-1(b) promulgated under the Exchange Act, or (F) any acquisition by a Person who is the beneficial owner (within the meaning of Rule 13d-3 promulgated under the Exchange Act) of 50% or more of the Outstanding Company Common Stock and/or the Outstanding Company Voting Securities on the Effective Date (or an affiliate, heir, descendant, or related party of or to such Person);

Consummation of a reorganization, merger, statutory share exchange or consolidation or similar corporate transaction involving the (c) Corporation or any corporation or other entity a majority of whose outstanding voting stock or voting power is beneficially owned directly or indirectly by the Corporation (a "Subsidiary"), a sale or other disposition of all or substantially all of the assets of the Corporation, or the acquisition of assets or stock of another entity by the Corporation or any of its Subsidiaries (each, a "Business Combination"), in each case unless, following such Business Combination, (1) all or substantially all of the individuals and entities that were the beneficial owners of the Outstanding Company Common Stock and the Outstanding Company Voting Securities immediately prior to such Business Combination beneficially own, directly or indirectly, more than 50% of the then-outstanding shares of common stock and the combined voting power of the then-outstanding voting securities entitled to vote generally in the election of directors, as the case may be, of the entity resulting from such Business Combination (including, without limitation, an entity that, as a result of such transaction, owns the Corporation or all or substantially all of the Corporation's assets directly or through one or more subsidiaries (a "Parent")), and (2) no Person (excluding any individual or entity described in clauses (C), (E) or (F) of paragraph (b) above) beneficially owns (within the meaning of Rule 13d-3 promulgated under the Exchange Act), directly or indirectly, more than 50% of, respectively, the then-outstanding shares of common stock of the entity resulting from such Business Combination or the combined voting power of the then-outstanding voting securities of such entity, except to the extent that the ownership in excess of 50% existed prior to the Business Combination;

provided, however, that a transaction shall not constitute a Change in Control Event if it is in connection with the underwritten public offering of the Corporation's securities.

"Code" means the Internal Revenue Code of 1986, as amended from time to time.

"Common Stock" means the shares of the Corporation's common stock, par value \$0.00001 per share, and such other securities or property as may become the subject of Awards, or become subject to Awards, pursuant to an adjustment made under Section 7.3.1 of this Plan.

"Corporation" means Cepton Technologies, Inc., a Delaware corporation, and its successors.

"Effective Date" means the date the Board approved this Plan.

"Eligible Person" has the meaning given to such term in Section 3 of this Plan.

"Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time.

"Fair Market Value," for purposes of this Plan and unless otherwise determined or provided by the Administrator in the circumstances, means as follows:

- (a) If the Common Stock is listed or admitted to trade on the New York Stock Exchange or other national securities exchange (the "Exchange"), the Fair Market Value shall equal the closing price of a share of Common Stock as reported on the composite tape for securities on the Exchange for the date in question, or, if no sales of Common Stock were made on the Exchange on that date, the closing price of a share of Common Stock as reported on said composite tape for the next preceding day on which sales of Common Stock were made on the Exchange. The Administrator may, however, provide with respect to one or more Awards that the Fair Market Value shall equal the closing price of a share of Common Stock as reported on the composite tape for securities listed on the Exchange on the last trading day preceding the date in question or the average of the high and low trading prices of a share of Common Stock as reported on the composite tape for securities listed on the Exchange for the date in question or the most recent trading day.
- (b) If the Common Stock is not listed or admitted to trade on a national securities exchange, the Fair Market Value shall be the value as reasonably determined by the Administrator for purposes of the Award in the circumstances.

The Administrator also may adopt a different methodology for determining Fair Market Value with respect to one or more Awards if a different methodology is necessary or advisable to secure any intended favorable tax, legal or other treatment for the particular Award(s) (for example, and without limitation, the Administrator may provide that Fair Market Value for purposes of one or more Awards will be based on an average of closing prices (or the average of high and low daily trading prices) for a specified period preceding the relevant date).

Any determination as to Fair Market Value made pursuant to this Plan shall be made without regard to any restriction other than a restriction which, by its terms, will never lapse, and shall be conclusive and binding on all persons with respect to Awards granted under this Plan.

"Incentive Stock Option" means an Option that is designated and intended as an "incentive stock option" within the meaning of Section 422 of the Code, the award of which contains such provisions (including but not limited to the receipt of stockholder approval of this Plan, if the award is made prior to such approval) and is made under such circumstances and to such persons as may be necessary to comply with that section.

- "Nonqualified Stock Option" means an Option that is not an "incentive stock option" within the meaning of Section 422 of the Code and includes any Option designated or intended as a Nonqualified Stock Option and any Option designated or intended as an Incentive Stock Option that fails to meet the applicable legal requirements thereof.
- "**Option**" means an option to purchase Common Stock granted under Section 5 of this Plan. The Administrator will designate any Option granted to an employee of the Corporation or an Affiliate as a Nonqualified Stock Option or an Incentive Stock Option.
- "Participant" means an Eligible Person who has been granted and holds an Award under this Plan.
- "Personal Representative" means the person or persons who, upon the disability or incompetence of a Participant, has acquired on behalf of the Participant, by legal proceeding or otherwise, the power to exercise the rights or receive benefits under this Plan by virtue of having become the legal representative of the Participant.
- "Plan" means this Cepton Technologies, Inc. Stock Incentive Plan, as it may hereafter be amended from time to time.
- **"Public Offering Date"** means the date the Common Stock is first registered under the Exchange Act and listed or quoted on a recognized national securities exchange.
- "Restricted Shares" or "Restricted Stock" means shares of Common Stock awarded to a Participant under this Plan, subject to payment of such consideration and such conditions on vesting (which may include, among others, the passage of time, specified performance objectives or other factors) and such transfer and other restrictions as are established in or pursuant to this Plan and the related Award Agreement, to the extent such remain unvested and restricted under the terms of the applicable Award Agreement.
- "Restricted Stock Award" means an award of Restricted Stock.
- "SAR" means a share appreciation right, representing the right, subject to the terms and conditions of the Plan and the applicable Award Agreement, to receive a payment, in cash and/or Common Stock (as specified in the applicable Award Agreement), equal to the excess of the Fair Market Value of a share of Common Stock on the date the SAR is exercised over the "base price" of the SAR, which base price shall be set forth in the applicable Award Agreement.
- "Securities Act" means the Securities Act of 1933, as amended from time to time.
- "Severance Date" with respect to a particular Participant means, unless otherwise provided in the applicable Award Agreement:
- (a) if the Participant is an Eligible Person under clause (a) of Section 3 and the Participant's employment by the Corporation or any of its Affiliates terminates (regardless of the reason), the last day that the Participant is actually employed by the Corporation or such Affiliate (unless, immediately following such termination of employment, the Participant is a member of the Board or, by express written agreement with the Corporation or any of its Affiliates, continues to provide other services to the Corporation or any Affiliate as an Eligible Person under clause (c) of Section 3, in which case the Participant's Severance Date shall not be the date of such termination of employment but shall be determined in accordance with clause (b) or (c) below, as applicable, in connection with the termination of the Participant's other services);

- (b) if the Participant is not an Eligible Person under clause (a) of Section 3 but is an Eligible Person under clause (b) thereof, and the Participant ceases to be a member of the Board (regardless of the reason), the last day that the Participant is actually a member of the Board (unless, immediately following such termination, the Participant is an employee of the Corporation or any of its Affiliates or, by express written agreement with the Corporation or any of its Affiliates, continues to provide other services to the Corporation or any Affiliate as an Eligible Person under clause (c) of Section 3, in which case the Participant's Severance Date shall not be the date of such termination but shall be determined in accordance with clause (a) above or (c) below, as applicable, in connection with the termination of the Participant's employment or other services);
- (c) if the Participant is not an Eligible Person under clause (a) or clause (b) of Section 3 but is an Eligible Person under clause (c) thereof, and the Participant ceases to provide services to the Corporation or any of its Affiliates as determined in accordance with Section 7.4.4 (regardless of the reason), the last day that the Participant actually provides services to the Corporation or such Affiliate as an Eligible Person under clause (c) of Section 3 (unless, immediately following such termination, the Participant is an employee of the Corporation or any of its Affiliates or is a member of the Board, in which case the Participant's Severance Date shall not be the date of such termination of services but shall be determined in accordance with clause (a) or (b) above, as applicable, in connection with the termination of the Participant's employment or membership on the Board).

"Stock Award" means an award granted under Section 6 of this Plan. A Stock Award may include: (a) Restricted Stock, stock bonuses, performance stock, stock units, phantom stock, dividend equivalents, or similar rights to purchase or acquire shares of Common Stock, whether at a fixed or variable price or ratio related to the Common Stock, upon the passage of time, the occurrence of one or more events, or the satisfaction of performance criteria or other conditions, or any combination thereof; or (b) any similar securities with a value derived from the value of or related to the Common Stock and/or returns thereon.

"Total Disability" means a "total and permanent disability" within the meaning of Section 22(e)(3) of the Code and, with respect to Awards other than Incentive Stock Options, such other disabilities, infirmities, afflictions, or conditions as the Administrator may include.

CEPTON, INC. CODE OF CONDUCT AND ETHICS

Cepton, Inc. and its subsidiaries (collectively, the "Company") believe that a strong commitment to high ethical, moral and legal principles in every aspect of the Company's business is essential for our success. Accordingly, we have adopted this Code of Conduct and Ethics (the "Code") that is applicable to each of the employees, officers and directors of the Company. We will make this Code available on our website at *www.cepton.com* and will disclose its availability in our annual report on Form 10-K and in our annual proxy statement.

Each of us is expected to read and abide by this document promptly upon receiving it. No written policy can definitively set forth the appropriate action for all business situations, but this Code is designed to establish basic principles for you to follow. Of course, the use of good judgment, coupled with a high sense of personal integrity, is always the best policy.

This Code should be read in conjunction with other policies applicable to an employee, officer or director of the Company. In addition to the ethical guidelines included in this Code, there are many laws and regulations that affect us in each of the markets where we do business. Each of us must comply with the requirements of all applicable law. If a law conflicts with a policy in this Code, you must comply with the law. Those who violate the law or standards of this Code will be subject to disciplinary action, up to and including immediate dismissal and termination of employment, and may be subject to substantial civil damages and criminal fines. We may also face substantial fines and penalties and may incur damage to our reputation and standing in the community. If you are in a situation which you believe may violate or lead to a violation of this Code, you should ask your supervisor or the Company's Chief Financial Officer, as the compliance officer for the Company (the "Compliance Officer"), how to handle the situation or follow the guidelines described in Section XVI below.

I. Personal Responsibilities of Employees, Officers and Directors

Each employee, officer and director must review this Code and sign the Employee Acknowledgment attached to this Code prior to or on his or her start date. All employees, officers and directors must review this Code carefully on an annual basis. All of us are expected to protect and enhance the assets and reputation of the Company. The honesty, integrity, sound judgment and professional and ethical conduct of the Company's employees, officers and directors is fundamental to the reputation, functioning and success of the Company.

Accordingly, in carrying out our duties, we must:

- Act with honesty and integrity, including the ethical handling of any actual or apparent conflict of interest between personal and professional relationships;
- Promote full, fair, accurate, timely and understandable disclosure in the reports and documents that the Company files with, or submits to, the Securities and Exchange Commission (the "SEC") and in other public communications made by the Company;
- Encourage and reward professional integrity in all aspects of our organization and eliminate barriers to responsible behavior, such as coercion, fear of reprisal or alienation from the Company;
- Provide for the education of all members of the Company about federal, state and local laws, rules and regulations and any applicable stock exchange rules relevant to the performance of their duties;
- Comply and take all reasonable actions to cause the Company to comply with applicable governmental laws, rules and regulations and any applicable stock exchange rules;
- Promptly report violations of this Code, including any violations of governmental laws, rules or regulations, and any applicable stock exchange rules, to the Compliance Officer or any other appropriate person identified in this Code;
- Ensure accountability for adherence to this Code; and
- Promote ethical and honest behavior in the workplace.

II. Conflicts of Interest

A "conflict of interest" exists when a person's private interests interfere—or appear to interfere—in any way with the interests of the Company. A conflict situation can arise when an employee, officer or director takes actions or has interests that may make it—or appear to make it—difficult to perform his or her Company work objectively and effectively. Conflicts of interest may also arise when an employee, officer or director, or members of his or her family, receives improper personal benefits, including material gifts or favors, as a result of his or her position in the Company. Loans to, or guarantees of obligations of, employees, officers, directors and their family members may create conflicts of interest.

Conflicts of interest may not always be apparent, so if you have a question regarding whether a particular situation is a conflict of interest, you should consult with your supervisor or the Compliance Officer. Any employee, officer or director who becomes aware of a conflict or potential conflict should bring it to the attention of a supervisor, manager or other appropriate personnel or follow the guidelines described in Section XVI below. Above all, the consistent use of good judgment by employees, officers and directors will help the Company avoid and prevent conflicts of interest.

This Code does not attempt to describe all possible conflicts of interest that could develop, but some of the more common conflicts from which employees, officers and directors must refrain are:

- Assisting a Competitor or Competing Against the Company. An obvious conflict of interest is providing assistance to an organization that provides services in competition with the current or proposed services of the Company. Without the consent of the Board of Directors of Cepton, Inc. (the "Board"), employees, officers and employee-directors, during their employment with the Company, may not: (a) work for such an organization as an employee, consultant or member of its board of directors; or (b) have any ownership interest (excluding indirect interests through broad-based mutual funds or similar investment funds in which you exercise no discretion as to the choice of investment), in any enterprise which competes with any business of the Company, except, solely in the case of a publicly-traded company, as a holder of less than one percent (1%) of the publicly traded stock in such a company. For the avoidance of doubt, non-employee directors are not subject to the foregoing restrictions in paragraphs (a) and (b). Such activities are prohibited because they divide your loyalty between the Company and that organization. If you are an employee, officer or employee-director and your investments in such organizations increases to more than the one percent (1%) threshold as specified above, you must promptly report these investments in writing to the Compliance Officer. You may also not market services in competition with the Company's current or proposed services. It is your responsibility to consult with your supervisor to determine whether any planned activity will compete with any of the Company's actual or proposed product lines or services before you pursue the activity.
- **B. Supplying the Company.** Generally, you may not be a supplier, vendor or business partner of the Company or represent or work for a supplier, vendor or business partner of the Company while you work for the Company. In addition, you may not accept money or benefits of any kind for any advice or services you may provide to a supplier, vendor or business partner in connection with its business with the Company, other than in accordance with the policies outlined in Section II. H. and Section IX of this Code.
- C. Someone Close to You Working in the Industry. You may find yourself in a situation where your spouse, another member of your immediate family, or someone else you are close to is employed by a competitor, supplier, vendor or business partner of the Company. Such situations call for extra sensitivity to security, confidentiality and conflicts of interest because the closeness of the relationship might lead you to inadvertently compromise the interests of the Company. There are several factors to consider in assessing such a situation, including the relationship between the Company and the other company, the nature of your responsibilities as a Company employee and those of the person close to you, and the access each of you has to your respective employer's confidential information. The very appearance of a conflict of interest can create problems, regardless of the behavior of the Company employee involved. You should review your specific situation with your supervisor to assess the nature and extent of any concern and how it can be resolved.

- D. Loans, Guarantees and Other Personal Financial Transactions. The Company will not make any loans to, or guarantee any personal obligations of, employees, officers, directors or their respective family members. You should not enter into any material personal financial transactions with the Company except Company-related corporate transactions (such as a cashless exercise of stock options) or other transactions that have been cleared with the Company's Chief Executive Officer (the "CEO") or the Compliance Officer. You also should not obtain a material loan or guarantee of personal obligations from, or enter into any other material personal financial transaction with, any organization which competes with any business of the Company or is a material customer, supplier, vendor or business partner of the Company. This guideline does not prohibit arm's-length or normal-course transactions with banks, brokerage firms or other financial institutions.
- **E. Family or Related Businesses.** Potential transactions with family businesses or other businesses in which you participate as an owner, partner, director, officer, employee, consultant or shareholder, and which may create a conflict of interest or may interfere with your duties to the Company must be disclosed in writing to the Compliance Officer for approval. The appearance of favoritism, potential for conflict and likelihood of discouraging other service providers in the future will be considered carefully by the Compliance Officer before the transaction is approved.
- **F. Improper Conduct and Activities.** You may not engage in any conduct or activities that are inconsistent with the Company's best interests or that disrupt or impair the Company's or its clients' relationship with any person or entity with which the Company has or proposes to enter into a business or contractual relationship.
- **G. Compensation from Non-Company Sources.** You may not accept compensation, in any form, for services performed for the Company from any source other than the Company. This provision is not intended to prevent members of our Board from having outside employment. However, each director must disclose to the Board any current or contemplated outside employment or similar relationships for the purpose of ensuring that no potential conflict of interest exists.
- **H. Gifts, Entertainment and Improper Payments.** You and members of your family may not request or accept from, or offer, promise or give to, any person or entity dealing or desiring to deal with the Company, any payments, gifts, entertainment, services, travel, lodging, or any other forms of compensation for personal benefit. For these purposes, a "gift" generally does not include articles of nominal value ordinarily used for sales promotion, and "entertainment" does not include ordinary business meals or reasonable entertainment (including tickets for sports, concerts or similar events) considered to be a normal part of a business relationship that is accepted only on a very infrequent basis and that does not detract or have the appearance of detracting from the integrity of the relationship. Participation in entertainment activities should be with a representative of the vendor, supplier or other entity in attendance at the activity. Giving or receiving any payment or gift in the nature of a bribe, kickback or other improper influence is absolutely prohibited.

III. Responsibilities of the CEO and Senior Financial Officers

In addition to the matters set forth in the remainder of this Code, the CEO and senior financial officers are subject to the following additional specific policies:

- **A. Disclosure.** The CEO and all senior financial officers are responsible for full, fair, accurate, timely and understandable disclosure in the periodic reports or documents required to be filed by the Company with the SEC. Accordingly, it is the responsibility of the CEO and each senior financial officer to promptly bring to the attention of the Compliance Officer and the Company's Chief Financial Officer (the "<u>CFO</u>") and to the Audit Committee any material information of which he or she may become aware that affects the disclosures made by the Company in its public filings or otherwise assist the Compliance Officer and the CFO in fulfilling the responsibilities specified in the Company's applicable policies.
- **B. Internal Controls; Fraud.** The CEO and each senior financial officer shall promptly bring to the attention of the Compliance Officer and the CFO and to the Audit Committee any information he or she may have concerning (a) significant deficiencies in the design and operation of internal controls which could adversely affect the Company's ability to record, process, summarize and report financial data or (b) any fraud, whether or not material, that involves management or other employees who have a significant role in the Company's financial reporting, disclosures or internal controls.
- **C. Code of Conduct.** The CEO and each senior financial officer shall promptly bring to the attention of the Compliance Officer and the CFO and to the Audit Committee any information he or she may have concerning any violation of this Code, including any actual or apparent conflicts of interest between personal and professional relationships, involving any management or other employees who have a significant role in the Company's financial reporting, disclosure or internal controls.
- **D. Violations.** The CEO and each senior financial officer shall promptly bring to the attention of the Compliance Officer and the CFO and to the Audit Committee any information he or she may have concerning evidence of a material violation of the securities or other laws, rules or regulations applicable to the Company and the operation of its business, by the Company or any agent thereof, or of violation of this Code.

IV. Insider Trading

Employees, officers and directors who have access to confidential information are not permitted to use or share that information for stock trading purposes except in the conduct of our business. All non-public information about the Company, its customers, vendors, suppliers, business partners and other firms the Company may be negotiating major transactions with, should be considered confidential information. To use non-public information for personal financial benefit or to "tip" others who might make an investment decision on the basis of this information is not only unethical, but also illegal. In order to assist in compliance with laws against insider trading, the Company has adopted a specific policy governing employees', officers' and directors' trading in securities of the Company. This insider trading policy has been distributed to every employee, officer and director of the Company. If you have any questions, please consult the Compliance Officer.

V. Confidentiality; Privacy and Protection of Personally Identifiable Information

Employees, officers and directors must maintain the confidentiality of confidential information (including letters, memos and internal Company documents) entrusted to them by the Company and people with whom the Company interacts, including its vendors, suppliers, business partners and customers, except when disclosure is authorized by the Compliance Officer or required by laws or regulations. Confidential information includes all non-public information that might be of use to competitors, harmful to the Company and its vendors, suppliers, business partners and customers, or otherwise subject to restrictions on disclosure by law. If an employee, officer or director leaves the Company, he or she must return to the Company all documents containing confidential information of the Company, whether in hard copy or electronic form, including without limitation all reports, information, and other materials relating to the Company's pipelines and research projects. Employees, officers and directors are prohibited from copying, preserving, falsifying, deleting without authorization, transferring, or sending relevant reports, information, document materials or processes for their personal or other benefit. This prohibition and the obligation to preserve confidential information continues even after employment ends.

In addition, customers and other people with whom the Company comes in contact entrust the Company with their personal information, which is defined as any piece of personally identifiable information, including, but not limited to, such personal information as name, mailing address, e-mail address, phone number, credit card number, driver's license number, social security number and information about health, race, religious beliefs, etc. The Company collects, processes, uses and retains personal information only in compliance with applicable laws. Employees, directors and officers are responsible for protecting this information both inside and outside of the Company and will do so in accordance with applicable laws, any data protection policies established by the Company and any extra standards required by contract. If personal information is to be transferred outside of any country, check to determine the necessity of a cross-border data transfer agreement. Internally, such information may only be disclosed for business purposes in accordance with applicable privacy laws. Externally, this information may not be disclosed.

VI. Corporate Opportunities

Employees, officers and directors are prohibited from taking for themselves, personally, opportunities that properly belong to the Company or are discovered through the use of corporate property, information or position without the consent of the Board. No employee, officer or director may use corporate property, information or position for improper personal gain. Employees, officers and directors owe a duty to the Company to advance its legitimate interests when the opportunity to do so arises.

VII. Books and Records

The Company is a publicly-owned company. As such, the Company relies on the public securities markets for capital to fund our activities. The Company's books and records are crucial to its business and form the basis of its financial statements and reports and other disclosures to shareholders and the public. Public investors rely upon the quality and integrity of the Company's financial statements, reports, press releases and other disclosures. Additionally, the Company's books and records are a source of essential data that guide the business decision-making and strategic planning of the Company. Accordingly, it is imperative that the Company maintain accurate books and records and report its financial results and condition accurately.

- A. Accuracy and Completeness. Each employee, officer and director must do his or her part to ensure that the books of account and financial records of the Company meet the highest standards of accuracy and completeness. This responsibility does not rest exclusively with the Company's accounting personnel. There is never an acceptable reason to make false or misleading entries. Undisclosed or unrecorded funds, payments or receipts are strictly prohibited. If you have reason to believe that any of the Company's books and records are not being maintained in an accurate or complete manner, or if you become aware of any breakdown in internal controls or in the Company's filings or other communications, you are expected to report this immediately to your supervisor, the Company's compliance and ethics hotline, as will be established by the Company, or to the Compliance Officer or the CFO. Similarly, the Company relies on you to speak up if you ever feel or if you actually are being pressured to destroy documents which would not normally be destroyed, or if you become aware that any misleading, incomplete or false statement was made to an accountant, auditor, attorney or government official in connection with any audit, examination or filing with a government agency, such as the SEC.
- **B. Financial Statements and Accounts.** All employees who are involved in supplying any kind of supporting documentation, determining account classification or approving transactions will be held responsible for ensuring that transactions are reported accurately, completely and in reasonable detail. All transactions must be recorded appropriately to ensure full accountability for all assets and activities of the Company and to supply the data needed in connection with the preparation of financial statements. Each employee involved in the preparation of the Company's financial statements must prepare them in accordance with generally accepted accounting principles in the United States ("GAAP") and other applicable standards and rules, so that the statements fairly and completely reflect the operations and financial condition of the Company.

- C. Payments and Expense Accounts. No payment on behalf of the Company may be approved or made if any part of it is to be used for any purpose other than that described by the supporting documents. All receipts and disbursements must be fully and accurately described in the books and records of the Company and must be supported by appropriate descriptive documentation. Employees should request reimbursement for business-related expenses in strict accordance with the Company's travel and entertainment policies.
- **D. Retention of Documents.** Documents and records must be retained for the periods of time specified by the Company. Each employee, officer and director is responsible for understanding and complying with the Company's document and record retention guidelines. Also, if you are aware of an imminent or ongoing investigation, audit or examination initiated by the Company or any government or regulatory agency, you should retain all documents and records in your custody or control relating to the matter under review, even if the age of the records is beyond the Company's retention guidelines. Please note that the knowing destruction or deliberate falsification of any document or data in order to impede a governmental or regulatory investigation, audit or examination may be the basis for immediate dismissal and termination of employment and may subject you to prosecution for obstruction of justice. If you are not sure that a document can be destroyed, consult your supervisor or the Compliance Officer.

VIII. Treatment of Employees

The Company has established comprehensive programs to ensure compliance with labor and employment laws, including equal employment opportunity policies and procedures, safety and health programs, and wage and hour procedures. For further information on particular labor and employment compliance policies, see the applicable sections of the Company's Employee Handbook or contact the Human Resources Department.

- A. Safety, Health and the Environment. The Company is committed to maintaining a safe work environment by eliminating recognized hazards in the workplace. The Company will, and you are required to, comply with all applicable environmental, health and safety ("EHS") laws and regulations and with all related Company policies. The Company is committed to maintaining safe workplaces, reducing the environmental impact of our operations and encouraging environmentally sustainable business operations. This also applies to our vendors, suppliers and business partners who acknowledge our standards and commitments. Employees, officers and directors are responsible for adhering to the applicable EHS requirements and for taking the necessary precautions to protect themselves and their colleagues. In addition, you must report to work in a condition suitable for performing your duties, free from the influence of illegal drugs or alcohol. To further your own safety and that of your fellow employees, you are also required to report to your supervisor any unsafe conditions, hazards, broken equipment, accidents, violence and employees visibly under the influence of illegal drugs or alcohol.
- B. Nondiscrimination and Equal Employment Opportunity. The Company is an equal opportunity employer that embraces and values a diverse workforce and pledges to provide equal employment opportunity to all job applicants and employees. We will not tolerate discrimination against applicants or employees based on any protected classification including without limitation their race, religion, color, gender, pregnancy, childbirth, or pregnancy-related conditions, age, marital status, national origin or ancestry, sexual orientation, citizenship status, disability or status as a disabled veteran. We require all employees to refrain from unlawful discrimination in any aspect of employment, including decisions concerning recruitment, hiring, termination, promotions, salary treatment, or any other term, condition or privilege of employment or career development. You must treat all employees, vendors, suppliers, business partners, contractors and customers of the Company and other people with whom we come in contact with respect, dignity and honesty. We will not tolerate the use of discriminatory slurs, or any other remarks, jokes or conduct or other forms of harassment based on race, religion, color, gender, age, marital status, national origin or ancestry, sexual orientation, citizenship status, disability, status as a disabled veteran, or any other protected status.
- C. Sexual and Other Forms of Harassment. The Company's policy, as well as the applicable laws, strictly prohibits all forms of harassment, violence and threatening behavior in the workplace, including unlawful harassment based on race, religion, color, gender, pregnancy, childbirth, or pregnancy-related conditions, age, marital status, national origin or ancestry, sexual orientation, citizenship status, disability, status as a disabled veteran or any other protected class. This prohibition also applies to the harassment of Company employees by non-employees. Sexual harassment includes linking, either explicitly or implicitly, a person's submission to, or rejection of, sexual advances to any decision regarding that person's terms or conditions of employment. It also includes the existence of a sexually abusive or hostile working environment. This can be characterized by unwelcome sexual advances or verbal or physical conduct of a sexual nature. The Company prohibits retaliation against individuals who report suspected violations or who participate in the investigation of any complaints of sexual or other forms of harassment.

D. Reporting Responsibilities and Procedures. If you believe that you have been subjected to harassment of any kind or any other type of unlawful discrimination, or if you feel retaliated against because of filing a complaint, you should report the matter to your supervisor, the Human Resources Department, the Company's compliance and ethics hotline, as will be established by the Company, or a corporate officer. Any manager who has knowledge of any incident of sexual or other prohibited forms of harassment is required to report such information to the Human Resources Department or a corporate officer. Complaints of harassment, abuse or discrimination will be taken seriously and investigated immediately and thoroughly by Human Resources. Investigations will be conducted as confidentially as possible. Employees found to have engaged in sexual harassment or any other kind of abusive behavior shall be subject to disciplinary action, which may include termination. However, false accusations of harassment also violate the Company's policy.

IX. Business Partner Relations

The Company seeks to do business only with vendors, suppliers and other business partners who conduct business ethically and legally. You must use care and good judgment in selecting and maintaining relationships with all of the Company's vendors, suppliers and other business partners. Employees, officers or directors who participate in the selection of any vendor, supplier or other business partner must use a selection process that is fair, lawful, does not improperly discriminate, and complies with all Company policies and procedures, and ensure that vendors, suppliers and other business partners are apprised of their obligation to abide by the Company's standards of business documents.

X. Computer Use

All Internet, Intranet and e-mail activities are to be conducted for legitimate business purposes only. You acknowledge that, to the extent permitted by the applicable laws, the Company owns and has all rights to monitor, inspect, disclose and expunge all electronic files and records on Company systems, and that you should have no expectation of privacy with respect to all such files and records. Employee use and/or access of all Company computing resources, including computers, mobile phones and tablets, networked services and Internet and e-mail access (including web surfing), must at all times comply with all Company policies (including any policy regarding Internet use as may be set forth in the Company's Employee Handbook) and applicable laws, including those relating to intellectual property, privacy, defamation (libel and slander) and unfair competition. You are reminded that all on-line, social media and e-mail activities, intentionally or not, are potentially public in nature. We must never act in a way that would bring liability, loss of credibility or embarrassment to the Company. Adhering to these guidelines with respect to computer use is very important, as violation may result in significant civil and even criminal penalties for both you and the Company.

XI. Intellectual Property

Each employee, officer and director has an obligation to protect the Company's intellectual property. These "intellectual properties" may not be tangible like our buildings or equipment, but they are among the most valuable of the Company's assets. Our patents, logos, brand, copyrights, software, know-how and trademarks are examples of the Company's intellectual property. We protect all of our intellectual property rights — even those that are not patentable or protected by copyright or trademark laws — to the fullest extent permitted by law. In keeping with the Company's high standards for quality and ethical conduct, each of us must protect the value of the intellectual property of the Company by protecting their confidentiality and safeguarding them from theft, infringement or misuse, including by using Company trademarks and brand resources properly and consistently. Furthermore, all employees, officers and directors shall maintain the secrecy of innovations for which the Company will seek or is seeking patent or trade secret protection. You must advise senior management or the Compliance Officer of infringements by others or if you are unsure about a proposed use of Company intellectual property or any other materials for public dissemination. In addition to the guidelines included in this Code, employees and officers are subject to the obligations provided by any proprietary information and inventions assignment agreements that they may enter into with the Company.

XII. Protection of Other Company Assets; Proprietary Information

You must endeavor to protect the Company's assets and property and ensure their efficient use. Our assets include, among other things, our office equipment, technology and information. Theft, carelessness and waste have a direct impact on our profitability. You must use all assets and property of the Company for legitimate business purposes only. You must report any suspected incident of fraud or theft immediately to your supervisor or the Compliance Officer for investigation. If you are an officer or director you must report such fraud or theft to the Board or a committee of the Board.

To protect the security of Company information, the Company reserves the right to restrict personal use of e-mail and Internet services. Such restrictions include website restrictions, size and type of files (for example, photos, music or video), or the dissemination of spam messages. All software used on Company computers must be legal, licensed and in compliance with Company policies. Employees are prohibited from installing any non-compliant software without the authorization of the IT Department.

Your obligation to protect our assets includes the Company's proprietary information. Proprietary information includes, but is not limited to, business, marketing and strategic plans, research and development strategies, customer and mailing lists, supplier, vendor and business partner names and pricing, records, salary information, intellectual property, information management system, any unpublished financial data and reports and any unique products, processes or information the Company has developed. Unauthorized use or distribution of this information violates this Code and may subject you to civil or criminal penalties.

XIII. Anti-Bribery and Anti-Corruption and Anti-Money Laundering; Political Contributions

The Company will not tolerate bribery or any form of corruption. Payments or promises to pay something of value to obtain or retain business or otherwise secure an improper advantage must never be made to a government official, employee or parties with whom the Company maintains a business and/or contractual relationship. The Company conducts its business in a manner consistent with foreign corrupt practices and/or any applicable governmental laws, rules and regulations. If you have any questions regarding this policy, you should contact the Compliance Officer.

A. Payments to Government Officials. The U.S. government has a number of laws and regulations regarding business gratuities that may be accepted by U.S. government personnel. The promise, offer or delivery to any official or employee of the U.S. government of a gift, favor or other gratuity on behalf of the Company, directly or indirectly, in violation of these rules would violate Company policy and could be a criminal offense. Foreign, state and local governments may have similar rules that you must also comply with. Employees, directors and officers may only transact business on behalf of the Company in foreign markets and with foreign government officials in accordance with the Company's established policies regarding foreign corrupt practices and/or any applicable governmental laws, rules and regulations.

This policy is not intended to curtail your freedom to support political candidates and causes within legal limits, but care should be exercised so that no action by you is perceived as an attempt to influence government decisions in matters affecting the Company. Any personal contribution to any political candidate, party or organization must not be represented as a contribution from the Company, and you should not identify employment by the Company in connection with your individual political activities.

The Company prohibits facilitation payments, which are small value payments made to a government official, in order to secure or speed up the performance of a routine or expected governmental action to which the payer is entitled. All employees are expressly prohibited from offering, promising, accepting, giving, soliciting, receiving or authorizing, a bribe or facilitation payment.

- **B.** Third-Party Risk. All employees must take reasonable precautions to ensure that all third parties they engage have solid qualifications, charge customary fees, have no apparent conflicts of interest, and are willing to sign a written contract that includes a statement that they will not make payments prohibited by law. Third parties include suppliers, agents, brokers, consultants, and other business partners. All employees should exercise due care in selecting such business partners to ensure they are reputable, honest, and qualified for their roles, and in monitoring their activity once selected. All employees must follow any mandatory due diligence procedures adopted by the Company in regards to third-party intermediaries.
- Counterparty Risk. The Company requires the employees to conduct a reasonable level of due diligence on all counterparties and proposed transactions in order to detect risks associated with foreign corrupt practices and/or any applicable governmental laws, rules and regulations. Prior to beginning a relationship with any counterparties, such as vendors, suppliers or business partners, the Company will conduct due diligence checks to identify, among others, the potential involvement of (i) trade controls targets; (ii) embargoed and sanctioned territories; and (iii) any indication of illegal or other illicit activities. Prior to entering into any transactions with new or existing counterparties, the Company will screen the names of any guarantors, intermediaries, or banks involved in the proposed transaction(s) against the applicable trade control lists.

XIV. Disclosure Obligations and Procedures

The Company's policy is to release data of public importance, including reports to be filed with the SEC, at the earliest appropriate time consistent with the need to both maintain confidentiality of information before final decisions are made and to avoid endangering the Company's business through disclosure of information potentially advantageous to competitors. All public statements, oral or written, must be accurate, with no material omissions. Information that could reasonably be expected to have an impact on the market for Company securities may be released only through the Company's Authorized Spokespersons (as defined in the Company's Disclosure Policy). All inquiries from financial analysts, media representatives, or financial consultants should also be directed to the Compliance Officer, the Public Relations Department or the CFO. Financial information and results, including projections, forecasts, and forward looking statements, should not be supplied in business proposals, presentations or advertising or presented to the press or released to the media without express prior approval and review by the CFO and the Public Relations Department. For further information on disclosure obligations and procedures, see the Company's Disclosure Policy.

Social media, such as Twitter, Facebook and LinkedIn, brings opportunities to build conversations and communities, but also carries certain risks. All employees, officers and directors of the Company must therefore engage in social media conduct in a cautious and careful manner and only the Company's Public Relations Department may represent the Company in social media. In addition, employees may not engage in social media postings that are inconsistent with the Company's confidentiality policies. The use of any social network, blog or public website to post non-public information related to the Company is prohibited without the Company's authorization.

XV. Amendments and Waivers of this Code

From time to time, the Company may amend certain provisions of this Code. Waivers of this Code for executive officers and directors of the Company must be granted by the Board. Waivers of this Code for all other employees of the Company may be granted only by the Board or the Audit Committee. Amendments to and waivers of this Code will be publicly disclosed as required by applicable laws and regulations.

XVI. Compliance with this Code

The Company takes this Code very seriously. You must understand this Code and take responsibility for seeking the advice of your supervisor or other appropriate officials of the Company if you need clarification on any point. Employees who have questions about this Code or wish to make an anonymous report of any suspected or actual violations of this Code are also encouraged to call the Company's confidential compliance and ethics hotline, as will be established by the Company.

- A. Reporting Violations. If you believe you have violated this Code or any applicable laws or regulations, you must report the violation so that the Company can take appropriate action. The fact that you reported the violation will be given consideration in determining appropriate disciplinary action, if any. In many cases, a prompt report of a violation can substantially reduce the adverse impact of a violation on all involved, on third parties, on the Company and on you. If you become aware that another employee, of whatever level of seniority, has, in all likelihood, violated this Code, including any law or regulation applicable to the Company's businesses, you are under a duty to report that violation, in order that the Company can take steps to rectify the problem and prevent a recurrence. Violations may be reported to your supervisor, the Human Resources Department, Compliance Officer or the CFO, as appropriate. Such reports will be treated confidentially to the extent possible, and no person will be subject to retaliation for reporting a suspected violation in good faith. Employees are expected to cooperate in internal investigations of misconduct and the Company prohibits retaliation against individuals who cooperate with the investigation of any complaints. Any employee may submit a good-faith concern regarding questionable accounting or auditing matters without fear of dismissal or retaliation of any kind.
- **B. Disciplinary Action.** The Company intends to use every reasonable effort to prevent the occurrence of conduct not in compliance with this Code and other Company policies, including those outlined in the Company's Employee Handbook, and to halt any such conduct that may occur as soon as reasonably possible after its discovery. Company employees and officers who violate this Code and other Company policies and procedures may be subject to disciplinary actions, which may include termination. In addition, disciplinary measures will apply to anyone who directs or approves infractions, or has knowledge of them and does not move promptly to correct them in accordance with Company policies. In addition, persons who violate the law during the course of their employment may be subject to criminal and civil penalties, as well as payment of civil damages to others.

The Board shall determine, or designate appropriate persons to determine, appropriate actions to be taken in the event of violations of this Code by the CEO or the Company's senior financial officers. Such actions shall be reasonably designed to deter wrongdoing and to promote accountability for adherence to this Code and shall include written notices to the individuals involved that the Board has determined that there has been a violation, censure by the Board, demotion or re-assignment of the individual involved, suspension with or without pay or benefits (as determined by the Board) and termination of the individual's employment. In determining what action is appropriate in a particular case, the Board or such designee shall take into account all relevant information, including the nature and severity of the violation, whether the violation appears to have been intentional or inadvertent, whether the individual in question has been advised prior to the violation as to the proper course of action and whether or not the individual in question had committed other violations in the past.

* * *

Cepton, Inc. (the "Company") Code of Conduct and Ethics

Employee Acknowledgement

I have received a copy and had an opportunity to read the Code of Conduct and Ethics (the "<u>Code</u>") attached. I understand that I may raise any questions I might have concerning the Code to my supervisor or the Company's Compliance Officer. I also understand that it is my responsibility to comply with the Code, including the procedures described in the section entitled "Compliance with this Code—Reporting Violations," and any revisions made to the Code.

	is Acknowledgement form. The signed copy of this Acknowled Acknowledgement form along with the Code for your records.	gment form will be retained in your
Signature of Employee	Date	
Employee's Name (Printed)		
	10	

February 10, 2022

Securities and Exchange Commission 100 F Street, N.E. Washington, DC 20549

Commissioners:

We have read the statements made by Cepton, Inc. (formerly Growth Capital Acquisition Corp.) under Item 4.01 of its Form 8-K dated February 10, 2022. We agree with the statements concerning our Firm in such Form 8-K; we are not in a position to agree or disagree with other statements of Cepton, Inc. (formerly Growth Capital Acquisition Corp.) contained therein.

Very truly yours,

/s/Marcum LLP

Marcum LLP

LIST OF SUBSIDIARIES OF CEPTON, INC.

Name of Subsidiary	Jurisdiction of Organization
Cepton Technologies, Inc.	Delaware



Cepton Technologies and Growth Capital Acquisition Corp. Announce Closing of Business Combination

SAN JOSE, CA, February 10, 2022 – Cepton Technologies, Inc. ("Cepton"), a Silicon Valley innovator and leader in high-performance MMT® lidar solutions and Growth Capital Acquisition Corp. ("GCAC") (Nasdaq: GCAC), announced today the completion of their previously announced business combination. The combined company has been renamed Cepton, Inc. and its common stock and warrants are expected to commence trading on the Nasdaq Capital Market under the new ticker symbols "CPTN" and "CPTNW", respectively, on Friday, February 11, 2022. The business combination was approved at a special meeting of GCAC's stockholders on Wednesday, February 9, 2022.

"Today is a special milestone for Cepton," said Dr. Jun Pei, Cepton's Co-Founder and CEO. "We founded Cepton with the goal of enabling safe and autonomous transportation for everyone, and today's announcement is another step in making that vision a reality. As a public company we remain focused on the goals of mass-market lidar adoption in consumer vehicles and expanding our market leadership position, while driving value for all of Cepton's stakeholders."

George Syllantavos, co-CEO of GCAC, added "We are thrilled with the successful completion of this business combination. Cepton is a company at the forefront of technological advancements with the largest known ADAS lidar series production award in the industry and ongoing engagements with all of the Top 10 global OEMs. We are excited about the future and continuing to support the company in its public phase."

"I am impressed by the elegant technical solutions that Cepton is bringing to the lidar industry in addressing automotive ADAS needs. I'm certain that Cepton's solutions will gain further commercial success in automotive and other markets. Time works in Cepton's favor now that its public life has began, and its vision is on track to materialize", said Akis Tsirigakis, co-CEO of GCAC.

The combined company will continue to be led by Dr. Jun Pei alongside the rest of the current Cepton management team. George Syllantavos, co-CEO of GCAC, will serve as a director of Cepton.

About Cepton Technologies, Inc.

Cepton is a Silicon Valley innovator of lidar-based solutions for automotive (ADAS/AV), smart cities, smart spaces and smart industrial applications. With its patented Micro Motion Technology (MMT®), Cepton aims to take lidar mainstream and achieve a balanced approach to performance, cost and reliability, while enabling scalable and intelligent 3D perception solutions across industries.

Cepton has been awarded the largest known ADAS lidar series production award in the industry to date, based on the number of vehicle models awarded, to support General Motors' Ultra Cruise program. Cepton is also engaged with all other Top 10 global OEMs.

Founded in 2016 and led by industry veterans with decades of collective experience across a wide range of advanced lidar and imaging technologies, Cepton is focused on the mass market commercialization of high performance, high quality lidar solutions. Cepton is headquartered in San Jose, CA and has a center of excellence facility in Troy, MI to provide local support to the OEM and Tier 1-studded Metro Detroit area. Cepton also has a presence in Germany, Canada, Japan, India and China to serve a fast-growing global customer base. For more information, visit www.cepton.com and follow Cepton on Twitter and LinkedIn.

About Growth Capital Acquisition Corp.

GCAC was formed as a Delaware blank check company, also commonly referred to as a special purpose acquisition company (or SPAC), formed for the purpose of entering into a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses or entities in any industry or geographic region. Prior to the business combination, GCAC was led by its Co-Chief Executive Officers, Akis Tsirigakis and George Syllantavos.

Advisors

- J.P. Morgan Securities LLC ("J.P. Morgan") served as financial advisor to Cepton and O'Melveny & Myers LLP served as legal counsel to Cepton. Maxim Group LLC ("Maxim") served as financial advisor to Growth Capital and Ellenoff Grossman & Schole LLP served as legal counsel to Growth Capital.
- J.P. Morgan acted as lead placement agent to Growth Capital. Maxim also served as joint placement agent. Skadden, Arps, Slate, Meagher & Flom LLP served as legal counsel to the placement agents.

Maxim and Craig-Hallum Capital Group LLC acted as capital markets advisors to Growth Capital. RBC Capital Markets acted as capital markets advisor to Cepton.

Cepton Technologies, Inc. Contacts

Investors: InvestorRelations@cepton.com Media: Faithy Li, media@cepton.com

Source: Cepton Technologies, Inc.

UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

References in this section to "we," "our," "us," "the Company" and "Cepton" generally refer to Cepton, Inc. and its consolidated subsidiaries after giving effect to the Business Combination and "Legacy Cepton" generally refers to Cepton Technologies, Inc. and its consolidated subsidiaries prior to the Business Combination.

Capitalized terms used but not defined in this exhibit shall have the meanings ascribed to them in the Current Report on Form 8-K (the "Report") filed with the Securities and Exchange Commission (the "SEC") on February 10, 2022 of which this exhibit is a part.

Introduction

The following unaudited pro forma condensed combined financial information has been prepared in accordance with Article 11 of Regulation S-X as amended by the final rule, Release No. 33-10786 "Amendments to Financial Disclosures about Acquired and Disposed Businesses."

GCAC was formed as a blank check company under the laws of the State of Delaware on January 4, 2010 under the name PinstripesNYS, Inc., and subsequently submitted a registration statement under the name Growth Capital Acquisition Corp. on February 27, 2020 for the purpose of effecting a merger, capital stock exchange, asset acquisition, stock purchase, reorganization or similar business combination with one or more businesses.

Cepton provides state-of-the-art, intelligent, lidar-based solutions for a range of markets such as automotive (ADAS/AV), smart cities, smart spaces, and smart industrial applications. Cepton's patented MMT[®]-based lidar technology enables reliable, scalable, and cost-effective solutions that deliver long range, high resolution 3D perception for smart applications.

The unaudited pro forma condensed combined balance sheet as of September 30, 2021 combines the historical balance sheet of GCAC as of September 30, 2021 with the historical balance sheet of Legacy Cepton as of September 30, 2021 on a pro forma basis as if the Business Combination and related transactions, summarized below, had been consummated on September 30, 2021.

GCAC and Legacy Cepton have different fiscal years. GCAC's fiscal year ends on March 31, whereas Legacy Cepton's fiscal year ends on December 31. The unaudited pro forma condensed combined statements of operations for the twelve months ended December 31, 2020 and for the nine months ended September 30, 2021 have been prepared utilizing Legacy Cepton's fiscal year end as that will be the year end for Cepton. Accordingly, the unaudited pro forma condensed combined statement of operations for the fiscal year ended December 31, 2020 combines the historical results of GCAC for its fiscal year ended March 31, 2021 (as restated) and the historical results of Legacy Cepton for the year ended December 31, 2020. The unaudited pro forma condensed combined statement of operations for the twelve months ended December 31, 2020 has been prepared utilizing period ends that differ by less than 93 days, as permitted by Rule 11-02 of Regulation S-X. The unaudited pro forma condensed combined statement of operations for the nine months ended September 30, 2021 combines the historical statement of operations of GCAC and Legacy Cepton for the nine months ended September 30, 2021. The historical statement of operations of GCAC for the nine months ended September 30, 2021, was derived from GCAC's unaudited condensed statement of operations for the nine months ended September 30, 2021 (as restated), and unaudited condensed statement of operations for the six months ended September 30, 2021. The unaudited pro forma condensed combined statements of operations for the nine months ended September 30, 2021 and for the year ended December 31, 2020 combine the historical statements of operations of GCAC and Legacy Cepton for such periods on a pro forma basis as if the Business Combination and related transactions, summarized below, had been consummated on January 1, 2020, the beginning of the earliest period presented.

The unaudited pro forma condensed combined balance sheet as of September 30, 2021 and the unaudited pro forma condensed combined statements of operations for the nine months ended September 30, 2021 and for the year ended December 31, 2020 are presented after giving effect to the Business Combination. In connection with the consummation of the Business Combination, the following occurred:

• Merger Sub, the wholly owned subsidiary of GCAC, merged with and into Legacy Cepton, with Legacy Cepton surviving as a wholly owned subsidiary of Cepton;

- the issuance of 142,020,456 shares of common stock as Consideration Shares based on the Per Share Stock Consideration Rate of 2.465 shares and 57,606,847 Legacy Cepton Outstanding Shares (reflecting Legacy Cepton outstanding common stock inclusive of converted and exercised preferred stock, Cepton Class F stock, and warrants).
- the conversion of all 21,671,491 outstanding shares of Legacy Cepton preferred stock and 8,372,143 shares of Legacy Cepton Class F stock into 30,043,634 shares of Legacy Cepton common stock that rolled over into 74,067,767 shares of Cepton common stock based on the Per Share Stock Consideration Rate;
- the redesignation of GCAC's outstanding 1,660,460 Public Shares (net of redemptions of 15,589,540 Public Shares) and 4,312,500 Sponsor Shares as Cepton common stock;
- the exercise of warrants to purchase an aggregate of 60,000 shares of Legacy Cepton common stock for shares of Cepton common stock that rolled over into 137,960 shares of Cepton common stock based on the Per Share Stock Consideration Rate;
- the conversion of all 3,236,692 outstanding vested Legacy Cepton options and all 2,541,610 outstanding unvested Legacy Cepton options into 7,979,544 vested options and 6,265,932 unvested options in Cepton, respectively, based on the Per Share Stock Consideration Rate (the "Converted Options"). In accordance with the Business Combination Agreement, the exercise price per share of all outstanding vested and unvested Legacy Cepton options will be adjusted by dividing the applicable exercise price per share immediately prior to the Business Combination by the Per Share Stock Consideration Rate. The Converted Options are presented on a diluted basis, calculated in accordance with the treasury stock method of accounting;
- the issuance of 5,950,000 shares of Cepton common stock to the PIPE Investors in exchange for \$59.5 million, or \$10.00 per share, in consideration;
- the recognition of a liability related to the issuance of up to 13,000,000 Earnout Shares contingently issuable to holders of Cepton common stock based upon achievement of the Share Price Milestones. Share Price Milestones are met if the share price of Cepton common stock equals or exceeds \$15.00 per share (first Share Price Milestone) and/or \$17.50 per share (second Share Price Milestone) for any 20 trading days within any consecutive 30-trading day period that occurs after the Closing, and on or prior to the three-year anniversary of the Closing. Upon achievement of the first Share Price Milestone, 7,000,000 Earnout Shares shall be deemed earned and issued and 6,000,000 Earnout Shares shall be deemed earned and issued upon achievement of the second Share Price Milestone. The Earnout Shares will not be issued if none of the Share Price Milestones are met as of the three-year anniversary of the Closing; and
- the issuance of 50,000 shares of Cepton common stock to Lincoln Park Capital Fund, LLC as consideration for entering into the Purchase Agreement. The Purchase Agreement was executed on November 24, 2021, and as such, the 50,000 shares have not been included within the proforma combined financial statements presented below.

The historical financial information of GCAC was derived from the unaudited financial statements of GCAC as of and for the nine months ended September 30, 2021 and from the audited financial statements for the year ended March 31, 2021 (as restated), included or incorporated by reference in this Report. The historical financial information of Legacy Cepton was derived from the unaudited consolidated financial statements of Legacy Cepton as of and for the nine months ended September 30, 2021; and from the audited consolidated financial statements for the year ended December 31, 2020, included or incorporated by reference in this Report. This information should be read together with GCAC's and Legacy Cepton's audited and unaudited financial statements and related notes, the sections titled "Other Information Related to GCAC — GCAC's Management's Discussion and Analysis of Financial Condition and Results of Operations," and "Cepton's Management's Discussion and Analysis of Financial Condition and Results of Operations" and other financial information included or incorporated by reference in this Report.

The pro forma combined financial statements have been presented for informational purposes only and are not necessarily indicative of what Cepton's financial position or results of operations actually would have been had the transactions been completed as of the dates indicated. In addition, the pro forma data do not purport to project the future financial position or operating results of Cepton. The actual financial position and results of operations may differ significantly from the pro forma amounts reflected herein due to a variety of factors.

Accounting for the Business Combination

The Business Combination will be accounted for as a reverse recapitalization in accordance with GAAP. Under this method of accounting, GCAC, who is the legal acquirer, will be treated as the "acquired" company for financial reporting purposes and Legacy Cepton will be treated as the accounting acquirer. Legacy Cepton has been determined to be the accounting acquirer based on evaluation of the following facts and circumstances:

- Legacy Cepton's existing stockholders will have more than 90% of the voting interest of Cepton;
- Legacy Cepton's senior management will comprise the senior management of Cepton;
- the directors nominated by Legacy Cepton will represent the majority of the board of directors of Cepton;
- Legacy Cepton is the larger entity based on historical revenues and business operations;
- Legacy Cepton's operations will comprise the ongoing operations of Cepton; and
- GCAC will assume the name Cepton, Inc.

Accordingly, for accounting purposes, the Business Combination will be treated as the equivalent of a capital transaction in which Legacy Cepton is issuing stock for the net assets of GCAC. The net assets of GCAC will be stated at historical cost, with no goodwill or other intangible assets recorded. Operations prior to the Business Combination will be those of Legacy Cepton.

Description of the Business Combination

Pursuant to the Business Combination Agreement, the aggregate stock consideration issued by Cepton in the Business Combination was \$1.539 billion, consisting of 153,943,416 newly issued shares of Cepton common stock valued at \$10.00 per share. Legacy Cepton shareholders received \$1.420 billion in the form of 142,020,456 newly issued shares of Cepton common stock. GCAC public shareholders received \$16.6 million in the form of 1,660,460 newly issued shares of Cepton common stock, the PIPE Investors received \$59.5 million in the form of 5,950,000 newly issued shares of Cepton common stock, and the Sponsor received \$43.1 million in the form of 4,312,500 newly issued shares of Cepton common stock in exchange for GCAC's existing Class B common stock. The following represents the consideration at closing of the Business Combination:

(in millions)

	400
Share issuance to GCAC shareholders, net of redemptions	\$ 16.6
Share issuance to Sponsor	43.1
Share issuance to Legacy Cepton shareholders	1,420.2
Share issuance to PIPE Investors	 59.5
Share Consideration at Closing	\$ 1,539.4

The value of share consideration issuable at the Closing was determined by application of the Exchange Ratio of 2.465, which is based on the implied value of \$10.00 per share prior to the Business Combination.

The following tables summarize the pro forma Cepton shares of common stock issued and outstanding immediately after the Business Combination both on an issued and outstanding share and diluted basis:

	Issued Outstanding S		Diluted Basis ⁽¹⁾		
	Shares	% Owned	Shares	% Owned	
Public Shares, net of redemptions	1,660,460	1.1%	1,660,460	1.0%	
Sponsor Shares	4,312,500	2.8%	4,312,500	2.7%	
Consideration Shares issued in the merger	142,020,456	92.2%	150,000,000	92.6%	
Shares issued to PIPE Investors	5,950,000	3.9%	5,950,000	3.7%	
Pro Forma common stock at September 30, 2021	153,943,416	100.0%	161,922,960	100.0%	

The following unaudited pro forma condensed combined balance sheet as of September 30, 2021 and the unaudited pro forma condensed combined statements of operation for the nine months ended September 30, 2021 and for the twelve months ended December 31, 2020 are based on the historical financial statements of GCAC and Legacy Cepton. The unaudited pro forma adjustments represent management's estimates based on information available as of the date of these unaudited pro forma combined financial statements. The assumptions and estimates underlying the unaudited pro forma adjustments are described in the accompanying notes. Actual results may differ materially from the assumptions used to present the accompanying unaudited pro forma condensed combined financial information as additional information becomes available and analyses are performed. Certain amounts that appear in this section may not sum due to rounding.

Diluted Basis is equal to the issued and outstanding share basis plus the 7,979,544 vested Converted Options.

UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET AS OF SEPTEMBER 30, 2021 (In thousands)

	(A) Legacy Cepton	(B) GCAC	Transaction Accounting Adjustments		Pro Forma Balance Sheet
Assets					
Current assets:					
Cash and cash equivalents	7,655	239	172,516	(1)	
	_		59,500	(2)	
	_		(39,157) (155,896)	(3) (9)	44,857
Short-term investments	8,448	_	(155,030)	(3)	8,448
Accounts receivable	775	<u> </u>			775
Inventories	2,840	_			2,840
Prepaid expenses and other current assets	5,547	46	3,934	(3)	9,527
Total current assets	25,265	285	40,897		66,447
Cash held in trust account		172,516	(172,516)	(1)	
Property and equipment, net	432	1/2,510	(1/2,510)	(1)	432
Other assets	373	_			373
Total assets	26,070	172,801	(131,619)		67,252
	20,070	172,001	(151,015)		07,232
Liabilities, convertible preferred stock, and stockholders' equity Current liabilities:					
Accounts payable	2,143	_	_		2,143
Accrued expenses and other current liabilities	2,946	576	(1,617)	(3)	1,905
Current portion of debt					
Total current liabilities	5,089	576	(1,617)		4,048
	1-				1=
Other long-term liabilities	15 —	12 000	(0.711)	(10)	15
Warrant Liability Earnout Liability	_	13,990	(8,711) 9,780	(10) (8)	5,279 9,780
Total liabilities		44.500		(0)	
Total natifices	5,104	14,566	(548)		19,122
Commitments and contingencies					
Commitments and contingencies	00.470		(00, 470)	(7)	
Legacy Cepton convertible preferred stock ^(C)	99,470	150 500	(99,470)	(7)	
GCAC Class A common stock subject to possible redemption ^(C)	_	172,500	(172,500)	(4)	_
Stockholders' (deficit) equity					
GCAC preferred stock ^(C)					
		<u> </u>		(F)	<u> </u>
Legacy Cepton common stock ^(C)	_	_		(5)	_
GCAC Class A common stock ^(C)	_	_	1	(2)	
			2 1	(4)	
			2	(5) (7)	
			(2)	(9)	4
GCAC Class B common stock ^(C)	_	_	(<u>-</u>)	(0)	T
Legacy Cepton Class F stock ^(C)	_	_	_		
Additional paid-in capital	6,115	_	59,499	(2)	
	,		(24,000)	(3)	
			(2,497)	(3)	
			172,498	(4)	
			(1)	(5)	
			(14,265)	(6)	
			99,468	(7)	
			(9,780)	(8)	
			8,711 (155,894)	(10) (9)	139,894
			(155,054)	(3)	155,054
Accumulated other comprehensive income (loss)	(40)	_			(40)
Retained earnings (accumulated deficit)	(84,619)	(14,265)	14,265	(6)	(-)
- · · · · · · · · · · · · · · · · · · ·		, , ,	(7,109)	(3)	(91,728)
Total stockholders' (deficit) equity	(78,504)	(14,265)	140,899		48,130
Total liabilities, convertible preferred stock, and stockholders'					
(deficit) equity	\$ 26,070	172,801	(131,619)		\$ 67,252

- (A) Obtained from the unaudited consolidated balance sheet of Legacy Cepton as of September 30, 2021.(B) Obtained from the unaudited balance sheet of GCAC as of September 30, 2021.
- (C) Authorized, issued and outstanding shares for each class of common stock and preferred stock as of September 30, 2021 and on a pro forma basis is as

	Se	ptember 30, 2021	<u> </u>	<u> </u>		
	Authorized	Issued	Outstanding	Authorized	Issued	Outstanding
Legacy Cepton convertible preferred stock	22,806,009	21,671,491	21,671,491			
Legacy Cepton common stock	75,000,000	27,507,253	27,507,253	_	_	_
Legacy Cepton Class F stock	8,402,000	8,372,143	8,372,143	_	_	_
GCAC preferred stock	1,000,000	_	_	5,000,000	_	_
GCAC Class A common stock subject to						
possible redemption	17,250,000	17,250,000	17,250,000	_	_	_
GCAC Class A common						
stock	100,000,000	_		350,000,000	153,943,416	153,943,416
GCAC Class B common						
stock	10,000,000	4,312,500	4,312,500	_	_	_
	10,000,000	4,312,500	4,312,500	_	_	_

UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS FOR THE NINE MONTHS ENDED September 30, 2021

(In thousands, except per share amounts)

	(A) Legacy Cepton		(B) GCAC		(C) Transaction Adjustments			ro Forma Income tatement
Lidar Sensor and Prototype Revenue	\$	1,989	\$		\$		\$	1,989
Development Revenue		1,235						1,235
Total Revenue		3,224		_		_		3,224
Lidar Sensor and Prototype Cost of Revenue		3,053		_		_		3,053
Development Cost of Revenue		3,104		<u> </u>		<u> </u>		3,104
Total Cost of Revenue		6,157						6,157
Gross Margin	\$	(2,933)	\$		\$		\$	(2,933)
Operating expenses								
Research and development		14,593		_		_		14,593
Selling, general and administrative		9,992		1,177		<u> </u>		11,169
Total operating expenses	\$	24,585	\$	1,177	\$	_	\$	25,762
Loss from operations	\$	(27,518)	\$	(1,177)	\$	_	\$	(28,695)
Income from investments held in Trust Account		_		18		(18)		_
Other expense:								
Warrant Transaction costs		_		(293)		293		
Excess value of UW warrants		_		(1,294)		1,294		_
Unrealized gain/(loss) on FV changes of warrants				3,088		(1,808)		1,281
Other income (expense), net		1,098		_		_		1,098
Interest income (expense), net		14	_					14
Loss before income taxes		(26,406)	_	342	_	(239)		(26,303)
Provision for income taxes		16						16
Net (loss) income	\$	(26,422)	\$	342	\$	(239)	\$	(26,319)
Other comprehensive loss								
Changes in unrealized gain on available-for-sale-securities		(4)		_		_		(4)
Foreign currency translation adjustment		(18)	_			<u> </u>		(18)
Comprehensive loss	\$	(26,444)	\$	342	\$	(239)	\$	(26,341)
Net loss per common share								
Weighted average shares of redeemable common stock outstanding, basic and diluted				17,250,000	1	36,693,416	1	153,943,416
Basic and diluted net income (loss) per share			\$	0.02		55,055,410	\$	(0.17)
Weighted average shares of non-redeemable common stock outstanding, basic and diluted		27 255 00 4		4 212 500				
	_	27,355,884	_	4,312,500				
Basic and diluted net income (loss) per share	\$	(0.97)	\$	0.02				

- (A) Obtained from the unaudited consolidated statement of operations of Legacy Cepton for the nine months ended September 30, 2021.
- (B) Derived from GCAC's unaudited condensed statement of operations for the nine months ended December 31, 2020 (as restated), audited condensed statement of operations for the year ended March 31, 2021 (as restated), and unaudited condensed statement of operations for the six months ended September 30, 2021.
- (C) Reflects (i) the elimination of investment income from the Trust Account, (ii) the elimination of warrant transaction costs and excess value of underwriter warrant charges as the warrant costs are included within the pro forma statement of operations for the year ended December 31, 2020, the earliest period presented, (iii) the reversal of the unrealized loss on change in fair value of warrant liabilities related to Public Warrants recognized in GCAC's statement of operations for the nine-month period ended September 30, 2021 on the basis of Cepton's conclusion that the Public Warrants will be equity instruments after the Business Combination, and (iv) the additional shares of Cepton common stock to be issued upon completion of the Business Combination, taking into account actual redemptions.

UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS FOR THE YEAR ENDED DECEMBER 31, 2020

(In thousands, except per share amounts)

	Year Ended December 31, 2020 (A) Legacy Cepton	Year ended March 31, 2021 (as restated) (B) GCAC	(C) Transaction Adjustments	Pro Forma Income Statement
Lidar Sensor and Prototype Revenue	\$ 2,006	\$ —	\$ —	\$ 2,006
Development Revenue				
Total revenue	2,006	_	_	2,006
Lidar Sensor and Prototype Cost of Revenue	3,746	_	_	3,746
Development Cost of Revenue	_	_	_	_
Total Cost of Revenue	3,746			3,746
Gross Margin	(1,740)		_	(1,740)
	() - '			() -)
Operating expenses				
Research and development	11,666	_	_	11,666
Selling, general and administrative	6,170	93	7,109	13,372
Total operating expenses	17,836	93	7,109	25,038
and the second s	17,050		-,,100	
Loss from operations	(19,576)	(93)	(7,109)	(26,778)
Income from investments held in Trust Account	_	6	(6)	_
Other expense:				
Warrant Transaction costs	_	(293)	180	(113)
Excess value of UW warrants	_	(1,294)	_	(1,294)
Unrealized gain/(loss) on FV changes of warrants	_	9,936	(6,120)	3,816
Interest income, net	149	_	_	149
Other (income) expense, net	(181)	_	_	(181)
Loss before income taxes	(19,608)	8,262	(13,055)	(24,401)
2000 DELVIE MEDIME UMED	(15,000)		(15,055)	(=1,101)
Provision for income taxes	26	<u>_</u>	<u>_</u>	26
Net (loss) income		0.202	(12.055)	
	(19,634)	8,262	(13,055)	(24,427)
Other comprehensive loss				
Changes in unrealized gain on available-for-sale securities	3	_	_	3
Foreign currency translation adjustment	(11)			(11)
Comprehensive loss	(19,642)	8,262	(13,055)	(24,435)
Net loss per common share				
Weighted average redeemable common shares outstanding, basic and diluted		17,250,000	136,693,416	153,943,416
Basic and diluted net income (loss) per share		\$ 0.38		\$ (0.16)
Weighted average shares of non-redeemable common stock outstanding, basic and diluted	27,068,162	4,312,500		
Basic and diluted net income (loss) per share				
Davie and anated net meome (1999) per snare	\$ (0.73)	\$ 0.38		

- (A) Obtained from the audited consolidated statement of operations of Legacy Cepton for the year ended December 31, 2020.
- (B) Obtained from the audited statement of operations of GCAC for the year ended March 31, 2021 (as restated).
- (C) Reflects the transaction costs incurred by GCAC in 2021 including, but not limited to, advisory fees, legal fees, and registration fees. This is a non-recurring item. Also, reflects the reversal of the impacts associated with the warrant liabilities related to Public Warrants recognized on GCAC's historical statement of operations for the period ended March 31, 2021 on the basis of Cepton's conclusion that the Public Warrants will be equity instruments after the Business Combination. Additionally, reflects the elimination of investment income from the Trust account. Finally, reflects the incremental shares of Cepton common stock to be issued upon completion of the Business Combination.

NOTES TO UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

1. Basis of Presentation

The Business Combination will be accounted for as a reverse recapitalization in accordance with GAAP. Under this method of accounting, GCAC, who is the legal acquirer, will be treated as the "acquired" company for financial reporting purposes and Legacy Cepton will be treated as the accounting acquirer. This determination was primarily based on the following facts and circumstances: (i) Legacy Cepton's existing stockholders will have more than 90% of the voting interest of Cepton; (ii) Legacy Cepton's senior management will comprise the senior management of Cepton; (iii) the directors nominated by Legacy Cepton will represent the majority of the board of directors of Cepton; (iv) Legacy Cepton is the larger entity based on historical revenues and business operations; (v) Legacy Cepton's operations will comprise the ongoing operations of Cepton; and (vi) GCAC will assume the name Cepton, Inc.. Accordingly, for accounting purposes, the Business Combination will be treated as the equivalent of a capital transaction in which Legacy Cepton is issuing stock for the net assets of GCAC. The net assets of GCAC will be stated at historical cost, with no goodwill or other intangible assets recorded. Operations prior to the Business Combination will be those of Legacy Cepton. The unaudited pro forma condensed combined balance sheet as of September 30, 2021 assumes the Business Combination occurred on September 30, 2021. The unaudited pro forma condensed combined statements of operation for the nine months ended September 30, 2021 and for the twelve months ended December 31, 2020 present the pro forma effect of the Business Combination as if it had been completed on January 1, 2020, the beginning of the earliest period presented. These periods are presented on the basis of Legacy Cepton as the accounting acquirer.

The unaudited pro forma condensed combined balance sheet as of September 30, 2021 has been prepared using, and should be read in conjunction with, the following:

- GCAC's unaudited balance sheet as of September 30, 2021 and the related notes for the period ended September 30, 2021, included or incorporated by reference in this Report; and
- Legacy Cepton's unaudited consolidated balance sheet as of September 30, 2021 and the related notes for the nine months ended September 30, 2021 included or incorporated by reference in this Report.

The unaudited pro forma condensed combined statement of operations for the nine months ended September 30, 2021 and for the twelve months ended December 31, 2020 have been prepared using, and should be read in conjunction, with the following:

- GCAC's historical statement of operations for the nine months ended September 30, 2021, as derived from GCAC's unaudited condensed statement of operations for the nine months ended December 31, 2020 (as restated), audited condensed statement of operations for the year ended March 31, 2021 (as restated), and unaudited condensed statement of operations for the six months ended September 30, 2021, and the related notes included or incorporated by reference in this Report; and
- Legacy Cepton's unaudited and audited consolidated statement of operations and comprehensive income (loss) for the nine months ended September 30, 2021 and for the year ended December 31, 2020, and the related notes included or incorporated by reference in this Report.

Management has made significant estimates and assumptions in its determination of the pro forma adjustments. As the unaudited pro forma condensed combined financial information has been prepared based on these preliminary estimates, the final amounts recorded may differ materially from the information presented.

The unaudited pro forma condensed combined financial information does not give effect to any anticipated synergies, operating efficiencies, tax savings or cost savings that may be associated with the Business Combination. The pro forma adjustments reflecting the consummation of the Business Combination are based on certain available information as of the date of these unaudited pro forma combined financial statements and certain assumptions and methodologies that Cepton believes are reasonable under the circumstances. The unaudited condensed pro forma adjustments, which are described in the accompanying notes, may be revised as additional information becomes available and is evaluated. Therefore, it is likely that the actual adjustments will differ from the pro forma adjustments and it is possible the difference may be material. Cepton believes that its assumptions and methodologies provide a reasonable basis for presenting all of the significant effects of the Business Combination based on information available to management at the time and that the pro forma adjustments give appropriate effect to those assumptions and are properly applied in the unaudited pro forma condensed combined financial information.

The unaudited pro forma condensed combined financial information is not necessarily indicative of what the actual results of operations and financial position would have been had the Business Combination taken place on the dates indicated, nor are they indicative of the future consolidated results of operations or financial position of Cepton. They should be read in conjunction with the historical financial statements and notes thereto of GCAC and Legacy Cepton.

2. Adjustments to Unaudited Pro Forma Condensed Combined Financial Information

The unaudited pro forma condensed combined financial information has been prepared to illustrate the effect of the Business Combination and has been prepared for informational purposes only.

The historical financial statements have been adjusted in the unaudited pro forma condensed combined financial information to give pro forma effect to events that are directly attributable to the Business Combination. Legacy Cepton and GCAC have not had any historical relationship prior to the Business Combination. Accordingly, no pro forma adjustments were required to eliminate activities between the companies.

The pro forma combined provision for income taxes does not necessarily reflect the amounts that would have resulted had Cepton filed consolidated income tax returns during the periods presented.

The pro forma basic and diluted earnings per share amounts presented in the unaudited pro forma condensed combined statement of operations are based upon the number of Cepton's shares outstanding, assuming the Business Combination had been completed on January 1, 2020, the beginning of the earliest period presented.

Adjustments to Unaudited Pro Forma Condensed Combined Balance Sheet

The adjustments included in the unaudited pro forma condensed combined balance sheet as of September 30, 2021 are as follows:

- (1) Reflects the release of cash currently invested in U.S. treasuries or money market funds held in the Trust Account.
- (2) Reflects the proceeds received from the PIPE Investment with the corresponding issuance of 5,950,000 shares of Cepton common stock at \$10.00 per share.
- (3) Represents preliminary estimated direct and incremental transaction costs incurred by GCAC and Legacy Cepton related to the Business Combination. Total transaction costs are estimated to be \$40.6 million related to underwriting/banking, legal, accounting and other miscellaneous fees, including \$2.5 million in deferred transaction costs incurred by Legacy Cepton. Of the \$2.5 million in deferred transaction costs incurred by Legacy Cepton, approximately \$1.0 million is reflected in accrued expenses and will be paid in cash upon the closing of the Business Combination. The remaining \$1.5 million of deferred transaction costs were previously paid for in cash by Legacy Cepton. Of the remaining \$38.1 million in total transaction costs, \$24.0 million is reflected in the unaudited pro forma condensed combined balance sheet as a reduction to Cepton's additional paid-in capital as they are incremental costs directly attributable to a proposed offering of securities and thus charged against the gross proceeds of the proposed offering, which includes the Business Combination and PIPE Investment. Additionally, this adjustment reflects the recognition of approximately \$6.4 million in prepaid assets associated with directors' and officers' liability insurance for Cepton. This adjustment also reflects the payment of \$0.6 million in accrued expenses recognized by GCAC related to the contemplated business combination. The remaining \$7.1 million are reflected in the unaudited pro forma condensed combined balance sheet as an increase to Cepton's accumulated deficit. These relate to the transaction costs incurred by GCAC during the nine month period ending September 30, 2021 including, but not limited to, advisory fees, legal fees and registration fees. These transaction costs are also reflected in the unaudited pro forma condensed combined statement of operations for the year ended December 31, 2020 as an adjustment to selling, general and administrative expense.

- (4) Reflects the reclassification of GCAC class A common stock previously subject to possible redemption to permanent equity immediately prior to the consummation of the Business Combination.
- (5) Reflects the recapitalization of Legacy Cepton through (a) the contribution of all the share capital in Legacy Cepton to GCAC in the amount of \$6.2 million and (b) the issuance of 142,020,456 shares of Cepton common stock at par value of \$0.00001.
- (6) Reflects the elimination of the historical retained earnings of GCAC, the legal acquirer, in the amount of \$14.3 million.
- (7) Reflects the reclassification of approximately \$99.5 million of Legacy Cepton's convertible preferred stock (21,671,491 shares at redemption value) to permanent equity. Legacy Cepton's convertible preferred stock converted into Legacy Cepton common stock at a 1:1 ratio.
- (8) Reflects the preliminary estimated fair value of the Earnout Shares contingently issuable to holders of Legacy Cepton common stock. The preliminary fair value was determined using the most reliable information currently available. The actual fair value could change materially once the final valuation is determined upon Closing. Refer to Note 4 for more information. Subsequent to the Business Combination, this liability will be remeasured to its fair value at the end of each reporting period and subsequent changes in the fair value will be recognized in Cepton's statement of operations within other income/expense.
- (9) Reflects the actual redemptions of 15,589,540 public shares for aggregate redemption payments of \$155,895,400 allocated to GCAC Class A Common Stock and additional paid-in capital using par value \$0.0001 per share and at a redemption price of \$10 per share.
- (10) Cepton has evaluated the accounting for GCAC's Public Warrants and Private Placement Warrants for Cepton under ASC 480 and ASC 815. Cepton has concluded that the Public Warrants qualify as equity instruments under ASC 815 after considering among other factors that after the Business Combination, Cepton will have a single class equity structure; thus, a tender offer subject to the terms of the warrant agreement will always result in a change of control, and holders of Public Warrants receive the same form of consideration as holders of Legacy Cepton common stock which treatment does not preclude equity classification. Separately, Cepton has concluded that the Private Placement Warrants will continue to be accounted for as a liability under ASC 815-40. The adjustment reflects the reclassification of GCAC's Public Warrants from liabilities to equity in connection with the consummation of the Business Combination.

3. Net loss per Share

Represents the net loss per share calculated using the historical weighted average shares outstanding, and the issuance of additional shares in connection with the Business Combination, assuming the shares were outstanding since January 1, 2020, the beginning of the earliest period presented. As the Business Combination is being reflected as if it had occurred at the beginning of the period presented, the calculation of weighted average shares outstanding for basic and diluted net loss per share assumes that the shares issuable relating to the Business Combination have been outstanding for the entire period presented.

The unaudited pro forma condensed combined financial information has been prepared using actual redemption amounts:

	 Nine Months Ended September 30, 2021		Twelve Months Ended ecember 31, 2020
Pro forma net loss	\$ (26,319)	\$	(24,427)
Basic weighted average shares outstanding	153,943,416		153,943,416
Net loss per share – Basic and Diluted ⁽³⁾	\$ (0.17)	\$	(0.16)
Public Shares, net of redemptions	1,660,460		1,660,460
Sponsor Shares	4,312,500		4,312,500
Cepton shares issued to existing Legacy Cepton stockholders ^{(1),(2)}	142,020,456		142,020,456
Cepton shares issued to PIPE Investors	 5,950,000		5,950,000
	153,943,416		153,943,416

(1) This total is based on the Legacy Cepton cap table as of September 30, 2021.

This total represents Cepton shares of common stock issued to the following stockholders:

- 67,814,726 shares to Legacy Cepton common stockholders (27,507,253 shares prior to conversion);
- 20,640,178 shares to Legacy Cepton Class F stockholders (8,372,142 shares prior to conversion);
- 53,427,590 shares to Legacy Cepton preferred stockholders (21,671,491 shares prior to conversion); and
- 137,960 shares to Legacy Cepton warrant holders (55,960 shares prior to conversion).
- (2) This amount excludes: (i) 7,979,544 shares reserved for vested Converted Options using the treasury stock method which will not be legally outstanding following the Business Combination and which are allocated from the 150,000,000 shares issuable to existing Legacy Cepton stockholders per the Business Combination Agreement and (ii) 50,000 Commitment Shares issued to Lincoln Park Capital Fund, LLC in consideration for entering into the Purchase Agreement dated November 24, 2021.
- (3) As a result of the pro forma net loss, the net loss per share amounts exclude the anti-dilutive impact from 17,365,694 Converted Options, 8,625,000 Public Warrants to purchase 8,625,000 shares of Cepton common stock, 5,175,000 Private Warrants to purchase 5,175,000 shares of Cepton common stock, and 169,492 unvested restricted shares of Cepton common stock outstanding as of September 30, 2021 on a pro forma basis.

4. Earnout Shares

The Earnout Shares are expected to be accounted for as liability classified equity instruments that are earned upon achievement of the Share Price Milestones, which provide for settlement provisions that are not indexed to Cepton common stock. The preliminary estimated fair value of the Earnout Shares is \$9.8 million.

The estimated fair value of the Earnout Shares was determined by a Monte Carlo simulation valuation model using a distribution of potential outcomes on a monthly basis over the three-year earnout period. The preliminary estimated fair value of the Earnout Shares was determined using the most reliable information currently available. Assumptions used in the preliminary valuation, which are subject to change at the Closing, were as follows:

Current stock price: the current stock price was set at the current value of \$9.86 per share for GCAC Class A common stock.

Expected volatility: the volatility rate of 60.0% was determined using an average of historical volatilities of selected industry peers deemed to be comparable to Legacy Cepton's business, corresponding to the expected term of the awards.

Risk-free interest rate: the risk-free interest rate of 0.4191% is based on the U.S. Treasury yield curve in effect at the time of issuance for zero-coupon U.S. Treasury notes with maturities corresponding to the expected three-year term of the earnout period.

Expected term: the expected term is the three-year term of the earnout period.

Expected dividend yield: the expected dividend yield is zero as Legacy Cepton has never declared or paid cash dividends and Cepton has no current plans to do so during the expected term.

The actual fair values of Earnout Shares are subject to change as additional information becomes available and additional analyses are performed and such changes could be material once the final valuation is determined at the Closing.